COLLECTIVE BARGAINING, INCOMES POLICY AND RELATIVE WAGE FLEXIBILITY IN GREEK MANUFACTURING, 1966-1988.

BY

CHRISTOS IOANNOU-GIANNAKIS

LONDON SCHOOL OF ECONOMICS AND POLITICAL SCIENCE

THESIS SUBMITTED FOR THE DEGREE OF DOCTOR OF PHILOSOPHY AT THE UNIVERSITY OF LONDON

•

FEBRUARY 1990

.

UMI Number: U048732

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



UMI U048732 Published by ProQuest LLC 2014. Copyright in the Dissertation held by the Author. Microform Edition © ProQuest LLC. All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code.



ProQuest LLC 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106-1346



THESES F 6702

x211092036

Abstract

Based on the assumption that industrial relations influence labour market outcomes, the thesis examines first, the characteristics and the evolution of bargaining structures and procedures in the Greek system of industrial relations, second, the governmental policies aimed at wage and employment regulation, and third, the extent to which, in the context of developments in industrial relations as well as in the context of incomes policy, there was room for relative wage flexibility in the Greek manufacturing sector.

The main conclusion of the thesis is that, despite the extensive and continuous regulation of wage determination procedures by successive governments, changes in industrial relations which occurred after 1975 and were marked by decentralised, fragmented and informal collective bargaining, were accompanied by flexibility in relative wages.

The sources of this flexibility, which is largely noncompetitive, are related to industry-specific productivity gains as well as to industry-specific rates of strike activity. Moreover, the rise of decentralised, fragmented and informal collective bargaining influenced the effectiveness of the normbased and the indexation incomes policies as far as variation in the inter-industry wage structure in the Greek manufacturing sector is concerned.

Table of Contents

Chapter 1. Introduction and Summary.

Chapter 2. Collective Bargaining: Institutions And Processes. 2.1. Introduction and Summary.

2.2. Collective Bargaining and Bargaining Structures.

2.2.1. Bargaining Structures: Levels and Units.

2.2.2. Decentralisation in Bargaining Structures.

2.2.3. Fragmentation and Coherence in Bargaining Structures.

2.2.4. Summary.

2.3. Collective Bargaining and the Formal Bargaining Structure in Greece.

2.3.1. Procedures , Levels and Units of Bargaining.

2.3.2. Bargaining Structure and Trade Unions.

2.3.3. The Formal Bargaining Structure: Stability or Change?

2.4. The Formal Bargaining Structure in Manufacturing.

2.4.1. Restrictions and Evolution.

2.4.2. Regulation of Formal Bargaining in Manufacturing.

2.4.3. Formal Pressures towards Decentralisation.

2.5. Conclusions.

Tables.

Chapter 3. Third Party Intervention and Compulsory Arbitration.

3.1. Introduction and Summary.

3.2. Arbitration and Collective Bargaining.

3.2.1. Third Party Intervention and Arbitration.

3.2.2. The Behaviour of Arbitrators.

3.2.3. Arbitration and Incomes Policy.

3.2.4. Summary.

- 3.3. Third Party Intervention in Greece: Compulsory-not-Quite-Arbitration.
 - 3.3.1. Provisions, Evolution and Origins.
 - 3.3.2. Compulsion versus Arbitration: A Corporatist Model?
 - 3.3.3. The Centrality of Compulsory Arbitration.
- 3.4. Strike Activity and Decentralisation of the Manufacturing Bargaining Structure.
 - 3.4.1. Compulsory Arbitration and Strike Activity.
 - 3.4.2. Strike Activity and Informal Collective Bargaining in Manufacturing.
- 3.5. Conclusions.

Tables.

- Chapter 4. Wage and Employment Regulation in Greece.
- 4.1. Introduction and Summary.
- 4.2. Incomes Policy in Greece.
 - 4.2.1. Types and Characteristics of Incomes Policy.
 - 4.2.2. The Benchmark Period: 1967-1974.
 - 4.2.3. The Period 1975-1981: Norm-Based Policy.
 - 4.2.4. The Period 1982-1988: Partial Indexation.
- 4.3. Wage Developments in Greek Manufacturing.
 - 4.3.1. Wages and the Wage Structure across Manufacturing Industries.

4.3.2. Indices of Flexibility in Manufacturing Relative Wages.4.4. Employment Regulation and Trends in Employment in Greek

Manufacturing.

- 4.4.1. The Legislation on Collective Dismissals.
- 4.4.2. Manufacturing Employment: Change, Industrial Distribution and Characteristics.
- 4.5. Conclusions.

Tables.

Chapter 5. Relative Wage Flexibility in Greek Manufacturing.

- 5.1. Introduction and Summary.
- 5.2. Relative Wage Flexibility and Relative Employment.

5.2.1. The Flexibility Debate and Relative Wages.

5.2.2. Competitive Flexibility in Relative Wages.

5.2.3. Non-Competitive Flexibility in Relative Wages.

- 5.2.4. Non-Competitive Flexibility and Employment Growth.
- 5.2.5. Summary.
- 5.3. Hypotheses to Test and their Rationales.
 - 5.3.1. Relative Wage Flexibility in the context of Greek Industrial Relations?
 - 5.3.2. Competitive Flexibility in Relative Wages.
 - 5.3.3. Non-Competitive Flexibility in Relative Wages.
- 5.4. The Competitive Hypothesis: Analysis of Bivariate Correlation between Wage and Employment Changes.
- 5.5. Industry-Specific Determinants of Relative Wages: Cross-Section Regression Analysis.
 - 5.5.1. Method and Variables.
 - 5.5.2. Long-run Industry-specific Flexibility.

- 5.5.3. Short-run Industry-specific Flexibility.
- 5.5.4. Employment Consequences of the Asymmetrical Response of Wages to Productivity.

5.6. Conclusions.

Tables.

Chapter 6. Incomes Policies and Relative Wages in Greek Manufacturing.

6.1. Introduction.

6.2. Wage Structure, Incomes Policy and Business Cycle.

6.2.1. Incomes Policy Rationales.

6.2.2. Incomes Policy and Wage Structure.

6.2.3. Business Cycle and Wage Structure.

6.3. Hypotheses to Test and their Rationales.

6.3.1. High-wage and Low-wage Sectors.

6.3.2. The Influence of Incomes Policy.

6.3.3. The Influence of Business Cycle.

6.3.4. The Influence of Strike Activity.

6.4. Specification of Equations and Regression Results.

6.5. Conclusions.

Tables.

Chapter 7. Conclusions and Policy Implications.

Appendices.

Appendix 1. Systems of Arbitration in the U.S., the U.K. and

Australia.

- Appendix 2. Employment of Salary and Wage Earners in Greek Manufacturing.
- Appendix 3. Relative Wage Flexibility in Greek Manufacturing, Correlation and Regression Results after Including in the Employment Variable both Salary and Wage Earners instead of Wage Earners Alone.

References.

List of Tables

- 2.3.1. Bargaining Units in the Formal Bargaining Structure, All Sectors, 1955-1988.
- 2.3.2. Bargaining Units by Bargaining Level, All Sectors, 1973-1988.
- 2.3.3. Special Collective Agreements by Economic Sector, Selected Years.
- 2.4.1. Evolution and Composition of the Formal Bargaining Structure in Manufacturing, 1954-1983.
- 2.4.2. Procedures of Regulation of Collective Agreements and Awards in the Formal System of Collective Bargaining in Manufacturing, 1954-1983.
- 2.4.3. Voluntary and Regulated Collective Bargaining in Manufacturing, by Level, 1954-1983.
- 2.4.4. Collective Agreements and Compulsory Arbitration Awards in Manufacturing, by Level, 1955-1983.
- 3.3.1. Collective Agreements and Compulsory Arbitration Awards, All Sectors, 1964-1988.
- 3.3.2. Compulsory Arbitration Awards as Percentage of Both Agreements and Awards, by Level, All Sectors, 1973-1988.
- 3.4.1. Strikers Density, Frequency of Strikes and Relative Days Lost in Greece, All Sectors, 1953-1988.
- 3.4.2. Bivariate Correlation Coefficients Relating Compulsory Arbitration and Strike Activity in Greece, All Sectors, 1953-1988.
- 3.4.3. Number of Strikes According to Organisational Level, 1976-1981.
- 3.4.4. Strikers Density, Frequency of Strikes and Relative Days

Lost in Manufacturing, 1959-1988.

- 3.4.5. Bivariate Correlation Coefficients Relating Bargaining Regulation and Strike Activity in Manufacturing, 1959-1988.
- 3.4.6. Strikes per 100 Manufacturing Establishments: Frequency, Dispersion and Stability, 1962-1981.
- 3.4.7. Strikes per 100 Manufacturing Establishments by Branch, 1962-1981.
- 3.4.8. Ranking of Industrial Branches According to Strikers Density, 1976-1981.
- 3.4.9. Ranking of Industrial Branches According to Relative Days Lost in Strikes, 1975-1988.
- 3.4.10. Inter-industry Variation in Strike Activity, 1975-1988.
- 4.2.1. The Norm-Based Incomes Policy, 1975-1981.
- 4.2.2. The Indexation Incomes Policy, 1982-1988.
- 4.3.1. Nominal, Real, Relative Wages, Productivity, Unit Labour Costs, Minimum Wage as Percentage of Average, and Female Wage as percentage of Male Wage in Greek Manufacturing Industries, 1965-1988.
- 4.3.2. Inflation and Labour Market Conditions in Greece, 1965-1988.
- 4.3.3. Keys for Industrial Branches in Manufacturing Sector.
- 4.3.4. Ranking of Industrial Branches in the Wage Structure, Descending Order, 1965-1988.
- 4.3.5. Determinants of the Wage Structure in Greek Manufacturing, 1976, 1981.
- 4.3.6. Variability in the Greek Manufacturing Wage Structure,

1965-1988.

- 4.3.7. Over time Variability in Branch Relative Wages in Greek Manufacturing, Selected Periods.
- 4.4.1. Cases of Collective Dismissals Under Minister of Labour Jurisdiction, 1983-1986.
- 4.4.2. Cases of Collective Dismissals Under Prefect's Jurisdiction, 1983-1986.
- 4.4.3. Rate of Change in Manufacturing Employment in OECD Countries, Selected Periods.
- 4.4.4. Share of Manufacturing Employment in OECD Countries, Selected Periods.
- 4.4.5. Share of and Change in Manufacturing Employment in Greece, 1965-1988.
- 4.4.6. Share of Salary Earners in Manufacturing Employment, by Branch, 1965-1988.
- 4.4.7. Share for Female Employees in Manufacturing Employment, by Branch, 1976-1988.
- 4.4.8. Changes in Employment of Wage Earners in Manufacturing, by Branch, Selected Periods.
- 4.4.9. Distribution of Wage Earners Across Manufacturing Industries, 1965-1988.
- 4.4.10. Average Employment of Wage Earners in Manufacturing Industries, by Branch, Selected Periods.
- 4.4.11. Annual Changes in Manufacturing Employment of Wage Earners, by Branch, 1965-1988.
- 5.4.1. Bivariate Correlation Coefficients Relating Changes in Nominal Wages and Changes in Employment of Wage Earners, 1966-88, 1967-74, 1975-81, 1982-88.

- 5.4.2. Bivariate Correlation Coefficients Relating Changes in Nominal Wages and Changes in Employment of Wage Earners, Short-run and Long-run Changes over the Period 1966-1988.
- 5.4.3. Bivariate Correlation Coefficients Relating Annual Changes in Nominal Wages and Annual Changes in Employment of Wage Earners, Comparison Between Coefficients for All Branches and Coefficients for Expanding Branches.
- 5.5.1. Estimates of the Industry Specific Determinants of Nominal Wage Changes Across Manufacturing Industries in Greece, Base-year to End-year Changes.
- 5.5.2. Estimates of the Industry Specific Determinants of Nominal Wage Changes Across Manufacturing Industries in Greece, Year to Year Changes.
- 5.5.3. Estimates of the Relationship Between Relative Nominal Wage Changes and Relative Productivity Change, Base-year to End-year Change.
- 5.5.4. Estimates of the Relationship Between Relative Nominal Wage Changes and Relative Productivity Change, Year-to-Year Change.
- 6.4.1. Regression Equations to Explain Inter-Branch Variance in Wages, 1967-1988, (Equations 1-4).
- 6.4.2. Regression Equations to Explain Inter-Branch Variance in Wages, 1967-1988, (Equations 5-8).
- 6.4.3. Regression Equations to Explain Inter-Branch Variance in Wages, 1967-1988, (Equations 9-12).
- 6.4.4. Regression Equations to Explain Inter-Branch Variance in Wages, 1967-1988, (Equations 4, 13-15).

List of Tables in Appendices

- A4.4.8. Changes in Employment of salary and wage Earners in Manufacturing, by Branch, Selected Periods.
- A4.4.9. Distribution of Salary and Wage Earners Across Manufacturing Industries, 1965-1988.
- A4.4.10.Average Employment of Salary and Wage Earners in Manufacturing Industries, Selected Periods.
- A4.4.11.Annual Changes in Manufacturing Employment of Salary and Wage Earners, by Branch, 1965-1988.
- A5.4.1. Bivariate Correlation Coefficients Relating Changes in Nominal Wages and Changes in Employment of Salary and Wage Earners, 1966-88, 1967-74, 1975-81, 1982-88.
- A5.4.2. Bivariate Correlation Coefficients Relating Changes in Nominal Wages and Changes in Employment of Salary and Wage Earners, Short-run and Long-run Changes over the Period 1966-1988.
- A5.4.3. Bivariate Correlation Coefficients Relating Annual Changes in Nominal Wages and Annual Changes in Employment of Salary and Wage Earners, Comparison Between Coefficients for All Branches and Coefficients for Expanding Branches.
- A5.5.1. Estimates of the Industry Specific Determinants of Nominal Wage Changes Across Manufacturing Industries in Greece, Base-year to End-year Changes.
- A5.5.2. Estimates of the Industry Specific Determinants of Nominal Wage Changes Across Manufacturing Industries in Greece, Year-to-Year Changes.

List of Graphs and Figures

- 2.3.1. Bargaining Structure and Structure of Trade-Unionism in Greece.
- 6.3.1. Industries in the Wage and Strike Structure, Average Ranking, 1975-1988.

List of Abbreviations

- ACAC : Australian Conciliation and Arbitration Committee.
- ACAS : Advisory Conciliation and Arbitration Services.
- CBI : Confederation of British Industry.
- GSEE : Greek General Confederation of Labour.
- GSEVE : Greek General Confederation of Handicraft and Small-scale Industry Employers.
- ILO : International Labour Organisation.
- OAE : Organisation for Companies Rehabilitation.
- OECD : Organisation for Economic Co-operation and Development.
- SEV : Confederation of Greek Industry.
- TUC : Trade-Unions Congress.
- NSSG : National Statistical Service of Greece.

CHAPTER 1. INTRODUCTION AND SUMMARY

The system of industrial relations in Greece is regulated as far as procedural (collective bargaining units and levels) and substantial (wages and employment adjustment) aspects are concerned. The purpose of the thesis is to examine the extent of this regulation, its consequences regarding the evolution of bargaining structures and procedures, and the extent to which, in the context of regulation and changing industrial relations, there has been any room for relative wage flexibility. In the same context, another purpose is to evaluate governmental incomes policies with regard to flexibility in the labour market.

It is a running theme of this thesis that developments in industrial relations have played a central role in relative wage flexibility in Greek manufacturing. More specifically, by developments in industrial relations we mean the changes towards decentralised and fragmented bargaining at the company and plant level through bargaining procedures evolving out of the formal system of industrial relations.

As in every industrialised market economy, in Greece too, wage setting is not just an economic process. Institutional arrangements and developments in industrial relations do matter as far as the workings of the labour market are concerned. Recently there has been an increasing interest on bargaining

structures, industrial relations systems and their relationship with economic performance and structural adjustment.

Considered in a comparative perspective, bargaining structures have been brought at the forefront of the research and debate regarding macroeconomic performance and its association with labour market characteristics. What comes out of this literature in various versions, as summarised by Calmfors and Driffil (1988), is that the speed of response or adjustment of collective bargaining to new economic conditions depends on the degree of centralisation at the collective bargaining structures. The underlying rationale is that large and centralised unions which bargain at the national level can, and do, take into account the macroeconomic consequences of their behaviour and in a sense become partners in macroeconomic policy- making so as to minimise the costs of inflation and unemployment.

However, Calmfors and Driffil (1988) challenged the literature which asserts that the more corporatist is an economy the better is its economic performance. And they argued that both extremes, namely the highly centralised and the highly decentralised systems of wage setting work best as far as the macroeconomic performance is concerned. The rationale underlying this argument is that in highly centralised systems of wage setting tradeunions enjoying relatively extensive market power, consider the macroeconomic consequences of their bargaining behaviour, while in highly decentralised systems of wage setting unions have a relatively limited market power and thus they cannot delay the

necessary adjustments in the labour market.

A study by OECD (1987) discussed and emphasised the importance of bargaining structures for economic performance and labour market flexibility. After a period, in the late 1970s and early 1980s, of widespread treating of trade-unions as scapegoats, the study by OECD (1987: 140) proposed a more balanced view and recognised that "in no OECD country have labour markets functioned outside the framework of a socially structured system of industrial relations" and that "it may be taken as a premise that collective bargaining will remain at the centre of labour market in OECD countries". From this point of view the policy implication is that "a durable improvement in the wage setting process hinges on the evolution of labour market institutions and in particular of collective bargaining" (OECD 1987:41).

In the literature on industrial relations and economic performance, Greece is a case hardly examined. This is largely due to problems regarding data availability. However, this does not mean that questions regarding the Greek system of industrial relations have not been raised by the parties concerned and policy-makers. Currently three questions are debated: compulsory arbitration for dispute resolution in collective bargaining, incomes policy and the system of automatic indexation, and employment regulation and protection and the legislation on collective dismissals. The criticism on these labour market issues has been growing during the last two decades.

With regard to arbitration since the mid-1970s, the system of

compulsory arbitration has been the frequent target of criticism mainly by the labour side. Political parties have repeatedly promised to reform this system. During the 1980s, throughout the course of Socialist governments, successive Ministers of Labour announced its reform, which never occurred. Currently new legislation regarding the system of arbitration is to be discussed in the Greek parliament. Despite the criticism, the system of collective bargaining in Greece, and especially compulsory arbitration, have not been sufficiently studied. Not even the criteria for the evaluation of its performance have been spelled out. This is one reason for why the will for reforms is not followed by clear ideas concerning the system which should be adopted.

Formal incomes policies were first adopted in 1975. During the 1960s, the Greek economy enjoyed low rates of inflation, with the annual average of the 1960-70 period being below 2.0 per cent. Inflation became an apparent and severe problem during the mid-1970s, and, since then has been fluctuating at double-Those persistently high levels of inflation digit levels. successive governments to undertake formal incomes obliged policies for controlling cost-push and demand-led inflationary pressures. Since 1975 there has been no period without formal incomes policy in force.

Apart from being a matter of political debate, incomes policy has been the target of systematic criticism in the literature. It is noteworthy that the first major and systematic criticism of incomes policy in Greece referred to its distributional aspects

and effects. Pavlopoulos (1987) criticised incomes policy adopted over the decade 1975-85 because they caused an increased labour share in national income and led real wages to grow faster than productivity. He concluded that the conditions under which the dramatic increase of the labour share in the national income occurred, indicate that trade-unions, under the proper institutional arrangements, can increase the labour share in national income. But, he argued, such an increase was not consistent with and was detrimental to economic growth. However, Pavlopoulos did not discuss the institutional arrangements regarding trade-unions, under which the increase occurred.

Kioulafas (1987) addressed another aspect of incomes policies, namely their effects upon wage differentials among individuals and concluded that the incomes policy adopted during the decade 1975-85 narrowed the wage differentials between individuals of different skills, education, and experience and led, contrary to the varying human capital characteristics of the labour force, to a very egalitarian wage structure.

Since the early 1980s, but mainly since the 1985-87 stabilisation programme, the Greek labour market has been the frequent target of criticism for its lack of flexibility. Two main policy issues concerning wage determination and employment adjustment are central in the debate. Both are related to the key aspects of labour market flexibility. First, there is the issue of indexation in wage determination. Second, there is the issue of controls upon collective dismissals. However, the flexibility

debate has not been properly and explicitly set up upon either evidence or, at least, coherent normative approaches of the Greek labour market. It has been, more or less, for both pro- and antiflexibility sides, set in rather ideological terms. Flexibility has been proposed as a panacea, without being quite clear how it is linked with the workings of the labour market in the Greek economy.

The socialist governments of the period 1982-88 always avoided being explicit on the flexibility issue and, indeed, during the 1985-87 economic stabilisation programme, backed off from a labour market flexibility drive because it was considered too costly in political terms. The opposition parties of the 1982-88 period, have not been clear enough on their views about flexibility in the labour market. In fact, the Conservative party has adopted the phraseology of the flexibility drive as proposed and implemented in Western industrialised market economies, mainly the U.K. version. The Communist party has been opposing altogether any policy towards labour market flexibility. Unions too have opposed any flexibility measures as an offensive against themselves and employees.

With regard to flexibility in the labour market there are two main questions. First has there been any flexibility in relative wages in the Greek manufacturing sector and, in turn, has this flexibility, or the lack of it, been a solution for problems arising from wage determination developments, or flexibility has been part of the problems. Second, there is the choice concerning the policy mix for moderation in wage determination on the one

hand, and job security on the other. Both the partial indexation and the employment regulation policies seem to provide to the typical employee a certain degree of income and job security. The policy issue is whether the menu should be modified, for instance, either radically towards de-indexation and more flexible legislation for collective dismissals, or in a rather corporatist way towards the use of job security as a quid pro quo for moderate wage determination.

In other words, the question of the relationship between industrial relations and economic performance, wage flexibility in particular, has been raised on different occasions by academics, politicians, trade-unions and employers associations. Furthermore, in the late 1980s policies and measures for deregulation of the capital, products and labour market have been initiated. But, the reform of the Greek labour market and the national system of industrial relations has neither been at the top of the agenda nor implemented, despite the accumulating and intensifying criticism regarding their long-lasting characteristics and wage and employment policies persistently used by successive governments.

To put the varying views on labour market and industrial relations issues, as well as the developments in Greek manufacturing over the last two decades, in an international and comparative industrial relations perspective, the distinction proposed by OECD (1987) between two quite different approaches to collective bargaining, namely the highly centralised and the

highly decentralised, provides a fruitful starting point.

The OECD (1987:41) study concluded that both highly decentralised bargaining, as in the U.S. and in Japan, and highly centralised bargaining, as in Austria, Sweden, Norway and Germany, have proved reasonably successful in obtaining "sensible wage outcomes" while labour market outcomes have been consistently poorer in those countries where collective bargaining structures fall between these extremes, as in the U.K., France, Italy and Belgium. Calmfors and Driffil (1988) arrived at the same conclusion.

On the basis of the policy orientation towards more flexibility in labour markets, the OECD study suggested that a greater degree of decentralisation of collective bargaining to the enterprise level would be the preferable policy stance for countries aiming at flexibility and structural adjustment. To secure that decentralised bargaining would not aggravate inflationary pressures and would not be accompanied by widespread industrial conflicts, the OECD study spelled out some preconditions regarding the legal framework and employers solidarity in the bargaining process.

Given these definitions and policy prescriptions, the Greek system of industrial relations cannot be easily classified into one of the three above mentioned categories. Is the structure of collective bargaining in Greece centralised, decentralised or does it fall between these two extremes?

Prima facie, it could be classified into the group centralised

collective bargaining. But developments in manufacturing industrial relations after 1975 indicate a trend towards decentralisation of collective bargaining to the company and plant level. Therefore, the Greek experience provides a basis for examining, in the context of governmental regulation of the labour market and changing institutions and processes of collective bargaining, the question of relative wage flexibility in manufacturing. This thesis examines exactly these questions.

The thesis consists of five main chapters. Chapters 2 and 3 examine procedural and institutional issues. Chapter 2 discusses the main characteristics of collective bargaining procedures and structures in Greece and examines the evolution of the formal bargaining structure. Chapter 3 discusses the role of compulsory arbitration and strike activity in the Greek system of industrial relations, examines the compatibility of compulsory arbitration with collective bargaining and analyses institutional developments associated with the strike behaviour of salary and wage earners.

Chapter 4 discusses the governmental involvement in wage determination and employment regulation. It analyses the rationales and the mechanics of incomes policies adopted in Greece since 1975 and examines the variability in the wage structure in Greek manufacturing. It also presents the existing legislation and the mechanics for employment regulation and examines the evolution of employment in the Greek manufacturing

sector.

Chapter 5 examines the extent to which the Greek manufacturing wage structure has been responsive either to competitive influences or to industry-specific conditions. Moreover, as these two different types of flexibility may exert opposite influence upon labour allocation and employment growth, this chapter discusses the extent to which such responsiveness, if any, has been employment enhancing or not. The examination of flexibility in relative wages is performed at the disaggregated level of twodigit industrial branches in manufacturing.

Chapter 6 examines the level and the direction of influence exerted by the norm-based and the indexation-based incomes policy the inter-industry wage structure in the upon Greek manufacturing sector over the period 1975-88 as well as the influence of aggregate inflation, unemployment and variation in The inter-industry wage structure in Greek strike activity. manufacturing is described by the coefficient of variation of average hourly pay of male workers across twenty two-digit branches. This analysis can offer an account of the wageinflation process at the disaggregated level and its association with the rise of the informal collective bargaining in the Greek manufacturing sector.

Chapter 7 summarises the findings of the thesis, draws the conclusions and discusses the policy implications. The main findings of the thesis are the following.

With regard to bargaining structures and procedures the main

findings of the thesis suggest that within the formal system of collective bargaining, despite the extensive legalistic control exerted by successive governments via the Ministry of Labour, there were tendencies towards more decentralised bargaining. These tendencies were noticeable even in the Greek manufacturing sector where collective bargaining at the company and the plant level is not permitted.

As far as the role of compulsory arbitration is concerned it is argued that arbitration has been incompatible with collective bargaining not because of its compulsory character but because of its lack of independence from the Ministry of Labour, which used the machinery of compulsory arbitration for implementing the governmental incomes policy. Moreover, arbitration is found unsuccessful in controlling industrial conflicts, the rise of which, especially in manufacturing, led to the creation of an informal system of collective bargaining at the plant level. Therefore, decentralisation and fragmentation of collective bargaining has been a dominant feature of manufacturing industrial relations after 1975.

The rise in strike activity and the set up of the informal system of collective bargaining after 1975, and throughout the period 1975-88, coincided with the use of formal incomes policies largely aimed to compress the wage structure. Especially the egalitarian indexation policy of the period 1982-88 was clearly designed to narrow wage differentials. However, variability in manufacturing wages was observed throughout the period 1966-88.

In the same period, contrary to what happened in the majority of industrialised market economies, employment in Greek manufacturing has been expanding, faster in 1967-74 and slower since then.

In the context of substantial and procedural regulation of the Greek labour market, it is found that this regulation permitted only episodes of competitive flexibility in relative wages, while the observed variability was mainly associated with noncompetitive flexibility caused by industry-specific factors such as changes in productivity and strike activity. The decentralised and informal collective bargaining is considered as the main institutional precondition favouring non-competitive flexibility in relative wages. Furthermore, egalitarian incomes policies, as they insulated wage formation in branches with below-average changes in productivity from their relative productivity performance, caused inter-industry asymmetries between changes in wages and changes in productivity which, over the period 1977-88, were employment hindering.

The examination of the influence exerted by the norm-based and the indexation incomes policies leads to the conclusion that since 1975, in the Greek manufacturing sector and in the context of collective bargaining, incomes policies dominated the variation in the inter-industry wage structure. However, the influence of the informal system of collective bargaining is noticeable, as it caused the business cycle variables, i.e. inflation and unemployment, to affect the wage structure in an unconventional way, by making them to work in favour of the high-

wage and high-bargaining power industrial branches. The influence exerted by strike activity was not homogeneous as not every strike is successful and strike-proneness does not always indicate bargaining power.

Overall, in the context of a regulated system of industrial relations and given the continuous use of incomes policy, in the Greek manufacturing sector, the move towards decentralised and fragmented collective bargaining, observed after 1975, caused a certain degree of flexibility in relative wages which was associated with industry-specific characteristics. Bargaining structures and relative wages have not been as rigid and ossified as an overview of the governmental policies for procedural and substantial regulation may imply.

However, the prevailing decentralisation and the observed flexibility were not part of a solution similar to that proposed by the OECD study. They rather were part of a twofold problem regarding first, the harmonisation of the formal system of collective bargaining with the real bargaining structures and procedures and second, the need for moving from permanent incomes policies to wage determination based more upon voluntary collective bargaining institutions and processes.

CHAPTER 2: COLLECTIVE BARGAINING: INSTITUTIONS AND PROCESSES.

2.1. INTRODUCTION AND SUMMARY.

The post-war system of industrial relations in Greece, as far as collective bargaining is concerned, has been based upon Law 3239 introduced in 1955. Excessive legalistic regulation of collective bargaining is one of its main features. Legalistic regulation applies to bargaining levels and units, to the coverage of collective agreements and to the qualification of trade-unions for collective bargaining.

The practical consequences of this legalistic regulation upon collective bargaining institutions and processes in Greece have not been sufficiently studied. Not even the criteria for such an examination have been spelled out. This is one reason for why the will for reforms is not followed by clear ideas concerning the system which should be adopted. Conflicting interests between the parties to the bargaining process are expected but it is important for the national system of industrial relations whether the bargaining structures, through which conflicting aspirations as far as wage and employment regulation are concerned, remain stable or not.

This chapter discusses collective bargaining procedures and structures and examines institutional developments associated with the bargaining behaviour of salary and wage earners. The analysis first, deals in general with the dynamics of the formal

system of collective bargaining set up by Law 3239 and, then, focuses upon developments in the manufacturing bargaining structure.

Section 2.2 discusses theoretical and methodological questions about bargaining structures and behaviour. Section 2.3 analyses the Greek formal system of collective bargaining with reference to trade-unions developments and to the characteristics and the evolution of the formal bargaining structure. Section 2.4 discusses the bargaining structure and the formal bargaining procedures in the manufacturing sector. Section 2.5 summarises the findings and presents the conclusions of this chapter.

Our analysis suggests that in the Greek system of industrial relations the formal structure of collective bargaining and the legalistic regulation of formal bargaining procedures have created rather ossified bargaining structures which exclude, especially in manufacturing, collective bargaining at the company or plant level. Nevertheless, during the 1975-81 period, even within the formal system of collective bargaining, pressures and tendencies towards decentralised and fragmented bargaining at the company and plant level occurred.

2.2. COLLECTIVE BARGAINING AND BARGAINING STRUCTURES

2.2.1. Bargaining Structures: Levels and Units.

Collective bargaining is one mode of wage and job regulation. Apart from it, as Flanders (1970:94) pointed out, five other modes can be identified: unilateral regulation by either employers or employees, tripartite regulation, state and social regulation. In the industrialised market economies there has been a long tradition of collective bargaining. At least during the last three decades, collective bargaining has been the dominant mode of wage and job regulation.

Historically, wage earners initiated collective action and organisation and, in most cases, employers begun to form their own collective organisations largely as reaction to growing levels of unionisation and trade-unions activity. As Sisson (1987:11) observed, in most of the industrialised countries, employers came to an accommodation with trade-unions mainly during the first half of this century, with the exceptions of the U.K. and Japan which came to an accommodation earlier and later respectively.

Analytically, bargaining structures and bargaining behaviour consist the two main aspects of collective bargaining. The development of collective bargaining has been mainly based upon the formation of bargaining structures through which trade-unions and employers regulate the employment relationship. While the

parties' behaviour has largely influenced the shape of national bargaining structures, legislation has in many cases played a key role in establishing bargaining structures as central elements of national systems of industrial relations. According to Cordova (1987:308) that was made possible by the incorporation of collective agreements, as a new source of rules and regulation in the field of labour, in the legal system of various countries.

In the literature there is no commonly agreed benchmark for distinguishing and articulating structural and behavioural aspects of collective bargaining, probably because the interactive relationship between them runs in both directions. A recent OECD (1987) study discussed structural aspects as the breadth of bargaining and the depth of collective agreements, as well as behavioural aspects as the constraints upon, and the societal role of, organisations involved in collective bargaining. The OECD study adopted the methodology proposed by Clegg (1976) who had argued that differences in the level at which collective bargaining takes place explain largely international variations in unions membership and structure, strikes, unions government and workplace organisation.

Even when, for methodological reasons, the distinction between bargaining structures and behaviour is made clear, there is no commonly agreed definition either of the concept of bargaining structure or of its elements or dimensions. The emphasis has been put upon varying aspects and dimensions. From this point of view, Windmuller (1987:82) proposed the use of the concept of

collective bargaining structure with reference to bargaining levels and bargaining units. The concept of bargaining levels refers to hierarchical and horizontal layers at which negotiations take place and which, in a sense, form the scaffolding of any bargaining structure. The concept of the bargaining unit comprehends the groupings of employees and employers represented in collective negotiations and subjected to terms embodied in agreements.

Windmuller (1987:81-119), in his comparative assessment of bargaining structures in the industrialised market economies, concluded, first, that great diversity in national bargaining structures is observed, and second, that their basic features, once put in place tend to persist. Nevertheless, the relative stability in bargaining structures does not imply that they do escape pressures for change. Such changes may be initiated by either of the parties to the bargaining process. If these changes become widespread and synchronised, then the national bargaining structure as a whole may become under severe pressures for change towards relative decentralisation or centralisation. Successive waves of such pressures may have long-lasting effects upon the coherence of bargaining structures.

2.2.2. Decentralisation in Bargaining Structures.

Chamberlain (1961:5) suggested that to understand the formation and evolution of bargaining units, beyond their formal and legal sense, especially as far as the workers contribution is

concerned, it is necessary to take into account that "a collective bargaining unit and the power structure within it are products of efforts by workers to achieve their aspirations which frequently collide as well as coincide with each other". In the same line of argument, Weber (1967:20) offered a more technical approach to the parties behaviour towards the bargaining structure. In the context of decision making on the basis of cost-benefit analysis "each party will seek to devise a structure that will maximise its capacity for inflicting real or expected costs on the other party in the course of the bargaining process".

From this point of view and with regard to the relative stability of the bargaining structure in the U.S., where bargaining units are certified by national or state labour relations agencies, Chamberlain (1961:9) argued that it is important to analyse collective bargaining structures in broader terms by taking into account the "interplay between the administratively determined and "appropriate" bargaining unit ... and the informal shifting and usually underground combinations of employees within the unit". Such informal shifting along with the administratively determined levels and units consist what we could designate as the real bargaining structure.

Relative long-run stability in many European and the U.S. bargaining structures is observed for most of the post-war period, and sometimes beyond, back in their formative periods. However, according to Clarke (1987:50), over the post-war period, even in the context of long-run relative stability, the levels at

which bargaining is conducted have tended to change in most European countries mainly towards bargaining at the level of the enterprise or plant. Apart from this slow process, there were also periods during which bargaining structures became under severe and systematic pressures which resulted in their further fragmentation.

In the late 1960s and early 1970s in Europe, where bargaining structures have been relatively more centralised than in the U.S., decentralisation became a strong tendency initiated by the shop-floor militancy of the workforce. It was associated with the strike activity at the plant level and characterised by the challenges to the managerial prerogatives (Dubois, 1978). Decentralisation of bargaining structures offered the means for opposing macroeconomic policies of wage restraint and microeconomic policies for rationalisation of industrial relations at the plant level (Soskice, 1978). The French May in 1968 and the Italian Hot Autumn in 1969 were tips of the iceberg, analysed in Crouch and Pizzorno (1978), which created major disturbances in French and Italian bargaining structures.

By contrast to the labour initiated widespread move towards decentralisation of bargaining structures, observed in Europe in the late 1960s and early 1970s, in the 1980s bargaining structures became also under pressure but the tendency from country to country as well as within countries was not homogeneous. Both moves towards decentralisation and centralisation were observed. Decentralisation of the bargaining
structure has again entered the agenda of industrial relations as part of the flexibility drive which has been initiated by employers and governments. An ILO (1984) review of developments in collective bargaining noticed the employers' interest in lowering the level of bargaining but concluded that this could hardly be generalised to all European countries. In a more recent overview Baglioni (1989) concluded that in Europe, while collective regulation of industrial relations remains central, there has been a general trend towards decentralisation of collective bargaining which often reflects employers decisions and initiatives.

The role of governments regarding changes in bargaining structures varies according to their policies. Incomes policies, to increase their chance for success, require centralised collective bargaining structures. Deregulation policies aimed at increased flexibility in the labour market favour decentralised bargaining structures. Therefore, the variety of policies adopted by governments have not contributed to the move towards a single pattern of bargaining structures.

However, the OECD (1987) study, even after concluding that decentralised bargaining provides a setting in which the conditions and prospects of individual enterprises bear on wage outcomes and that decentralisation would contribute to smoother adjustment to changing patterns of labour demand, remains sceptical about the possible benefits of decentralised bargaining as it recognised that both theoretical analysis and the post-war history of industrial relations do not support the inference that

more decentralised systems always function better in the sense of what OECD perceives as "economically sensible wage outcomes".

2.2.3. Fragmentation and Coherence in Bargaining Structures.

Bargaining structures, considered in a comparative perspective, have been brought at the forefront of the research and debate macroeconomic performance regarding and labour market characteristics in industrialised market economies. The alleged relatively better unemployment performance of the corporatist mode of wage and job regulation has been the theme of both normative and positive approaches. This literature has unfolded on the basis of the concept of corporatism which is mainly developed in the context of the political theory and sociology. The common denominator of all definitions of corporatism is the concept of centralised bargaining structure. Indeed, in the comparative research on industrial relations and macroeconomic performance, centralisation in the wage setting is the cornerstone of the corporatist model.

Calmfors and Driffil (1988) summarised varying definitions of corporatism as well as the variations in the specification of centralisation in wage setting. For instance, Schmitter (1981) emphasised centralisation of the trade-unions structure. Blyth (1979) focused upon centralisation with regard to levels of bargaining and the existence of join central bodies to which employers and unions participate. Cameron (1984) emphasised the centralisation of trade-unionism along with the unionisation

density in the economy. Bruno and Sachs (1985) emphasised centralisation in bargaining structures and the existence of consensus between labour and employers. But in all cases the ranking of countries, according to their degree of corporatism, was very similar.

To these specifications Calmfors and Driffil added their own emphasising the extent of inter-union and inter-employer cooperation in wage bargaining. They preferred to focus upon the behavioural rather than the formal context of wage setting. Their specification of centralisation and corporatism is similar to that proposed by Lash (1985), and used by Newell and Symons (1987), which emphasised both the tripartite character of the corporatist institutions as well as their policy purposes for high levels of employment through co-ordination and moderation in wage demands. Tarantelli (1984) also proposed an analogous specification of centralised industrial relations.

Driffil (1988) basis Calmfors and on the of their conceptualisation of corporatism, challenged the literature which asserts that the more corporatist is an economy the better is its economic performance and criticised the thesis that centralisation at the bargaining structure is that component of corporatism which may explain the relation between corporatism and economic performance. They argued that both highly centralised and highly decentralised systems of wage setting work best as far as the macroeconomic performance is concerned.

A similar argument was made in the OECD (1987:119-151) study

which adopted an approach to industrial relations in OECD countries beyond the dichotomy of corporatist and non-corporatist systems and took into account both structural and behavioural aspects of collective bargaining. That study concluded that it is not simply the degree of centralisation but rather the degree of coherence of bargaining structures, i.e. the lack of multiunion bargaining and overlapping bargaining structures, that has a major influence on how labour markets adjust.

2.2.4. Summary.

To summarise, in the industrialised market economies collective bargaining has been a dominant mode of wage and job regulation. Bargaining structures, consisting of various levels and units, provide the basis for collective bargaining activities. However, the behaviour of both parties to the bargaining process may transform bargaining structures. A tendency towards more decentralised bargaining has been observed in European industrial relations. Beyond the dichotomy between centralised and decentralised bargaining structures, their stability or change are important regarding whether or not they lead to further coherence or fragmentation. Coherence or fragmentation may be observed in both corporatist and highly decentralised bargaining structures.

2.3. COLLECTIVE BARGAINING AND THE FORMAL BARGAINING STRUCTURE IN GREECE.

2.3.1. Procedures, Levels and Units of Collective Bargaining.

Greece belongs to those countries where legislation plays a central role in the regulation of industrial relations. Progressive legislation was introduced in the early decades of the century and when in 1948 the ILO studied the Greek labour law observed that legislation prior to the dictatorship period (1936-40) compared quite favourably with that of most other countries (ILO, 1949:16,221). However, it is not only the presence of legislation and its content but, indeed, its application for the regulation of procedural and/or substantial issues in industrial relations which determines whether it is satisfactory or not.

In post-war Greece, new labour legislation has been introduced which reassured and strengthened the legalistic character of the system of industrial relations. Since 1955 industrial relations have been dominated by Law 3239 which set up the institutional framework for collective bargaining. In fact, the system of collective bargaining implemented by Law 3239 was not completely new. It was rather similar to that introduced in 1954 by Law 3086. Law 3239 is still in force and, indeed, consists a form of strong state regulation of collective bargaining procedures. On the basis of the definitions proposed by Flanders (1970:94)

regarding the modes of wage and job regulation, the Greek formal system seems to be an hybrid form of collective bargaining subject to direct state regulation.

According to Law 3239, collective agreements should be in written form and must be signed by representatives of the bargaining parties concerned. Then, to be ratified and to be given lawful enforcement, collective agreements are required to pass through formal processes which involve the Ministry of Labour. They should be submitted to, and registered in, the Ministry of Labour in order to be published in the Government Gazette (article 2.2). Collective agreements take effect on publication unless stated otherwise in them (art.2.3).

The parties to collective bargaining entitled to conclude collective agreements are confined to trade-unions and employers' associations which already have acquired legal status as entities representing strictly professional interests of sections of employees and employers respectively (art.1.2). There are also restrictions concerning the number of trade-unions which may qualify for collective bargaining (art.23.1). In 1969 the Legislative Decree 186 (art.1), clearly influenced by the French labour law, introduced the notion of the most representative and the simply representative union. In each sector of employment, broadly defined in terms of industry, occupation and/or region, there could be only two representative tradeunions entitled to bargain and to contract collective agreements. Any other trade-unions active in the same jurisdiction are not entitled to bargain and to conclude agreements.

With respect to the coverage of collective agreements, according to Law 3239, there are two regimes. Collective agreements can be either rulings binding solely the members of the parties involved in the bargaining process or binding rulings across the board for the industry, occupation or region concerned. In principle, and in most cases, collective agreements apply to, and are legally binding for, members of the signatories trade-unions and employers' associations (art.5.1). However, there can be cases of collective agreements with extended coverage.

First, when a collective agreement is concluded and signed under the auspices, and in the presence, of the Minister of Labour or his/her representative, its coverage is extended to every employee and employer in the sector, branch, occupation or region concerned (art.5.3). Second, a collective agreement contracted by the most representative trade-union covers also the simply representative union, i.e. the second trade-union entitled to bargain and to contract collective agreements in the jurisdiction concerned, if the second union did not let be known its disagreement with the collective agreement contracted by the most representative trade-union (Legislative Decree 186/1969, art.4.1). Third, the Minister of Labour can extend the coverage of a collective agreement to the entire number of employers of a region, if in the region concerned the employers' association which contracted it employs more than 60 per cent of the sectoral workforce (art.5.2).

The minimum and typical duration of collective agreements is

fixed to one year. Before the introduction of the Legislative Decree 186 in 1969 the minimum period was six months. But the typical duration has always been one year. After the one year period, collective agreements are terminated as one of the contracting parties, normally the trade-union, notifies the other of its intention to terminate the existing agreement and to negotiate a new one (art.4.2). This implies that a new bargaining round starts. Collective agreements are often signed months after expiration of the contract and are applied retrospectively if it is so agreed.

However, it is possible for a collective agreement to be terminated before its normal duration and new bargaining processes to be opened if there have been substantial changes in the conditions under which it was concluded. Arbitration courts are empowered to decide whether such changes occurred and new bargaining process should be allowed (art.15.2). Thus, the life of collective agreements may be shortened. But it can also be prolonged. The Minister of Labour is entitled to extend the validity of a collective agreement, for a maximum period of twelve months, even when one of the contracting parties or both of them have notified their intention to terminate it (art.4.3).

Law 3239 (art.7.1) has also defined clearly the levels and units of collective bargaining. Collective bargaining cannot take place at any level and/or between any unit of employers and employees. There can be four categories of annually signed collective agreements. It is noteworthy that these four

categories, and the relevant bargaining levels and procedures, can be traced back to the provisions of the Legislation Act 1367 introduced in 1938, which, after a turbulent mid-war period, aimed, for the first time ever in Greece, to institutionalise collective bargaining and industrial relations.

The four categories of annually signed collective agreements are the following:

i) The National General Collective Agreement which establishes minimum wages for men, women and apprentices at national level (art.7.2). It is concluded and signed by the General Confederation of Labour (GSEE) and the employers' Confederations (SEV and GSEVE are the principal ones representing large-scale manufacturing and handicraft employers respectively). In Greece, as in many other European countries, there is legal extension of the terms of the National General Collective Agreement to non-union firms.

ii) The National Collective Agreements by Occupation which cover specific occupational groups at national level. They are concluded and signed by the occupational Federations of employees and the relevant employers' Federations (art.7.3).

iii) The Local Collective Agreements by Occupation which cover specific occupational groups at local or regional level, when these groups are not covered by National Occupational Collective Agreements. They are concluded and signed by the regional occupational Federations and the relevant employers' Federations or local representatives of national Federations of employers (art.7.4).

iv) The Special Collective Agreements which cover employees of one single company. According to Law 3239 single employers and their employees, or part of them, have no right to bargain and to reach collective agreements at the company or the plant level. Collective agreements, in order to be lawful, need as a prerequisite at least two employers, i.e. an association of them. The Special Collective Agreements are an exception. They are permitted only for clearly stated organisations, usually not covered by other categories of collective agreements, such as banks, local government, public utilities and corporations which belong, in a wide sense, to the public sector, and employ labour under private employment contracts (art.7.5).

These four categories of collective agreements are not of equal importance. By custom, every other collective agreement is concluded and signed after the National General. The National General Collective Agreement operates as the guideline for the agreements contracted at lower levels of the bargaining hierarchy. This is partly a symptom of the relatively centralised structure of the formal system of collective bargaining.

In addition to this extensive regulation of procedural aspects of collective bargaining, to protect the national interest, the Ministry of Co-ordination and the Ministry of Labour were given the right to examine, ratify, amend or reject collective agreements reached voluntarily by trade-unions and employers (and compulsory arbitration awards) so as to secure that they

remain in line with the governmental economic and social policy (art.20.2).

The rationale for this additional control over substantial aspects of collective bargaining was based upon the view that under the then prevailing economic conditions (for more than a decade wages and salaries did not catch up with the rising cost of living and the gap between prices and wages was increasing geometrically), the lack of any control of collective bargaining procedures would have been detrimental for wage and salary earners. These controls were designed, as argued in the introductory report of Law 3239, first, so as to protect salary and wage earners and second, so that living standards and the political regime be maintained without social policy affecting the economic development.

This right to amend and reject collective agreements (and compulsory arbitration awards) was slightly modified in 1957 by Law 3755 (art.8). From then on they had not the right to amend and reject collective agreements concerning increases in nominal wages not exceeding the annual rate of 3 per cent. This, it was argued in the introductory report of Law 3755, was due to "the general conditions of the national economy" which would not permit a higher limit. The 3 per cent limit was considered as a good start for relaxing the controls when economic development would enable the government to drop altogether the right to control collective agreements and compulsory arbitration awards.

In fact, this right was dropped in 1974 by the Legislative Decree

73 (art.2.1). But this did not mean that the Ministry of Labour was deprived by the right to control procedural and legal aspects of collective agreements. The same decree (art.2.2) ruled that the Minister of Labour maintained the right to control the legality of collective agreements and compulsory arbitration awards as well as to ask the parties to the bargaining process or the arbitration court to re-examine the agreement or award concerned.

2.3.2. Bargaining Structure and Trade-Unions.

The legalistic institutionalisation of collective bargaining, by fixing the bargaining levels and the bargaining units designated to contract collective agreements, has probably influenced the structure of trade-unionism in Greece. This is because employees intending to remain law-abiding and to undertake lawful collective activity, have had to conform themselves to bargaining levels, units, forms and procedures defined by the legislation. This is not to say that legislation is the ultimate determinant of both the bargaining and the trade-unions structure. But, at least, it has set up limits to the structure through which employees have been allowed to undertake collective activities concerning wage determination and regulation of industrial relations.

By prohibiting bargaining at the company and the plant level, the legislation has enhanced the rather centralised collective bargaining of the occupation-based and industry-based unionism.

This type of bargaining, especially the occupation-based, had been a preeminent tradition in Greek industrial relations over the first half of the century (Seferis, 1976). The bargaining structure and the trade-union structure favoured by these legal provisions can be represented as a pyramid-shaped one, which consists of three separate levels, as shown in Figure 2.3.1. The nearest to the basis of the pyramid, the more heterogeneous becomes the structure of trade-unionism in Greece. However, the formal bargaining structure appears less complex and more coherent than the trade-union structure, as the right to bargain is restricted to certain levels of the hierarchy.

The primary level contains sectional, craft, company and general unions. None of them is entitled to bargain and to contract collective agreements. The secondary level contains both Federations, based upon occupational or industrial divisions of the labour force, and Labour Centres which are organised according to the administrative division of the country and include primary unions of a certain region. Nevertheless, and this is important for the shape of both the bargaining structure and the trade-unions structure, only Federations are entitled to bargain and to contract collective agreements. In other words, within the formal system of wage determination, Federations have been given a central role in the function of the bargaining structure. The tertiary level contains the General Confederation of Labour (GSEE) which is recognised by the Law 3239 (art.7.2) as the sole representative organisation entitled to bargain and to conclude the annual National General Collective Agreement.





Source: GSEE

What this representation means to do is to provide a basis for drawing the limits of the legislated formal system of collective bargaining. These limits refer first, to the formal bargaining structure, that is the certified levels and units of collective bargaining, and second, to the structure of tradeunions with regard to the formal bargaining levels. While every union has a place in the pyramid-shaped structure, not all of them consist entities recognised as bargaining units. The great majority of primary level unions are not considered by the law as authorised bargaining units. Consequently, any bargaining processes evolving out of the national, federal or specific company levels, are considered neither as part of the formal bargaining structure nor as lawful.

Apart from the exclusion of primary level unions from the formal system of collective bargaining, within the formal system itself chronic problems have been encountered regarding the recognition of trade-unions which can qualify and be entitled to contract collective agreements. First, this chronic perplexity has to do with the definition of the most representative and the simply representative trade-unions for each jurisdiction. When there is only one union alleging representation and coverage of a certain industrial branch or occupation at the national or regional level, no question is raised on whether or not it is representative. But, in many cases during the post-war period, there have been more than two trade-unions claiming representation of the same section of the labour force.

Second, the structure of representation of the trade-unions in Figure 2.3.1 does not mean that each primary union belongs to one Federation, or that each Federation is a member of the General Confederation of Labour. Such a coherence is not observed. Indeed, trade-unions, and especially Federations, are not organised strictly according to industrial and occupational divisions of the labour force. There are cases of Federations which are rather general and include primary unions from various industrial branches and occupations. The Federation of Manufacturing Enterprise Unions (OVES), founded in 1979, is a recent and typical case of inter-industry Federation.

In this context, inter-union and intra-union disputes over recognition rights have been a permanent feature of trade-union developments. As many scholars of Greek trade-unions have pointed out organisational problems (exclusions, misrepresentations, cancellations etc.) have always plagued trade-unionism in Greece (Jechinis, 1967, Fakiolas, 1978, Katsanevas, 1984). Most tradeunions have been run with important political interference (Katsanevas, 1984:227-248). Behind these organisational problems, which have been reflected in the bargaining structure, lies the fact that the Greek system of industrial relations has been dominated by a strong adversarial tradition directly linked to political developments, divisions and rivalries (Fakiolas, 1978:340-392, Ioannou, 1989:54-68).

To understand how political divisions have been interwoven with trade-union developments we must take into account that, over the

post-war period in Greece, the official trade-unionism has been dominated by pro-governmental officers appointed, directly or not, via judicial arrangements, by the Ministry of Labour. Katsanevas (1984:216-218) described this phenomenon, observed not only at the top level of the General Confederation of Labour (GSEE) but also at lower levels of the trade-unions structure and hierarchy, as "paternalistic unionism".

As Koukoules (1988) has shown, on the basis of the article 69 of the Civil Code adopted in the late 1940s, the judiciary has had the right to, and the task of, appointing new executives at the General Confederation of Labour as well as at every Federation, whenever representatives of a single trade-union claimed that the executive committee of the General Confederation of Labour, or of the Federation in question, is not representative of the rank and file and applied for the appointment of a new executive committee. During the post-war period successive governments, by using this legislation and machinery, and via the referrals of internal trade-union conflicts by pro-governmental trade-unions to the judiciary, have been appointing executives of their own political persuasion.

In 1967-74 there was a shake-out in trade-unionism and only the then pro-governmental trade-union officers were left unimpaired to perform the role of representing the workforce. The same happened, to a lesser extent, in 1975-81. But even then, the majority of leaderships in trade-unions entitled to bargain, to conclude collective agreements and to invoke the compulsory arbitration, remained pro-governmental, or at least not

affiliated to any of the left-wing opposition parties. For instance in the 18th Congress of GSEE in 1976 only 29 per cent of the representatives (167 out of 569) were affiliated to the socialist or the communist parties and held 31 per cent of seats (11 out of 35) in the then elected executive committee (Fakiolas, 1978:343-344, Katsanevas, 1984:232).

The same pattern of relations between trade-unions and governments persisted in the period 1982-88, although the tradition of appointing pro-governmental trade-union leaderships was limited to the top level of the trade-union hierarchy, i.e. the General Confederation of Labour. When, after the imposition of the 1985-87 stabilisation program, the ruling party lost the majority in the General Confederation of Labour (GSEE), progovernmental officers were appointed. Until early in 1989, the General Confederation of Labour was led by pro-governmental trade-unionists who, in line with the long national tradition, were appointees.

Behind these synchronised changes in politics and trade-union leaderships and the persisting pattern of political affiliation of trade-union leaderships to the party in power, lies the fact that the official trade-unionism has been always related to, and depended on, the Ministry of Labour in a relationship dominated by two factors. First, the prevailing role of the judiciary in regulating the internal affairs of Greek trade-unions and second, the financial dependence of most trade-unions on rather discretionary funding by the Ministry of Labour. Greek trade-

unions are not financially independent. In practice they do not rely upon the contributions of their members but are financed by a Ministry of Labour dependent fund (Fakiolas, 1978:180-190, Katsanevas, 1984: 199-215, Koukoules, 1984).

Federations were acting as parties to the bargaining process led by largely appointed leaders and officers financially dependent on the Ministry of Labour. And when questions regarding interunion conflicts about coverage were raised there was the Legislative Decree 186 (art.2-3), introduced in 1969, which empowered ever since the arbitration court to determine the representative trade-unions in each jurisdiction. In 1974 the Legislative Decree 73, by adopting the membership as the single major criterion, clarified the standards for determining the most representative and the simply representative trade-unions. But even their membership claims were not a sound base for selecting the representative unions. In many cases electoral malpractices dominated internal democracy of trade-unions. (Katsanevas, 1984:194-198).

The provisions of Law 3239 which have given the government a central role in collective bargaining, and more than thirty years of experience, indicate that collective bargaining in Greece is different from collective bargaining in the U.S., the U.K. and other industrialised market economies. The difference is both of degree and of kind. First, of degree because of the persistently low union density which only in the early 1980s peaked near 30 per cent (Ioannou, 1988:83). Of course, a union density of 30 per cent is higher than that in the U.S. but its not the level of

union density which makes the difference. It is also the limited coverage of collective bargaining arrangements, let alone that of the annual National General Collective Agreements which establish the national minimum wage. Second, of kind because of the type of relationship between trade-unions, employers and the state, with direct state interventionism in Greek industrial relations being predominant. Surprising as it may seem, trade-union officers largely appointed and, indeed, primarily financed by the Ministry of Labour, have been supposed to be the main actors within the legalistic system of collective bargaining, which also has been thoroughly regulated by the Ministry of Labour and the judiciary.

2.3.3. The Formal Bargaining Structure: Stability or Change?

The legalistic regulation of collective bargaining and the restrictions imposed upon trade-unions create the impression of a rather ossified bargaining structure. However, even in this context there can be some possibility for changes in the formal bargaining structure. Such changes can be caused, at least, by structural changes in the economy and in the relevant composition of the labour force as well as by changes in the political environment within which labour undertakes collective activities. During the post-war period, and mainly during the 1960s and the 1970s, impressive changes occurred in all these three areas setting the scene of Greek industrial relations (Fakiolas, 1978:133-148, Ioannou, 1989:49-53).

To examine whether the formal bargaining structure has been ossified and to discern any changes we analyse the scant information available at the Ministry of Labour for the period since the implementation of the present system of collective bargaining. Table 2.3.1 describes the number of both collective agreements and compulsory arbitration awards registered in the Ministry of Labour. This number depicts the quantity of bargaining units existing within the formal system of collective bargaining across all sectors of the economy. In Table 2.3.1 the density of bargaining units in the labour force, i.e. the number of bargaining units per 100000 employees in employment, is also estimated.

On average, at each side of the bargaining process, there were recorded less than 300 bargaining units, i.e. less than 20 bargaining units per 100000 salary and wage earners. This quantity of bargaining units does not imply an equal number of trade-unions and employers associations. The comparison with the number of trade-unions entitled to contract collective agreements, shown in Figure 2.3.1, indicates a large incongruity between these two sources of information regarding the number of actors in the formal bargaining structure. This incongruity is due to the fact that, for two reasons, there is not a one-to-one relationship between trade-unions and employers associations on the one hand and bargaining units on the other.

First, every trade-union, or Federation in particular, does not contract only one annual collective agreement. It can contract

more than one according to its membership and its coverage of distinct jurisdictions in the labour market. Second, in every jurisdiction there can be two representative trade-unions contracting, at least, two different collective agreements, different as far as their content and coverage is concerned. Therefore, there is not direct relationship between the number of bargaining units and the number of trade-unions involved in collective bargaining. The same asymmetry applies to the employers side too.

Despite the annual fluctuation observed over the period 1955-88, from Table 2.3.1 becomes clear that in the 1970s there was a strong increasing trend in the number of bargaining units acting within the formal bargaining structure. Indeed, during the period 1975-81 the number of bargaining units increased dramatically both in absolute and relative terms. From 1974 with 155 registered bargaining units and a 10.49 density, to 1981 the increase is impressive as there were 563 registered bargaining units, i.e. 33 bargaining units per 100000 salary and wage earners. It is noteworthy that during the period of the military regime, in 1967-74, in comparison with the earlier period 1964-66, the density of bargaining units in the labour market decreased. Then, in 1975-81 the density more than doubled and in 1982-88 decreased again.

The wide annual fluctuations observed in the 1980s are mainly due to, and are explained by, the type of incomes policies adopted during the period 1982-88. The mandatory incomes policies of 1983 and 1986-87 caused formal collective bargaining processes to

freeze. The imposition of a "wage freeze" by decree in 1983 made redundant and unlawful the great majority of annually contracted collective agreements, as well as the annual settlement of disputes through arbitration. In 1983 only 137 collective agreements and awards were registered, while in 1982 the relevant number was 516. Therefore the density of bargaining units decreased from 30 in 1982 to 8 in 1983. For the same reason in 1984 the number of agreements and awards, and consequently the number of bargaining units acting within the formal system of collective bargaining, rose again at its 1982 level. Similar decrease is observed in 1986-87 because of the tight and mandatory incomes policy which was part of the 1985-87 stabilisation program.

The impressive increase in the number of bargaining units during the 1975-81 period indicates, to a certain extent, the growth of collective bargaining procedures in the Greek system of industrial relations. But this is indicated only to a certain extent for two reasons. First, because these estimates include those bargaining units either succeeding or failing to reach collective agreements voluntarily. The disputes of the latter are settled by arbitration courts. Therefore, the resulting increase in bargaining units only in a wide sense can be considered as indicator of collective bargaining growth, i.e. in the sense that arbitration is also part of the bargaining process. Second, because statistics referring to the number of agreements and awards may be misleading if used to hint the extent of employees covered by collective bargaining arrangements. A few industry-

wide or national agreements may carry more weight that a relatively high number of regional or special agreements.

To analyse the composition of the formal bargaining structure, Table 2.3.2 presents the breakdown of the total number of bargaining units by bargaining level and category. This analysis suggests that the formal bargaining structure, as far as the number of bargaining units is concerned, is composed by two major categories, first those units acting at the national industry-wide or occupation-wide level and second, those acting at the special organisation and company level. This percentage composition does not imply any analogy with regard to the coverage of the labour force by collective agreements and arbitration awards of the relevant levels.

The distribution of bargaining units in different levels of the formal bargaining structure indicates the occurrence of changes towards decentralisation and fragmentation in 1975-81 and towards centralisation in 1982-88. Compared with the period 1973-74, the 1975-81 average share of bargaining units acting at the national level for industry-wide or occupation-wide bargaining decreased from 45.63 to 31.53 per cent. Analogous decrease is observed in the average share of bargaining units at the local or regional level. The average share of units acting at the lowest level of collective bargaining from 37.79 in 1973-74, soared at 55.42 per cent in 1975-81. These changes indicate decentralisation in the formal bargaining structure.

But the tendency towards decentralisation and fragmentation was

reversed in 1982-88. Then a move towards centralisation is observed as the average share of bargaining units at the lowest level of the bargaining structure decreased and the average shares of national and regional bargaining increased. This reversed tendency can be explained by the fact that after 1982, under the socialist government, in the context of what it was perceived as democratisation of the Greek trade-union movement, the opposition of the 1975-81 official trade-unionism, which as opposition was mainly active out of the formal system of collective bargaining at the company and plant level, took over the control of the official trade-union structure and tried to re-establish the dominant role of national industry-wide or occupation-based collective bargaining. The incomes policy of the period was also favourable to more centralised collective bargaining.

To discern the sectoral distribution of the increasing number of bargaining units at the special company level, Table 2.3.3 estimates the share of broad sectors of the economy in the total number of special collective agreements (the relevant arbitration awards are not available) registered with the Ministry of Labour in 1974, 1975, 1982 and 1988. We selected these years as they coincide with peaks and troughs in the rapid expansion of special collective agreements. From 1974 to 1975 the number of special collective agreements more than doubled. It doubled again from 1975 to 1982, and then from 1982 to 1988 their number nearly halved.

From Table 2.3.3 we draw the conclusion that special collective agreements and the relevant bargaining units were primarily concentrated in the area of public sector institutions, organisations, corporations and companies employing labour on the basis of private contracts. Only a small percentage of special collective agreements refers to manufacturing industries, and this happens for companies mainly operating under state ownership. Bargaining units at the special organisation or company level are confined by the law to clearly stated sectors of the economy so that formal bargaining at the company level has little to do with manufacturing industries in the private sector.

The composition of the total number of collective agreements in the mid-1970s, as described in Table 2.3.3, indicates the existence of fragmented bargaining even at the level of stateowned organisations and corporations. Its evolution suggests that changes occurred in both directions, i.e. towards further fragmentation or more coherence. From 1975 to 1982 in transport, communications, electricity and water industries the tendency was towards more centralisation and coherence in collective bargaining, while the labour employed under private contracts by Ministries and dependent institutions moved towards more decentralised and fragmented bargaining. However, from 1982 to 1988 in transport and communications a tendency for more fragmented bargaining is observed.

To summarise, in the formal system of industrial relations during the period 1975-81 collective bargaining expanded in terms of multiplied bargaining units. The changing composition of the

formal bargaining structure suggests that the expansion of collective bargaining in the 1970s was primarily caused by the surge of special collective agreements and compulsory arbitration awards bearing on collective bargaining at the company level. In other words, in 1975-81 within the formal system of collective bargaining the significance of industry-wide or occupation-wide bargaining declined. This development indicates a trend towards decentralisation and further fragmentation of the formal bargaining structure. However, in 1982-88 this trend was reversed towards centralisation. The relatively lower density in bargaining units in 1982-88 reflects the influence of two factors, first, the move towards national bargaining and second, the strict incomes policies and the "freeze" imposed upon collective bargaining in 1983 and 1986-87.

2.4. THE FORMAL BARGAINING STRUCTURE IN MANUFACTURING.

.4.1. Restrictions and Evolution.

In the formal bargaining structure across all sectors changes towards more decentralised bargaining occurred in 1975-81, and the significance of industry-wide or occupation-based bargaining at the national level declined. But prima facie this decline does not refer to manufacturing, as within the formal system these types of bargaining are defined by the law as the main levels at which manufacturing collective bargaining can take place. The legal restrictions imposed upon collective bargaining are stronger in the manufacturing sector because the limitation on bargaining levels and units applies mainly in manufacturing and special collective agreements are not normally allowed. From this point of view, the legal restrictions imposed upon collective bargaining may imply a far more ossified bargaining structure in manufacturing than in others sectors of the economy.

Nevertheless, in manufacturing there were more reasons for the bargaining structure to change. While changes in political relations influence equally all sectors across the board, structural changes in the economy and the resulting change in employment structures may have exerted stronger influences upon the manufacturing bargaining structure than in other sectors of the economy because, over the period under consideration, along with the rapid expansion in manufacturing employment, its further

concentration occurred. The number of manufacturing establishments employing more than 100 employees increased from 436 in 1969 to 701 in 1978 and their share in large-scale manufacturing employment (i.e. establishments employing more than 10 employees) increased from 48.1 to 54.2 per cent.

To examine the characteristics and the evolution of the formal bargaining structure in the Greek manufacturing sector we analyse the number of agreements and awards registered with the Ministry of Labour over the period 1954-1983. Unfortunately the records from where we draw these series of information have been discontinued since 1984.

Table 2.4.1 shows the number of manufacturing bargaining units registered within the formal system of collective bargaining and estimates their density in the manufacturing labour force. On average 67 bargaining units were registered over the period 1954-83 and there were 26 bargaining units per 100000 wage and salary earners employed in large scale manufacturing industry. Its evolution over time does not indicate any systematic pattern and In comparison to 1955-60 the density of bargaining units trend. nearly doubled in 1961-66. In 1967-74 decreased and since then remained stable at levels similar to those prevailing in early 1960s. This trajectory can be indication either of decline in the significance of manufacturing collective bargaining or of rapid centralisation in the manufacturing bargaining structure. But the evolution in the composition of the formal manufacturing bargaining structure suggests that the relatively low density of bargaining units in 1975-81, compared to 1961-66, is mainly due

to the further centralisation of the bargaining structure.

The composition of the total number of bargaining units by level over the period 1954-83, presented in Table 2.4.1, shows that the main bulk of manufacturing collective agreements and awards registered with the Ministry of Labour refers to national industry-wide or occupation-based collective agreements. It also indicates the existence of a clear pattern of change in the formal bargaining structure from local and regional collective bargaining to national bargaining. The share of manufacturing bargaining units acting at the national level increased from 22.18 in 1955-60 to 72.58 per cent in 1975-81. By contrast, the share of manufacturing bargaining units acting at the local or regional level decreased from 69.95 in 1955-60 to 11.84 per cent in 1975-81. This trade-off clearly indicates a tendency towards more centralised bargaining. The share of bargaining units at the special company level had also an increasing tendency but remained at low levels.

2.4.2. Regulation of the Formal Bargaining Structure in Manufacturing.

Table 2.4.2 provides a breakdown of the total number of manufacturing bargaining units involved in collective agreements and compulsory arbitration awards according to the seven different procedures of regulation of formal collective bargaining available according to Law 3239. These are i) the voluntarily reached collective agreements, ii) collective

agreements made compulsory for non-members of the signatory parties, iii) compulsory arbitration awards, iv) agreements and awards amended by the Ministry of Labour, v) agreements and awards corrected by the Ministry of Labour, vi) agreements and awards rejected by the Ministry of Labour and vii) agreements and awards of which the duration was extended unilaterally by the Ministry of Labour.

These procedures indicate the degree of voluntarism and compulsion in manufacturing bargaining procedures. The share of collective agreements voluntarily reached and of those with extended coverage represent the area of voluntary collective bargaining while the rest represent the regulated area of the formal system of collective bargaining. Over the period 1955-83 the majority of bargaining units falls into the category of regulated collective bargaining. In 1955-60, i.e. during the period of the implementation of the system, 73.20 per cent of the manufacturing bargaining units were under the direct regulation of the Ministry of Labour. This regulation diminishes in the following periods but remains at levels above 50 per cent.

Certain methods of regulation were used in some periods and abandoned completely in others. In general the use of direct methods of regulation, i.e. amending, correcting, rejecting and extending the duration of collective agreements and awards, decreased from period to period and was replaced by arbitration i.e. the indirect method of collective bargaining regulation.

For instance, column 7 in Table 2.4.2 indicates that the

provisions permitting the extension of collective agreements for one more year was mainly used in the 1960s and till the early 1970s, i.e. over a period of low inflation -if any at all. The same procedure was also used extensively in 1976, probably without much success as in the following years it completely vanished. This practice of extending the duration of collective agreements may be compatible and consistent with periods of low inflation as then there is no strong need new agreements to be contracted for keeping up with inflation. The abandonment of this procedure can be attributed to the rise and persistence of price inflation which made necessary, for wages to keep up, annual bargaining rounds and new collective agreements.

The period of implementation of the formal system of collective bargaining was also the golden era of direct governmental intervention in manufacturing collective bargaining. Column 8 in Table 2.4.2 indicates that in 1955-60 27 per cent of agreements and awards were amended by the Ministry of Labour, and, as already mentioned, 73.20 per cent fall in the category of one way or another regulated settlements. In 1967-74 the degree of intervention for amendment of collective agreements was relatively lower, but the degree of intervention for extending the duration of agreements and awards had its highest share of all periods. In 1967-74 while the total degree of regulation was near the all periods average, if we take into account the political conditions under the military regime, it seems reasonable to infer that the degree of interventionism was quite However, in 1975-81 under conditions of parliamentary high.

democracy and more liberal industrial relations compulsory arbitration became the predominant method for regulation of collective bargaining in manufacturing.

The composition of the voluntary and regulated areas of the formal bargaining structure in manufacturing by category indicates that regulation was mainly exerted at the national level and the local level and not at the level of special collective agreements. Table 2.4.3 provides a breakdown of the voluntary and the regulated areas of the formal bargaining structure by bargaining level. The regulated area refers mainly to national collective bargaining. Regulation of bargaining units active at the company level remained all over the period low while regulation of national collective bargaining in manufacturing was steadily increasing from 18 per cent in 1955-60 to 82 per cent in 1975-81. In fact the focus of regulation moved from local collective bargaining in 1955-66, to national collective bargaining in 1975-83. However, in the voluntary area of the formal manufacturing bargaining structure the share of collective bargaining at the company level is characterised by a steadily increasing trend.

2.4.3. Formal Pressures Towards Decentralisation.

This asymmetry between the voluntary and the regulated areas of the formal bargaining structure in manufacturing is more clearly indicated by Table 2.4.4 where only collective agreements and compulsory arbitration awards are analysed by bargaining level.

Both their average shares and their trends suggest that in the voluntary area occurred relatively more decentralised bargaining and a tendency towards further decentralisation while in the regulated area centralised bargaining dominated and the tendency was towards further centralisation. Consider first the share of compulsory arbitration awards at the national level of bargaining. It increased from 18 per cent in 1955-60 to 83 per cent in 1975-81. The share of collective agreements at the national level increased too, but never exceeded 53 per cent. Consider now the share of voluntarily reached collective agreements at the special company level. It increased from 13 per cent in 1955-60 to 31 per cent in 1975-81 while the relevant share of compulsory arbitration awards was always at levels below 10 per cent and decreasing.

The increasing presence of bargaining units acting at the company level, in the voluntary area of the bargaining structure and contracting special collective agreements reflects the fact that the Ministry of Labour had, in face of the changing structure of manufacturing, to ease the restrictions upon collective bargaining at the plant level in order to reconcile the restrictive legislation on collective bargaining and its machinery with the changing employment structure in manufacturing as well as pressures exerted upon the bargaining structure. The Legislative Decree 186 (art.9) adopted in May 1969 ruled that disputes which have not the typical characteristics of the formal dispute (i.e. level, unit and scope of bargaining) so that to be settled through the formal procedures of collective bargaining or

compulsory arbitration, but concern the employment relationship, could be settled by a committee through a procedure of tripartite collaboration. Members of the committee are the Labour Inspector, the union representative and the employers representative. In the event of settlement, a written statement is signed which, nevertheless, is not binding for the parties involved and their members.

However, later in the mid 1970s the Ministry of Labour had in many cases, following trade-union requests and pressures, to grant to many tripartite collaboration agreements the status of special collective agreements so that to make them binding for the signatory parties. These exceptions mainly applied to some of the largest, in term of employment, companies and plants in Greek manufacturing. Of course, the national industry-wide or occupation-based bargaining remained dominant in the formal bargaining structure in manufacturing but these developments indicate that in 1975-81 even in the context of the formal manufacturing bargaining structure its voluntary area was characterised by a trend towards decentralisation.

To summarise, the formal manufacturing bargaining structure was relatively more ossified than bargaining structure in other parts of the economy because of the restrictions regarding collective bargaining at the company level. Therefore, it mainly consisted of bargaining units acting at the national or regional level and the tendency was from local and regional to national collective bargaining. Indeed, bargaining procedures were excessively

regulated via both direct and indirect procedures. After 1975 direct regulation or collective bargaining by the Ministry of Labour (i.e. amending, correcting, rejecting and extending agreements and awards) gave way to indirect methods of regulation, namely compulsory arbitration. Regulation and especially compulsory arbitration applied to manufacturing collective bargaining at the national level. However, in the voluntary area of the formal bargaining structure in manufacturing a tendency towards decentralised bargaining at the company and plant level occurred after 1975. After all, even in the over-regulated and restricted formal bargaining structure in manufacturing pressures towards decentralisation were observed.
2.4. CONCLUSIONS

Collective bargaining has a long tradition in industrialised market economies where bargaining structures have been changing towards more decentralisation. Contrary to this tradition, in Greece there has been no long tradition of wage setting through collective bargaining. Indeed, both procedural and substantial issues of wage and job regulation are subject to strong governmental intervention as within the formal system of collective bargaining the government maintains the right to intervene on numerous aspects and occasions. Therefore, the system of industrial relations is over-regulated by the existing legislation.

Collective bargaining is restricted to certain levels of the bargaining structure. The formal system of collective bargaining excludes company and plant level collective agreements. These restrictions and the regulation of trade-unions create a rather ossified bargaining structure. Nevertheless, in 1975-81, even within the formal bargaining structure strong tendencies towards decentralised collective bargaining, i.e. at the company level, occurred through the increase of special collective agreements. The procedural arrangements provided by the formal system of collective bargaining, which favours centralised bargaining and excludes decentralised bargaining at the company and plant level, contributed to the fragmentation of the real bargaining structure, especially in the Greek

manufacturing sector.

In the context of the formal system of collective bargaining, wage formation has never been a process independent of direct and indirect state intervention. The legal right of the Ministry of Labour to reject and modify any collective agreement or compulsory arbitration award if considered as not being in line with the governmental policy, dominated wage determination processes, throughout the post-war period.

Table 2.3.1

BARGAINING UNITS IN THE FORMAL BARGAINING STRUCTURE, ALL SECTORS 1955-1988

Year	Numbe	r of Ba	argainin	g Units ^a	Density of Ba	argainin	g Units ^b
	Agre	ements	Awards	Both	Agreements	Awards	Both
19	55:	42	n.a	n.a	3.76	n.a	n.a
19	56:	69	n.a	n.a	6.09	n.a	n.a
19	57:	51	n.a	n.a	4.43	n.a	n.a
19	58:	40	n.a	n.a	3.43	n.a	n.a
19	59:	46	n.a	n.a	3.88	n.a	n.a
19	60:	58	n.a	n.a	4.82	n.a	n.a
19	61:	64	n.a	n.a	5.25	n.a	n.a
19	62:	72	n.a	n.a	5.83	n.a	n.a
19	63:	58	n.a	n.a	4.64	n.a	n.a
19	64:	97	93	190	7.67	7.35	15.02
19	65:	121	93	214	9.45	7.27	16.72
19	66:	105	82	187	8.11	6.33	14.44
19	67:	63	7 9	142	4.81	6.03	10.84
19	68:	63	93	156	4.76	7.02	11.77
19	69:	50	100	150	3.73	7.46	11.20
19	70:	53	67	120	3.91	4.95	8.86
19	71:	46	43	89	3.36	3.14	6.50
19	72:	88	38	126	6.26	2.70	8.96
19	73:	88	60	148	10.96	4.16	15.12
19	74:	70	85	155	4.74	5.75	10.49
19	75:	131	143	274	8.65	9.45	18.10
19	76:	144	120	264	9.29	7.74	17.03
19	77:	146	179	325	9.20	11.29	20.49
19	78:	166	214	380	10.23	13.19	23.43
19	79:	173	247	420	10.43	14.90	25.33
19	80:	220	299	519	12.99	17.65	30.63
19	81:	233	330	563	13.47	19.07	32.54
19	82:	284	232	516	16.57	13.53	30.10
19	83:	57	80	137	3.36	4.71	8.07
19	84:	252	264	516	14.67	15.37	30.03
19	85:	175	167	342	9.89	9.44	19.32
19	86:	44	82	126	2.48	4.62	7.10
19	87:	76	84	160	4.24	4.68	8.92
19	88:	210	83	293	11.64	4.60	16.24
Avera	ge						
1955-	60:	51	n.a	n.a_	4.40	n.a_	n.a_
1961-	66:	86	89 ^C	197 ^C	6.83	6.98 ^C	15.39 ^C
1967-	74:	65	71	136	5.32	5.15	10.47
1975-	81:	173	219	392	10.61	13.33	23.94
1982-	88:	157	142	299	8.98	8.14	17.12

Notes: a. The number of bargaining units refers to both collective agreements and compulsory arbitration awards. b. Density of Bargaining Units = Number of Units per 100000 employees in employment. c. Average of the period 1964-66.

Source: Ministry of Labour.

Table 2.3.2

	BARGAINING	<u>UNITS 1</u> 1973	BY <u>Bargai</u> 1988	NING LEVEL	
Level	General Nat:	Occupa ional	tional Local	Special	Total
Year	\$	ક	*	8	
1973: 1974: 1975: 1976: 1977: 1978: 1979: 1980: 1981: 1982: 1983:	0.00 0.00 1.09 0.38 0.31 0.26 0.48 0.19 0.00 0.19 0.00	49.32 41.94 32.48 31.06 33.54 26.84 32.14 31.98 32.68 40.89 40.15 42.05	15.75 17.42 17.88 19.32 13.54 8.16 9.76 10.02 9.95 15.50 12.41 7.17	34.93 40.65 48.54 49.24 52.62 64.74 57.62 57.80 57.37 43.41 47.45	146 155 274 264 325 380 420 519 563 516 137 516
1984: 1985: 1986: 1987: 1988: Average 1973-74: 1975-81: 1982-88:	0.19 0.29 0.79 0.63 0.34 0.00 0.39 0.35	42.05 45.03 48.41 38.75 43.00 45.63 31.53 42.61	7.17 14.04 23.81 21.25 17.06 16.59 12.66 15.89	50.58 40.64 26.98 39.38 39.59 37.79 55.42 41.15	516 342 126 160 293 301 392 299

Source: Ministry of Labour.

-

Table	2.	3	. 3
-------	----	---	-----

Year	1974	1975	1982	1988
Sector	*	*	*	ક
Manufacturing Electricity and	11.8	4.7	6.4	7.1
Water	8.8	8.2	4.1	4.1
Transport and Communications Banks and	55.9	42.3	23.8	44.9
Insurances	2.9	7.1	4.7	6.1
Ministries	5.9	10.6	19.1	12.3
Local Government	-	10.6	2.9	2.0
Health and				
Social Security	-	-	2.3	2.0
Air Force	-	4.7	17.5	4.1
Education and Research	2.9	5.9	10.5	7.1
Miscellaneous	11.8	5.9	8.7	10.3
Total Number of				
Special Collective Agreements	34	85	172	98

<u>SPECIAL COLLECTIVE AGREEMENTS BY ECONOMIC SECTOR</u> Selected Years

•

Note: Data refer to the number of annual collective agreements concluded during the year in question.

Source: Ministry of Labour

Level National Local Special All Year & & & & & & & & & & & & & & & & & & &	Density ^a b
Year % % % % 1954 25.00 62.50 12.50 100 16 1955 30.30 54.55 15.15 100 33 1956 18.92 78.38 2.70 100 37 1957 n.a n.a n.a n.a n.a 1958 18.60 81.40 0.00 100 43 1959 19.15 70.21 10.64 100 47 1960 23.91 65.22 10.87 100 46 1961 35.71 51.43 12.86 100 70	.b
195425.0062.5012.5010016195530.3054.5515.1510033195618.9278.382.70100371957n.an.an.an.an.a195818.6081.400.0010043195919.1570.2110.6410047196023.9165.2210.8710046196135.71514312.8610070	b
195530.3054.5515.1510033195618.9278.382.70100371957n.an.an.an.an.a195818.6081.400.0010043195919.1570.2110.6410047196023.9165.2210.8710046196135.71514312.8610070	23.38 23.58 35.48 35.60 36.71 53.22
195618.9278.382.70100371957n.an.an.an.an.a195818.6081.400.0010043195919.1570.2110.6410047196023.9165.2210.8710046196135.71514312.8610070	23.38 23.58 35.48 35.60 36.71 53.22
1957n.an.an.an.an.a195818.6081.400.0010043195919.1570.2110.6410047196023.9165.2210.8710046196135.7151.4312.8610070	23.38 23.58 35.48 35.60 36.71 53.22
195818.6081.400.0010043195919.1570.2110.6410047196023.9165.2210.8710046196135.7151.4312.8610070	23.38 23.58 35.48 35.60 36.71 53.22
1959 19.15 70.21 10.64 100 47 1960 23.91 65.22 10.87 100 46 1961 35.71 51 43 12 86 100 70	23.38 23.58 35.48 35.60 36.71 53.22
1960 23.91 65.22 10.87 100 46 1961 35.71 51 43 12.86 100 70	23.58 35.48 35.60 36.71 53.22
1961 35.71 51 43 12 86 100 70	35.48 35.60 36.71 53.22
	35.60 36.71 53.22
1962 35.21 53.52 11.27 100 71	36.71 53.22
1963 36.49 58.11 5.41 100 74	53.22
1964 47.27 41.82 10.91 100 110	
1965 31.73 54.81 13.46 100 104	47.50
1966 40.59 52.48 6.93 100 101	44.46
1967 55.71 35.71 8.57 100 70	31.26
1968 58.62 31.03 10.34 100 58	26.42
1969 50.79 41.27 7.94 100 63	28.11
1970 64.71 25.49 9.80 100 51	20.49
1971 70.21 12.77 17.02 100 47	17.72
1972 70.59 20.59 8.82 100 68	24.37
1973 51.92 25.00 23.08 100 52	17.31
1974 61.29 20.97 17.74 100 62	18.68
1975 71.88 14.06 14.06 100 64	19.06
1976 59.48 17.24 23.28 100 116	33.97
1977 76.83 6.10 17.07 100 82	23.49
1978 77.11 8.43 14.46 100 83	23.61
1979 71.25 10.00 18.75 100 80	22.71
1980 68.42 15.79 15.79 100 95	27.29
1981 83.10 11.27 5.63 100 71	21.23
1982 78.26 14.13 7.61 100 92	27.34
1983 67.86 10.71 21.43 100 28	7.62
Average	
1955-60 22.18 69.95 7.87 41	23.48 ^C
1961-66 37.83 52.03 10.14 88	42.16
1967-74 60.48 26.60 12.92 59	23.05
1975-81 72.58 11.84 15.58 84	24.48
1982-83 73.06 12.42 14.52 60	17.48
Note: a. Number of bargaining units per 100000 sala earners in large-scale manufacturing ind establishments employing more than 10 empl b. There are no data on manufacturing employm c. Average of the 1959-60 period.	ary and wage ustries (i.e. loyees). ment for 1954-58.

Source: Ministry of Labour

PROCEI	DURES O	F <u>Regu</u>	<u>ILATION</u>	<u>of</u> <u>co</u>	LLECTI	VE AGRI	<u>ZEMENTS</u>	AND AWARDS
<u>in the</u>	FORMAL	<u>Byste</u>	em of c	COLLECT:	IVE BA	RGAINII	<u>NG IN M</u>	<u>LANUFACTURING</u>
1954-1983								
Procedui	re							
	[1]	[2]	[3]	[4]	[5]	[6]	[7]	[8]
Year								
1954	81.25	0.00	18.75	0.00	0.00	0.00	0.00	18.75
1955	6.06	0.00	87.88	6.06	0.00	0.00	0.00	93.94
1956	43.24	0.00	16.22	35.14	5.41	0.00	0.00	56.76
1958	12.12	3.03	24.24	36.36	0.00	0.00	24.24	84.85
1959	28.26	8.70	26.09	26.09	0.00	2.17	8.70	63.04
1960	23.91	8.70	15.22	30.43	4.35	2.17	15.22	67.39
1961	62.86	8.57	4.29	20.00	0.00	2.86	1.43	28.57
1962	35.21	5.63	15.49	12.68	1.41	1.41	28.17	59.15
1963	32.43	8.11	22.97	12.16	2.70	0.00	21.62	59.46
1964	46.36	9.09	18.18	4.55	0.91	0.00	20.91	44.55
1965	44.23	9.62	15.38	7.69	0.00	1.92	21.15	46.15
1966	29 70	2 97	21 78	10 89	0.00	0 00	34 65	67.33
1967	51 29	2.27	15 71	10.09	0.00	0.00	17 14	42.86
1969	32 76	5 17	15 52	13 70	0.00	0.00	32 76	62 07
1908	26 51	2 17	10.02	1 50	0.00	0.00	36 51	60 32
1909	47 06	3.17	22.22	11 76	0.00	0.00	27 /5	19 02
1970	47.00	J.92 0 51	9.00	6 20	0.00	0.00	27.45	4 9.02 53 10
1971	20.71	0.01	10.04	0.30	1 47	0.00	27 0/	53.13
1972	59.71	0.02	14./1	7.35	1.4/	2 95	27.54	
1973	51.92	1.92	25.00	7.69	1 (1	3.00	2.02	40.15
1974	50.00	8.00	27.42	9.08	1.01	0.00	3.23	41.94
1975	40.88	0.25	35.94	1.50	0.00	0.00	9.30	40.00
1976	37.07	13./9	31.03	0.86	0.80	1 22	10.30	
1977	25.61	13.41	48.78	0.00	0.00	1.22	10.98	00.98
1978	31.33	20.48	43.78	0.00	0.00	0.00	2.41	40.19
1979	30.00	15.00	52.50	0.00	0.00	0.00	2.50	55.00
1980	32.63	25.20	42.11	0.00	0.00	0.00	0.00	42.11
1981	32.39	23.94	42.25	1.41	0.00	0.00	0.00	43.00
1982	2/.1/	23.91	40.74	2.17	0.00	0.00	0.00	48.91
1983	28.57	7.14	60.71	0.00	0.00	0.00	3.5/	64.29
Average								7 0 00
1955-60	22.72	4.08	33.93	26.82	1.95	0.87	9.63	73.20
1961-66	41.80	7.33	16.35	11.33	0.84	1.03	21.32	50.87
1967-74	43.82	5.31	17.63	8.53	0.39	0.48	23.85	50.88
1975-81	33.70	16.88	42.63	0.55	0.12	0.17	5.95	49.42
1982-83	27.87	15.53	53.73	1.09	0.00	0.00	1.79	56.60
Notes:	[1] Col	lectiv	ve Agro	eements	_	_		-
	[2] Col	lectiv	ve Agro	eements	made	compul	sory fo	or non-members
	[3] Com	pulso	ry Arb	itratio	n Awar	ds.		
	[4] Ag:	reemen	ts an	d Awaro	ls Ame	ended	by the	e Ministry of
	Lab	our.						
	[5] Agr	eement	ts and	Awards	Rejec	teđ		
	[6] Agr	eement	ts and	Awards	Corre	cted.		
	[7] Agr	eement	ts and	Awards	of wh	nich th	ne Mini	stry of Labour
	ext	ended	the d	uration	•			
[8] Share of Regulated settlements: Sum of 3,4,5,6,7.								

Source: Ministry of Labour

			1954-198	3					
	Volun	tary ^a		Requ	Regulated ^b				
Level	National	Local	Special	National	Local	Special			
Year			-			-			
1954	30.77	53.85	15.38	0.00	100.00	0.00			
1955	50.00	0.00	50.00	29.03	58.06	12.90			
1956	25.00	75.00	0.00	14.29	80.95	4.76			
1957	n.a	n.a	n.a	n.a	n.a	n.a			
1958	40.00	60.00	0.00	15.79	84.21	0.00			
1959	35.29	58.82	5.88	10.00	76.67	13.33			
1960	33.33	60.00	6.67	19.35	67.74	12.90			
1961	34.00	52.00	14.00	40.00	50.00	10.00			
1962	34.48	58.62	6.90	35.71	50.00	14.29			
1963	33.33	60.00	6.67	38.64	56.82	4.55			
1964	42.62	42.62	14.75	53.06	40.82	6.12			
1965	32.14	55.36	12.50	31.25	54.17	14.58			
1966	48.48	39.39	12.12	36.76	58.82	4.41			
1967	47.50	42.50	10.00	66.67	26.67	6.67			
1968	50.00	45.45	4.55	63.89	22.22	13.89			
1969	36.00	52.00	12.00	60.53	34.21	5.26			
1970	73.08	15.38	11.54	56.00	36.00	8.00			
1971	72.73	13.64	13.64	68.00	12.00	20.00			
1972	72.73	15.15	12.12	68.57	25.71	5.71			
1973	50.00	25.00	25.00	54.17	25.00	20.83			
1974	50.00	27.78	22.22	76.92	11.54	11.54			
1975	64.71	17.65	17.65	80.00	10.00	10.00			
1976	59.32	22.03	18.64	59.65	12.28	28.07			
1977	59.38	12.50	28.13	88.00	2.00	10.00			
1978	65.12	11.63	23.26	90.00	5.00	5.00			
1979	58.33	11.11	30.56	81.82	9.09	9.09			
1980	54.55	21.82	23.64	87.50	7.50	5.00			
1981	82.50	10.00	7.50	83.87	12.90	3.23			
1982	70.21	17.02	12.77	86.67	11.11	2.22			
1983	50.00	10.00	40.00	77.78	11.11	11.11			
Average									
1955-60	36.73	50.76	12.51	17.69	73.53	8.78			
TAPT-00	37.51	51.33	11.16	39.24	51.77	8.99			
190/-/4	56.50	29.61	13.88	64.34	24.17	11.49			
1000 00 19/5-81	63.41	15.25	21.34	81.55	8.40	10.06			
TAQ5-Q3	6U.II 	13.51	26.38	82.22	11.11	6.67			

VOLUNTARY AND REGULATED COLLECTIVE BARGAINING IN MANUFACTURING BY LEVEL

Notes: a. The Voluntary area includes columns 1 and 2 from Table 5.2.2.

b. The Regulated area includes columns 3, 4, 5, 6 and 7 from Table 5.2.2.

				1955-83				
Level	Natio	nal	Loc	al	Spec	ial	Tot	al
Year	Α	С	A	С	Ā	С	А	С
	¥	8	8	*	*	*		
1955	50.00	24.14	0.00	62.07	50.00	13.79	2	29
1956	25.00	16.67	75.00	83.33	0.00	0.00	16	6
1957	n.a	n.a	n.a	n.a	n.a	n.a	n.a	n.a
1958	25.00	25.00	75.00	75.00	0.00	0.00	4	8
1959	30.77	7.69	61.54	69.23	7.69	23.08	13	13
1960	27.27	8.00	63.64	71.43	9.09	14.29	11	7
1961	27.27	0.00	56.82	100.00	15.91	0.00	44	3
1962	28.00	18.18	64.00	54.55	8.00	27.27	25	11
1963	33.33	41.18	58.33	52.94	8.33	5.88	24	17
1964	35.29	55.00	47.06	40.00	17.65	5.00	51	20
1965	28.26	43.75	56.52	50.00	15.22	6.25	46	16
1966	46.67	45.45	40.00	54.55	13.33	0.00	30	22
1967	44.74	90.91	44.74	9.09	10.53	0.00	38	11
1968	42.11	77.78	52.63	11.11	5.26	11.11	19	9
1969	39.13	71.43	47.83	28.57	13.04	0.00	23	14
1970	70.83	80.00	16.67	20.00	12.50	0.00	24	5
1971	66.67	80.00	16.67	0.00	16.67	20.00	18	5
1972	66.67	70.00	18.52	20.00	14.81	10.00	27	10
1973	51.85	61.54	22.22	15.38	25.93	23.08	27	13
1974	48.39	88.24	25.81	11.76	25.81	0.00	31	17
1975	60.00	82.61	20.00	4.35	20.00	13.04	30	23
1976	51.16	55.56	23.26	19.44	25.58	25.00	43	36
1977	42.86	92.50	14.29	0.00	42.86	7.50	21	40
1978	53.85	92.11	11.54	5.26	34.62	2.63	26	38
1979	45.83	85.71	8.33	9.52	45.83	4.76	24	42
1980	38.71	87.50	25.81	7.50	35.48	5.00	31	40
1981	73.91	86.67	13.04	10.00	13.04	3.33	23	30
1982	56.00	90.70	20.00	9.30	24.00	0.00	25	43
1983	50.00	76.47	0.00	11.76	50.00	11.76	8	17
Average	9							
1955-60	0 31.61	17.56	55.03	72.21	13.36	10.23		
1961-6	5 33.14	33.93	53.79	58.67	13.07	7.40		
1967-74	4 53.80	77.49	30.63	14.49	15.57	8.02		
1975-8:	1 52.33	83.24	16.61	8.01	31.06	8.75		
1982-8:	3 53.00	83.•58	10.00	10.53	37.00	5.88		

<u>COLLECTIVE AGREEMENTS AND COMPULSORY ARBITRATION AWARDS</u> <u>IN MANUFACTURING BY LEVEL</u>

Notes: A = Collective Agreements. C = Compulsory Arbitration Awards.

Source: Ministry of Labour.

.

.

CHAPTER 3. THIRD PARTY INTERVENTION AND COMPULSORY ARBITRATION.

3.1. INTRODUCTION AND SUMMARY

Compulsory arbitration is another main feature of the post-war system of industrial relations in Greece which has been based upon Law 3239 introduced in 1955. Arbitration is an additional way of regulation of collective bargaining evolving within the formal system. Its main aim has been to control industrial conflicts arising during formal collective bargaining. Its compulsory character has been the target of frequent criticism during the last two decades. But is the problem just the compulsory character of arbitration?

This chapter discusses the role of compulsory arbitration in theory and in practice and examines the extent to which arbitration has been used and the institutional developments associated with the bargaining behaviour of salary and wage earners. Strike activity is used as the major indicator of their bargaining behaviour. The analysis focuses upon developments in the manufacturing bargaining structure.

Section 3.2 discusses theoretical and methodological questions about the compatibility of arbitration with collective bargaining. Section 3.3 presents and evaluates the Greek system of compulsory arbitration in use since 1955. Section 3.4. presents the strike activity record in Greece and discusses its significance regarding the bargaining structure and wage setting

procedures in the manufacturing sector. Section 3.5. summarises the findings and presents the conclusions of this chapter.

Our analysis suggests that the problem in the Greek system of industrial relations is not simply one concerning the operation of arbitration as it incorporates both the formal structure of collective bargaining and the characteristics of compulsory arbitration. In Greece arbitration is not compatible with collective bargaining, not because of its compulsory element but because of the lack of the essential characteristics that an efficient arbitration system should maintain. It is not independent from the Ministry of Labour and is dominated by incomes policies.

The combination of regulation of both procedural and substantial aspects of collective bargaining has been associated with increasing trends in strike activity dominating Greek industrial relations since 1975. Especially in the manufacturing sector in 1975-81 strike activity at the company and plant level extended the system of collective bargaining from the formal secondary and tertiary levels towards the primary level of the trade-unions structure creating thus an informal system of collective bargaining. Its characteristics and its frequency make strike activity the more relevant industrial relations variable for the analysis of developments as far as wage determination in the Greek manufacturing sector is concerned.

3.2. ARBITRATION AND COLLECTIVE BARGAINING.

3.2.1. Third Party Intervention and Arbitration.

In wage setting under collective bargaining cases of impasse arise between unions and employers. Such an impasse is caused if, as Hunter (1983) analysed, the initial positions and the bargaining concessions of the parties to the bargaining process do not lead to the creation of a contract zone within which settlements may be reached. Strike activity and lock-outs provide to both parties the means for imposing disagreement costs upon the other party so as to promote the creation of a contract zone and lead to a settlement of the dispute.

As strikes impose costs on both parties, and in the case of essential services imply costs to the public, governments become involved in policies aiming to affect the behaviour of employees and employers in cases of impasse in collective bargaining. Longlasting institutional policies regarding third party intervention can be considered as governmental involvement in the procedural aspects of wage determination. Such policies, undertaken for facilitating the creation of contract zones in cases of impasse in collective bargaining, are aimed also at promoting stability in the system of industrial relations by minimising the number of industrial disputes and the intensity of conflict.

Third-party intervention policies aimed at the behaviour of

employees and employers involved in collective bargaining take various forms. Conciliation and mediation are weaker forms of third party intervention. As Webb (1986:249) pointed out, "conciliation is usually described in terms of the third party acting as a "go-between" without making a substantive contribution to negotiations" while "mediation is described as a more influential, substantive role where the third party may employ techniques ... directed at discovering negotiated solutions to the dispute".

Arbitration is a stronger form of intermediation as arbitral bodies have usually the authority to suggest or even dictate a certain dispute resolution. Arbitration can take different forms. It can be voluntary or compulsory, binding or not, conventional or pendulum, i.e. based on final offers by the parties involved in the dispute. Strike activity and arbitration are considered as alternative impasse resolution procedures. As Ashenfelter (1985:29) put it "the purpose of arbitration systems is to produce the settlement of disputes in a way that is less costly than the alternatives".

According to ILO (1980) systems of arbitration have been used in some 22 countries and are the principal mode of labour dispute resolution in a further 14 countries. From those systems three main models of arbitration can be discerned, namely those used in the U.S. the U.K. and Australia. In the U.S. binding arbitration is mainly used in the public sector and is provided by independent professional arbitrators. In the U.K. arbitration is

a voluntary process and arbitration services are provided by ACAS, an independent institution. In Australia compulsory arbitration dominates the national system of industrial relations. ACAC an independent and autonomous committee of industrial relations experts plays the central role in the federal and state system of compulsory arbitration. The basic characteristics of these systems are discussed in Appendix 1.

3.2.2. The Behaviour of Arbitrators.

From a theoretical standpoint the main question about arbitration is whether or not it is compatible with collective bargaining. This question can be discussed at two levels. First at a micro level concerning unions and employers behaviour given the availability of independent arbitration procedures. This type of approach to third party intervention as conflict resolution device has been criticised as being based on "an instrumental individualistic model of social relations" and as ignoring "the social control aspect of third parties function". (Webb,1986). Second, at the macro level with regard to the influence, formal or informal, that governments may exert upon arbitration procedures and outcomes. In both cases, arbitrators behaviour towards employers, unions and governments is central for determining whether or not arbitration is compatible with collective bargaining.

For those considering arbitration incompatible with collective bargaining, arbitration has a "freezing" or "chilling" effect

upon collective bargaining because, so the argument runs, favours the persistent failure of the bargaining parties to create a contract zone and bargain within this voluntarily created zone. In the broad literature on arbitration, the discussion on the "freezing" or "chilling" effect depends on the assumption made with regard to arbitrators behaviour.

Stevens (1966), for instance, argued that conventional arbitration tends to discourage serious bargaining because conventional arbitrators are perceived by bargainers to split the difference between the parties to the bargaining process. However, Stevens (1966:49-50) argued that a strong compulsory arbitration system, i.e. a system of final-offer-arbitration, may be compatible with collective bargaining as the availability to the parties of an arbitration strategy, regarding one party's offer vis-a-vis the other party's expected offer, under this type of arbitration "would serve some of the functions usually associated with a strike strategy".

Farber and Katz (1979) formalised Stevens' analysis and examined the conditions under which such a "chilling" effect may be caused. According to them, a contract zone may be generated either by the costs of arbitration or, by at least one bargainer having overly pessimistic expectations about the arbitrator's award. Conversely, a contract zone will tend to be small or nonexistent when at least one bargainer has overly optimistic expectations about an arbitrators award.

In other theoretical attempts to examine the compatibility of

arbitration with collective bargaining, Bloom (1981) argued that even if the parties have no uncertainty about the arbitration award they will still have an incentive to negotiate as long as their expected costs of arbitration exceed their expected costs of negotiation. Farber (1980) analysed theoretically the system of final-offer-arbitration and one of his conclusions was the recognition of the stochastic nature, i.e. the unpredictability, of arbitrators decisions. Thus, in theory, arbitration can be, under certain circumstances, compatible with collective bargaining.

In the U.S. literature it has been argued that arbitration does not cause any "freezing" or "chilling" effect upon collective Ashenfelter (1985), in a survey of the results of bargaining. quantitative analyses of interest arbitration systems operating in the U.S., concluded that arbitrators do not behave in the way assumed by the "chilling" effect literature. In other words, their behaviour is not dominated by a "splitting the differences" attitude, regardless the type of systems under which they operate, and they independently arrive to a reasonable award based on the facts and on a unique assessment of each individual case. "The chilling effect disappears so long as arbitrators introduce exogenous information into their decisions in a way that is to some extent unpredictable by the parties" (Ashenfelter 1985:7). The recognition of the stochastic nature of arbitrators decisions was also stressed in Ashenfelter (1987).

3.2.3. Arbitration and Incomes Policy.

The second aspect on whether or not arbitration is compatible with collective bargaining has to do with the question whether and to what extent, autonomous arbitration can, in practice, remain independent from governmental incomes policies. Governments are concerned not only with the procedural aspects of collective bargaining, which are dealt with long-lasting policies on third-party intervention for conflict resolution, but, indeed, with the substantial issues of the wage setting process. Over the post-war period, incomes policy has been a widely used macroeconomic tool (Flanagan et al, 1983). As arbitration requires freedom of scope, which is denied by the prescriptions of any incomes policy, a contradiction is created between arbitration and incomes policy.

Hunter (1983) discussed those problems posed by arbitration during periods of incomes policy and pointed out that "whatever the form of incomes policy, the intention must be to affect the outcome of bargaining, and hence by implication, the process of bargaining itself. The establishment of a pay limit or norm, for example, may influence the starting point for bargaining on one or both sides, or the rate of convergence to a contract range". Furthermore, in conditions of an independent arbitration service the government may need, and therefore legislate, the right to overrule awards directly, when it considers the outcome of the arbitration process as damaging to its incomes policy. But, then, arbitration awards are neither independent nor of stochastic

nature, i.e. not predictable by the parties of the dispute.

According to Clarke (1987b:398) in many countries state arbitration services have been faced with friction and problems as their independent even-handed attitude has been questioned by either side of collective bargaining when dealing with cases in which governments have been interested on the outcome. The obvious solution to these problems has been the separation of state arbitration services from direct governmental administration, as it happened in the U.K. with the creation of the autonomous ACAS in 1975 which assumed the arbitration functions earlier carried out by the Department of Employment.

Arbitration with defined criteria is an alternative to the right to overrule awards. In the context of an independent and autonomous arbitral body, as Hunter (1983:74) put it "one response to the charge that arbitration does not take into account the national interest as expressed in governmental policy is the suggestion that the national interest should be explicitly represented in arbitration". In this case, the autonomous arbitral body becomes another independent actor of key importance for the implementation of incomes policy.

The extent to which autonomous arbitration services maintain their independence or become part of the governmental machinery for incomes policy implementation is a matter of empirical evidence. In the context of the U.K. and the Australian systems of industrial relations where more extensive use of arbitration machinery is observed and where successive incomes policies have

been at work, the question of compatibility of arbitration with collective bargaining becomes more complicate.

Nevertheless, the voluntary character of arbitration in the U.K. seems to be rather compatible with collective bargaining. Indeed, according to Hunter (1983:72), in the U.K. "arbitrators have generally taken the view that arbitration should take incomes policy into account, but should not allow the policy to be the overriding consideration governing the award". Recent developments such as the refusal of the government, which acts both as employer and third-party, to refer the ambulance drivers' pay dispute to arbitration indicates that arbitration may well be independent from governmental policies and, from this point of view, can be considered as compatible with collective bargaining.

As far as the Australian system of compulsory arbitration is concerned it appears that it is rather compatible with collective bargaining too. Bamber and Lansbury (1987:11) observed that "there is relatively little "compulsion" in practice and the arbitration tribunals rely mainly on advice and persuasion". Braun (1975) on whether or not compulsory arbitration in Australia served as a form of incomes policy, had criticised the attempts by the Government, employers and economists "to convert the Commission into a recognised instrument for achieving a noninflationary development of wages" because they "were distorting a valuable institution which should have been used within the broader framework of an incomes policy instead of being virtually the sole instrument of such a policy".

However, Lansbury and Davis (1987:102) noticed that the lack of power to act in industrial relations, particularly in regard to pay policy, has frustrated Australian governments of all political persuasion. The Fraser government (1975-83), for instance, had strong exchanges with the Commission which responded that they were not an arm of Governmental economic policy but "an independent body required to act according to equity, good conscience and the substantive merits of each case".

Yet, the distinct pattern of short strike activity observed in Australia is, to a certain extent, associated with the system of compulsory arbitration. First it consists of tactical strikes designed to soften up employers or to pressurise the tribunal, and second, it consists of the grievance and protest strikes which reflect frustration with the inability of local industrial relations machinery to resolve local disputes. According to Mulvey (1986:21) this strike pattern is not linked only to the workings of the arbitration system but is also associated with the malfunction of poorly developed arrangements for industrial relations at plant level. But, as Creigh (1986) concluded, there is no evidence that the existence of the arbitration system as opposed to those based on collective bargaining has significantly affected stoppage incidence over the period 1962-81.

3.2.4. Summary

To summarise, third party intervention, and arbitration in particular, is used as an alternative to strike activity for

resolving impasse situations in the bargaining process. Both in theory and in practice compulsory arbitration can be compatible with collective bargaining if the arbitral bodies are autonomous, remain independent from governmental policies and do not become part of the governmental machinery for incomes policy implementation. Otherwise, if arbitral bodies do not maintain such characteristics, "chilling" or "freezing" effects upon collective bargaining can be caused. In any case, the relationship between arbitration and incomes policy is, at least, a permanent source of friction. Finally, as the Australian experience suggests, compulsory arbitration may influence the pattern of strike activity.

3.3. THIRD PARTY INTERVENTION IN GREECE: COMPULSORY-NOT-QUITE-ARBITRATION.

3.3.1. Provisions, Evolution and Origins.

In the event that unions and employers are unable to reach voluntarily a collective agreement, Law 3239 (art.9.1) offers a mechanism for referring the dispute to the jurisdiction of compulsory arbitration. Reference can be made unilaterally, or jointly, or may be derived from statutory authority. In other words, either parties to the dispute have the right to invoke the arbitration procedure. Indeed, the Ministry of Labour has too the right to activate arbitration proceedings (art.17.1).

Arbitration in Greece is clearly a compulsory process. In addition to the compulsory features of reference of a dispute to arbitration, during the arbitration process there is no right to strike (art.18.2). Consequently, any industrial action in progress should be abandoned after the referral of the dispute to the arbitration court. Sanctions are imposed when a party does not comply with the compulsory arbitration award (art.18.3). Arbitration awards are legally binding rulings as collective agreements (art.19.2). However, an arbitrated award can be challenged at an arbitration court of higher level, the award of which is final and cannot be challenged (art.12).

Under compulsory arbitration, disputes are settled by an arbitral body, the compulsory arbitration court, which is composed of four

members: i) one judge as president of the court, ii) one official of the Ministry of Labour, iii) one trade-union representative appointed by the trade-union involved or the General Confederation of Labour and iv) one representative of the employers Association involved in the dispute (art.10.1). The composition of the compulsory arbitration court, according to the introductory report of Law 3239, was designed so as to secure impartiality and ability to take into account all the aspects of any dispute referred to. For this reason the judge and the Ministry of Labour official became rather permanent members of the arbitral body.

The system of compulsory arbitration was introduced along with the formal system of collective bargaining in 1955 by Law 3239. The system of compulsory arbitration was the relatively new element provided by Law 3239. In the earlier legislation the procedure of dispute resolution through arbitration was not homogeneous as far as the bodies dealing with dispute resolution are concerned. According to Law 3086 labour disputes were referred to arbitral bodies of two different types, one rather legal and the other administrative, not because of the varied character of the disputes (distinguishing for instance between interest and rights disputes), but according to their location, i.e. the place where the parties involved in the dispute were based.

In addition to institutional homogeneity, another reason given for that reform was that in the case of disputes settled by

judges the industrial relations affairs were dealt as being strictly of juridical nature and the outcome was wholly depended on the judge. Moreover, as the introductory report of the Law 3239 stressed, the earlier system was not accepted by both parties because serious questions on lack of impartiality of the arbitral bodies were raised.

The Greek system of compulsory arbitration has many similarities with arbitration systems used in other countries. Similarities exist with the systems used in Australia and in New Zealand, the compulsory arbitration machinery used in the U.K. during the second world-war period, and the machinery for wage controls enacted during the Korea war in the U.S. Probably, the comparative labour law was used in designing the Greek system of compulsory arbitration. Of course, there is also the Greek tradition, mainly the fascist-corporatist model of collective bargaining established by, and used during, the Metaxas dictatorship (1936-1940). However, our comparative analysis suggests that the British system of war-period arbitration was used as the main model for the design of the Greek system of compulsory arbitration.

The resemblance between the Greek system and the National Arbitration Tribunal constituted in August 1940 under the Emergency Powers (Defence) Act for settling trade disputes during the war period in the U.K. (Turner-Samuels, 1951) is impressive. In that system a dispute is reported by either party to the Minister, who after considering mediation and conciliation, is empowered to refer the dispute to the National Arbitration

The National Arbitration Order forbids strike action Tribunal. unless the dispute is reported to the Minister and is not referred by him for settlement. Till the Tribunal make its award and during the period of reference, lock-outs and strikes are prohibited. An award is possible to be made retrospective to the date on which the issue of the dispute arose. The terms of the award become incorporated on the existing agreement. Questions of revision are to be dealt with in the same way as every other Rates of wage and conditions of employment are to be dispute. regarded as minima. The tribunal consists of five members appointed by the Minister of Labour: three independent members and two members chosen to represent employers and unions. Panels of the latter have to be constituted by the Minister after consultation with CBI and TUC.

The Greek system of compulsory arbitration has also some similarities with the system of arbitration proposed by Meade (1982:108-110) for the U.K. He called it "not-quite-compulsoryarbitration" and it has some similarities with the system of final-offer-arbitration. Its four major features, as summarised by Dolton and Treble (1985), have to do first, with the arbitrators preferences (the statutory duty of the arbitral body to lay stress on the promotion of employment in the sector under review), second, the right to activate the arbitral procedure (it would operate only if at least one of the parties to the dispute or the government itself wished to), third, the provision for a cooling-off period (during which the parties would be free to negotiate for an agreed settlement) and fourth, the effect of

arbitral decisions upon strikes and lock-outs (it would not be unlawful for either side in the dispute to take industrial action, but if this is in opposition to the terms of an award the party should penalised).

The conditions under which all the above mentioned systems were introduced or proposed indicate that nearly all systems of compulsory arbitration are introduced during periods of social crisis and tension in industrial relations. The Australian and the New Zealand systems were both introduced and implemented after a period of strikes, strife, unemployment and severe confrontations between unions and employers (Holt, 1986, Plowman, 1989). The British war-period system was clearly related to emergency conditions while the Meade proposal was made in the aftermath of a period of major crisis in British industrial relations and in the context of stagflation.

In Greece the formal system of collective bargaining, and especially the system of compulsory arbitration, was introduced after a period of civil-war, which made ILO (1949:19) to observe, with reference to industrial relations, that "nothing in Greece is understandable except against the background of political insecurity and economic difficulties which are the factors dominating Greek life today". The expressed aim of Law 3239 was two-fold. Apart from the regulation of labour disputes, it aimed at the development of tripartite collaboration and, therefore, provided for the constitution, with tripartite participation, of the National Council for Social Policy which was designed to

study and advise on living conditions, standards of living and social policy (art.28-33). That initiative was in line with the recommendations of the ILO examination committee, which visited Greece in 1948, as well as with the suggestions of the American Mission in Greece. The tripartite body was supposed to contribute in programming with regard to social policy.

However, that institution never played an important role in the national system of industrial relations. The institution of similar tripartite bodies at the national level has been a recurring phenomenon but, not surprisingly, they did not prosper in the environment of excessive state regulation in industrial relations. Similar institutional arrangements such as the Committee of Prices and Incomes, established by the military government in May 1972, with a view to supervise price and income developments, or the institution of a Prices and Incomes Committee under the chairmanship of the Minister of Co-ordination and Planning in August 1974 produced little change in the statutory regulation of wages and industrial relations.

As analysed in section 3.2, the governmental intervention even in the organisational issues of trade-unionism created the tradition of an always pro-governmental leadership in the General Confederation of Labour. Under conditions of continuous governmental intervention to the internal organisational issues of the Greek trade-unions, and with the trade-unions side being the one having, literary, its leadership reshuffled according to political changes, any tripartite collaboration at the national level cannot be meaningful, effective and efficient and cannot

incorporate the trade-unions rank-and-file. Thirty years of experience indicate that the recipe of tripartite collaboration is called up every time that less popular policies are to be imposed as far as wage determination is concerned.

It is noteworthy that, in spite of the socialist promises since 1981 for a radical change in the area of trade-unionism and industrial relations, those constants of the Greek national system of industrial relations were in full work during the 1980s. Developments during the second term of the socialist government, in the period 1985-89, are indicative for the treatment of tripartite collaboration in Greek industrial relations. After the June 1985 election was won a stabilisation programme was being designed. At the same time, the National Council for Planning and Development was established, in which representatives of government, business and labour were to participate. That was supposed to be the highest tripartite advisory body for economic policy. But the government was also prepared to follow, and did follow, the tradition. When, after the imposition of the stabilisation program, the ruling party lost the majority in the General Confederation of Labour (GSEE), pro-governmental leadership was appointed. Until early in 1989, the General Confederation of Greek Workers was led by progovernmental trade-unionists who, in line with the deep seated national tradition, were appointees.

3.3.2. Compulsion versus Arbitration: A Corporatist Model?

The similarities between the Greek system of compulsory arbitration and systems either used elsewhere or consisting normative approaches to industrial relations indicate that apart from the universal problems arising in the context of collective bargaining, comparativism, as far as labour legislation is concerned, has been a major influence upon policy-making in industrial relations. However, as Kahn-Freund (1974:27) stressed "the use of the comparative method requires a knowledge not only of the foreign law, but also of its social and above all its political context".

In Greece the framework of rather centralised collective bargaining, the availability of compulsory arbitration and the set up, from time to time, of institutions aiming at the tripartite collaboration may create the impression of a corporatist model. The formal system of collective bargaining and the trade-unions structure may appear as its corner-stone. In fact, due to the Civil War (1946-49) and its repercussions, the Greek post-war corporatism, if any, is completely different from the "voluntarist" or "bargained" north European corporatism analysed by Crouch (1977), Schmitter and Lembruch (1979). In the mixture of coercion and voluntarism, as Alivizatos (1983) has shown, coercion maintained, at least until 1974, the predominant role.

Apart from the context of regulated collective bargaining,

another important aspect which distinguishes the system of compulsory arbitration used in Greece from those in the U.S., the U.K. and Australia has to do with the behaviour of the third party which is supposed to perform the role of arbitrator by considering the parties' offers along with external information in order to arrive to the final award. In the Greek system arbitrator is neither the independent individual professional arbitrator as in the U.S., nor the independent arbitral body appointed by autonomous institutions as the ACAS and the ACAC in the U.K. and in Australia respectively. It is the judge and the civil servant of the Ministry of Labour.

In fact, Law 3239 stipulates that arbitral bodies should consider disputes without any influence and as independent agents (art. 11.2-3). It also defines the general criteria upon which awards should be based (art. 15.3). In practice, as it was recognised by high rank Ministry of Labour civil servants with large experience in arbitration proceedings, it is the Ministry of Labour representative who influences and even dictates the outcome of the arbitration process. Judges are mainly responsible for the procedural part of the compulsory arbitration process. The Ministry of Labour representative plays the key role in settling the substantial issues of the dispute (Committee of Review on Incomes Policy, 1976:115). Thus, the Ministry of Labour itself is the arbitrator and, therefore, there is complete identity between the Government and the third party which is supposed to arbitrate.

Under these circumstances, the third parties of the arbitral body

are far from independent. The criticism of the judiciary for lack of autonomy has been a permanent feature in the Greek political and social life. As Alivizatos (1983) and Pollis (1987) have shown, successive governments have manipulated both the law and the judiciary to promote their policies regardless of constitutional and legal restrictions. Moreover, as judges lack any special knowledge on industrial relations and labour market issues, and given the presence of the Ministry of Labour representative, there is no much room for judges to remain, during the arbitration procedure, autonomous and independent from the policy of the Ministry of Labour. Judges simply agree with the theses supported by the Ministry of Labour representatives (Committee of Review on Incomes Policy, 1976:115).

The Ministry of Labour has preferences originating from the, explicit or implicit, governmental incomes policy. Its representatives, not surprisingly, have to follow the instructions of the political leadership of the Ministry, the Minister himself (up to the present there has never been a woman minister), and his advisers. In this context arbitrators not only have had to take account of the incomes policy but they have been supposed to implement it. This is due not only to the character of the civil service as part of the governmental machinery, but, indeed, to the functional relationship between the Ministry of Co-ordination (currently Ministry of National Economy) and the Ministry of Labour.

In practice, the Ministry of Labour is assigned the role to

implement and to handle the incomes policy adopted by the Government and supervised by the Ministry of National Economy. Using the categorisation proposed by Meade (1982:98), we can say that the Ministry of National Economy acts as a "centralised wage-guidance institution" while the Ministry of Labour acts as a "centralised wage-fixing institution". In many cases, when a dispute became of political importance, the Minister was involved in person in mediation and arbitration procedures. His advisers are often involved, informally, in arbitration processes. But usually, when the dispute takes place at low levels of the bargaining structure the job is left to the Ministry of Labour civil servants acting as members of the arbitral body.

Recently, in 1988, there have been episodes of deviant behaviour of the Ministry of Labour regarding the implementation of the incomes policy, as the then Minister of Labour (Mr. Genimatas) vetoed the policy of the Minister of National Economy (Mr. Simitis), with regard to the level of automatic indexation. Such episodes, which were the tip of the iceberg of socialists' intraparty and intra-governmental conflict, are the exceptions verifying the existence of this working relationship between these two ministries. After all, this relationship affects directly, if not decisively, the arbitrators behaviour.

Usually, in conditions of an independent arbitration service, governments adopting incomes policies may need the right to overrule arbitration awards where they see the outcome of the arbitration process as damaging to their policies. When the arbitral body is largely a governmental body, as it is in the

case of the Greek system of compulsory arbitration, such a right might be considered redundant. However, since 1955 in Greece, Law 3239 (art.20.2) provided for the governmental right to amend or to reject collective agreements and compulsory arbitration awards (in the outcome of which the government itself plays the central role) when considered as not in line with the governmental wage policy. This right was dropped in 1974, in a move towards a more liberal system of collective bargaining.

Arbitration with defined criteria and representation of the national interest, as expressed by the governmental policy, in the arbitral body are usually considered as alternatives to the right to overrule awards. In the Greek system of compulsory arbitration, Law 3239 (art.15.3) defines the criteria upon which the arbitral bodies should base their awards. Indeed, it provides that the Ministry of Labour has the right to deal with the substantial issues settled by compulsory arbitration court, where also the Ministry of Labour civil servant represents the national interest as perceived and expressed by the government in office. Thus, the Ministry of Labour is clearly and directly represented in the arbitration process. The governmental, as well as the national interest, is therefore over-represented in compulsory arbitration proceedings, before and after them. This unique over-representation completes a system of strong legalistic and bureaucratic regulation upon wage determination.

From an economic point of view and in the context of incomes policy, the right to amend and to reject collective agreements

and compulsory arbitration awards has some similarity with the tax-based incomes policy proposed by Layard (1982). The rejection of collective agreements or compulsory arbitration awards considered not in accordance with the governmental policy can be understood as sort of the Layard incomes policy with the tax multiplier of the excess increase in wages being indefinite. Furthermore, some similarities, regarding the key role of judges and Ministry of Labour representatives in the arbitration processes, exist between the Greek system of compulsory arbitration and the rationale of the incomes policy proposed by Meade (1982:83-118).

In this environment there is not much uncertainty regarding the impartiality of arbitrators and concerning arbitration outcomes. Therefore, there is not room for considering the arbitrators behaviour as one of stochastic nature. While in the U.K. and in Australia arbitrators may well take the view that incomes policy should be taken into account but not as the factor governing the award, in Greece, as the arbitrator is a Ministry of Labour civil servant and as the Ministry of Labour is assigned with the implementation of incomes policy, we cannot expect the third party to simply take into account the incomes policy. As the Committee of Review on Incomes Policy (1976:114-115) concluded, in arbitration procedures the governmental incomes policy is the overriding consideration.

Probably this difference in arbitrator behaviour lies behind the fact that in the U.K., for instance, there have been cases where governments opposed the referral of a dispute to

arbitration because of running the risk of contravention of their policy. In Greece, however, successive governments keen to implement their incomes policy had no hesitations about the arbitration outcome and, thus, they used the compulsory arbitration procedure systematically and extensively. The formal system of collective bargaining and especially the compulsory arbitration machinery was designed to be, and was used, as an instrument of national incomes policy.

3.3.3. The Centrality of Compulsory Arbitration.

Compulsory arbitration has dominated the formal system of wage determination. Table 3.3.1 presents the number of both collective agreements and compulsory arbitration awards registered with the Ministry of Labour and indicates the extent to which compulsory arbitration has been used. Since 1964, nearly half of the disputes evolving within the formal system of collective bargaining have been settled by compulsory arbitration courts. Data referring to the number of employees covered by compulsory arbitration awards would be more precise in measuring its coverage in the Greek labour market. Unfortunately, neither the Ministry of Labour ever recorded such information, nor data on the membership of trade-unions involved in compulsory arbitration processes are available. Only data on the number of disputes settled by compulsory arbitration are available.

Through the years the number of disputes resolved by arbitration increased not only in absolute terms but percentage-

The average share of arbitration increased from wise as well. 45.42 in 1964-66 to 54.87 per cent in 1975-81. During the period 1982-88 it was lower, namely 49.89 per cent, but this slight decrease is due to the fact that in 1988 the share of compulsory arbitration awards in dispute settlement was only 28.33 per cent, i.e. the lowest ever recorded. In 1988, after the end of the 1985-87 stabilisation program, the government, to show its determination to promote collective bargaining and to reform the system of compulsory arbitration, decided to minimise the use of compulsory arbitration. Therefore, the 1988 developments influence the image of the whole period 1982-88. But the average share of arbitrated settlements in 1982-87 is much the same as in 1975-81. Despite these changes between periods the overall comparison between the number of collective agreements reached voluntarily and the number of disputes settled through compulsory arbitration reinforces the line of argument that compulsory arbitration played a dominant role.

Compulsory arbitration played a dominant role at every level of the formal bargaining structure. Table 3.3.2 presents the shares by level of compulsory arbitration awards in the total number of both collective agreements and compulsory arbitration awards registered over the period 1973-88. The role of compulsory arbitration is more predominant in settling disputes at the level of the National General and the National Occupational collective bargaining. It is noteworthy that while the recourse to arbitration for the settlement of disputes at the national level of industry-wide or occupation-wide bargaining increased through
the periods 1973-74, 1975-81 and 1982-88, the reverse trend is observed with regard to disputes evolving at the special company level. For every ten disputes evolving at the national level seven of them are normally settled by arbitration courts. However, for every ten disputes evolving at the lowest level of the formal bargaining structure in 1975-81 five of them, and in 1982-88 only three of them, are settled by arbitration. In other words, within the formal bargaining structure the move towards decentralised and fragmented bargaining is followed by relatively less use of arbitration procedures.

Tables 3.3.1 and 3.3.2 suggest that within the formal system of collective bargaining, compulsory arbitration was the preeminent method of wage and job regulation. It was not predominant only in the 1960s but throughout the 1970s and the 1980s. After 1975 in particular, primarily for dispute resolution at the national and the regional level of collective bargaining, arbitration was resorted to extensively. Given that the parties had probably to bargain repeatedly in the face of the same fixed rules, the question is what was their behaviour towards arbitration, and especially which side invoked the arbitration procedure and to what extent.

Trade-unions and employers can exert some discretion on whether or not to refer a dispute to arbitration but this is where their discretion ends as they cannot oppose referrals and do not have the right to select arbitrators either. As during the awardmaking process third parties are not really autonomous there is

no point for such a selection. However, the parties behaviour would be rather stable unless important changes occurred in arbitrators behaviour and would be associated with what they expect from arbitration. Even though both parties participate, not always directly, to the arbitral body, third parties alone can determine the arbitration outcome as the award is decided by majority vote and in the event that two members are in favour and two against it is the judge's vote which determines the outcome (art.11.4). In other words, third parties play the central and dominant role.

Third parties have preferences, arising from incomes policy considerations, and the intention to affect the outcome. The two parties of the dispute have their own preferences and normally are not equally in favour of, or in opposition to incomes policy. While one side may be reluctant to refer the dispute to arbitration the other may be keen to do so. The knowledge that this "preferential" arbitration is always available reduces the efforts of the party feeling more familiar with the governmental incomes policy to reach an agreement voluntarily during the negotiations phase. Moreover, the knowledge of the arbitration's preferences makes the optimistic party to maintain a rigid bargaining position.

As in most cases over the period 1967-88 the employers' side was more supportive for the governmental incomes policy and its offers were closer to the favoured award (with the exception of the incomes policy adopted in 1982 by the newly elected Socialist government), the trade-union side should have been keen to avoid

the referral of disputes to arbitration. Instead, it should have sought better settlements through strike activity and collective bargaining. Consequently, trade-unions should have been the more unwilling party for referral of disputes to arbitration.

Systematic data on which side initiated arbitration proceedings are not available. But successive Ministers and Ministry of Labour civil servants in interviews and public statements said that the majority of compulsory arbitration processes was invoked and initiated by trade-unions. Surprising as it may seem, according to Ministers and civil servants responsible for arbitration proceedings, more than 70 per cent of the compulsory arbitration processes were initiated by the trade-union side. There is no further way to certify it by using statistical information, but both Ministers and civil servants repeatedly made similar statements and nobody argued against. Trade-union officers interviewed agreed with those statements.

Is this a paradox? Yes, it is given first, the role of compulsory arbitration in the implementation of successive incomes policies, and second the above described rationale on why the trade-union side should have been the less optimistic side as far as the arbitration outcome is concerned. However, it seems that trade-union leaders were more familiar with referring disputes to compulsory arbitration, than the theoretical overview of the system suggests. Indeed, as shown in Tables 3.3.1 and 3.3.2, the increasing share of arbitration awards indicates that the parties' dependence upon compulsory arbitration, especially

of those involved in industry-wide or occupation-wide collective bargaining at the national level, increased over time. It appears that trade-unions had the initiative towards this increasing dependence. How can this trade-union behaviour be explained?

An explanation put forward by some trade-unionists focuses upon the lack of independence of trade-union leaders of a certain category of unions active within the formal system of collective bargaining. Trade-union officers have been largely, directly or not, appointed by the Ministry of Labour. Therefore, they, so the argument runs, simply do properly their job as "agents" of the Ministry of Labour within the trade-union movement by providing the Ministry of Labour with the chance to regulate collective agreements. Although there is lot of truth in this argument, a more complete explanation should take into account that tradeunions of low bargaining power, if any at all, might have had to make compromises and settle for the compulsory arbitration award. Therefore, they had to take the initiative and refer disputes to compulsory arbitration in order to get awards effective from the date of reference. The earliest the dispute is referred to arbitration the more retrospective is the arbitration award.

In this context arbitration procedures may well operate as a "face-saving" process for trade-union leaderships. "Heroic" claims cost nothing and provide the trade-union leaders with credentials of militancy, quite necessary to maintain control over their rank-and-file. For the referral of the dispute to arbitration the employers' rigid bargaining position is blamed.

For the outcome the blame goes to arbitrators and the Ministry of Labour. However, high wage claims which remain persistently unfulfilled could reinforce trade-unions' weakness as members may end up considering the unions not effective in getting their wage demands and therefore being reluctant to join them. Such a development is consistent with the increasing financial dependence of trade-unions by the Ministry of Labour fund.

explain further this type of behaviour of union leaders То there is another dimension which concerns the reproduction of the trade-union structure. As Sabethai (1986:4) observed "it has been pointed out repeatedly in the past that compulsory arbitration (apart from limiting the free exercise of the right to collective bargaining) has permitted the survival of unions without any real representativity, merely on the strength of right to be parties to a collective dispute and their formal thus initiate arbitration proceedings". From this point of view one of the causes, as well as effects, of the union initiated referrals of disputes to arbitration have been the conservation and perpetuation of the existing trade-union structure.

To this respect, arbitration procedures sustained the conservation of the formal bargaining structure as well. Compulsory arbitration courts played a central role by granting legal recognition to trade-unions of low bargaining power and seriously questioned representativity. Therefore, it is not surprising that, as indicated in Table 3.3.2, the share of arbitrated settlements bearing at industry-wide or occupation-

wide bargaining at the national level was persistently high and slightly increasing after 1973. It is in this area of the formal structure that most bargaining trade-unions lacking representativity and bargaining power can be traced. An observation made by Braun (1975:206) regarding the Australian system of compulsory arbitration is quite relevant to this category of Greek trade-unions, as she referred to "the existence of close and continuous personal contacts, on an almost day-to-day basis, between relatively few individuals with considerable authority in the employers' organisations, unions and government departments" which "are obviously easier to achieve in a small than in a large economy".

The existence of similar relations between a large category of trade-union officers and the governmental apparatus carrying out industrial relations policies, and their behaviour regarding compulsory arbitration lead to the conclusion that in Greece the system of arbitration has operated as the vehicle of a mutual exchange between governments and trade-union leaders and officers. On the one hand, governments exert a degree of control upon trade-union developments and, thus, facilitate the implementation of their policies. On the other, trade-union officers maintain their positions and status. At the same time, the formal bargaining structure is preserved as much as possible.

Compulsory arbitration plays the central role in all these processes. Therefore, its centrality in the formal system of collective bargaining arises not only from its extensive and increasing use, but, indeed, concerns its crucial role for the

survival and reproduction of bargaining units and trade-unions acting within the formal system. In such conditions of compulsory-not-quite-arbitration, as the Committee of Review on Incomes Policies (1976:114-115) observed, the parties' offers during the bargaining process, as well as their claims presented to arbitration proceedings, are designed more as "face-saving" than as credible tactical bargaining positions. This is more plausible for the trade-unions side. This behaviour indicates a "chilling" or "freezing" effect upon collective bargaining. But, it is not arbitration per se which bears the responsibility. It is the predominance of compulsion and the lack of independence and autonomy of the arbitral bodies versus the Ministry of Labour and the governmental incomes policy.

3.4. STRIKE ACTIVITY AND DECENTRALISATION OF THE MANUFACTURING BARGAINING STRUCTURE.

3.4.1. Compulsory Arbitration and Strike Activity.

Arbitration is considered and used as an impasse resolution procedure alternative to strike activity. If arbitration is successful, other things being equal, there should be a negative association between the use of strike activity and the use of arbitration procedures. Normally, as Feuille (1979) pointed out, strikes and arbitration are negatively correlated. From this point of view, strike activity provides a major criterion for evaluating the effectiveness of arbitration systems.

The Greek system of compulsory arbitration initiated by Law 3239 aimed exactly at the regulation of labour disputes. Compulsory arbitration was adopted as a means for minimising industrial conflicts and their costs. During the arbitration process the right to strike is suspended and consequently, any industrial action in progress should be abandoned after the referral of the dispute to the arbitration court. And strikes challenging arbitration awards are consider unlawful. On this basis the system of compulsory arbitration aimed also at the promotion of a tripartite collaboration through the tripartite character of the arbitral bodies as well as through the establishment of a tripartite committee at the national level to study and advise on living conditions, standards of living and social policy.

Ashenfelter (1985:30) argued that "whether arbitration will grow in popularity depends on whether it is a less costly system than the alternatives and on whether the parties are able to obtain the information and experience necessary for determining whether it is less costly". On this micro dimension of conflict resolution through arbitration depends its growth and its in national industrial performance а system of relations. However, as argued in section 3.3 the Greek system can not be considered as a system of independent arbitration because of the dependence of the arbitral body from the Ministry of Labour. In other words, it really is a compulsory system but it is not a system of arbitration because of the arbitrators behaviour which no way can be considered as of stochastic nature. The popularity of the Greek system of compulsory arbitration may not depend only on whether it is more or less costly than strike activity. It may also depend on its character as part of the governmental machinery for incomes policy implementation.

Surprising enough, over most of the post-war period this system of compulsory-not-quite-arbitration has been popular among tradeunion leaders acting within the formal system of collective bargaining. This popularity might suggest that, despite its idiosynchratic characteristics, trade-unions prefer dispute resolution through arbitration rather than through strike activity. The questions regarding union leaders' independence from the Ministry of Labour may imply that arbitration has not been equally popular to the unions rank-and-file and among

salary and wage earners in general. In the case of the Greek system of compulsory arbitration the distinction between tradeunions leadership and rank-and-file, proposed by Ashenfelter and Johnson (1969), is particularly appropriate. According to Ashenfelter and Johnson there are three parties in the bargaining process and not two, namely management, union leaders and union rank-and-file.

In the model of Ashenfelter and Johnson, strikes occur during the bargaining process as the rank-and-file has high expectations about the wage setting. Leaders are more realistic but agree to take strike action for political reasons, i.e. to avoid the political cost of concluding an agreement at lower levels than their rank-and-file expects. Obviously, in the Greek system of compulsory arbitration trade-union leaders do not seem to behave this way as they themselves prefer to refer disputes to arbitration rather than to undertake strike activity. But has arbitration been popular among their rank-and-file too? Have they followed the same behaviour?

Strike activity provides the best available criterion for evaluating compulsory arbitration from the industrial relations point of view because industrial action reflects the attitudes and the behaviour of unions rank-and-file. The formal bargaining structure, the labour initiated extensive use of arbitration and the structure of the official trade-unionism may create the impression of a corporatist and efficiently operating system of industrial relations. But to evaluate the performance and effectiveness of the Greek system of compulsory arbitration we

examine the strike activity record over the long period compulsory arbitration has been in use.

Three indices on strike activity, given in Table 3.4.1, denote that, in Greece, since the institution of the formal system of collective bargaining in the mid-50s, there has been a steadily increasing trend which had its peak in the period 1979-80 and then remained at high levels. This trend in strike activity over the past three decades indicates that the formal system of collective bargaining and the mechanism of compulsory arbitration were not effective in resolving labour disputes and controlling strike activity. Table 3.4.1 makes clear that from the period 1953-62 to the period 1976-81 the strikers' density increased eight times, the relative days lost increased ten times and the frequency of strikes increased three times. The period 1976-81 is the one with the highest strike activity ever recorded in Greece. In the period 1982-88 strike activity remained at similar high levels. The period 1967-74 is omitted because under the military regime the right to strike was suspended and trade-unions freedom curtailed.

To discern any relationship between compulsory arbitration and strike activity we examine the bivariate correlation between a set of variables indicating the use of arbitration and strike performance. Methodologicaly it is not easy to discern and quantify the influence of compulsory arbitration upon strike activity because strike activity is determined by both institutional and economic factors (Kennan, 1986, Paldam, 1983,

Paldam and Pedersen, 1984). Yet, the results provided in Table 3.4.2 suggest the existence of a positive relationship between the use of arbitration and the recourse to strike activity. As the data on arbitration refer to the number of disputes settled by arbitration courts, the strikes frequency variable is the most relevant to be associated with the number of disputes settled by arbitration. Correlation coefficients for this variable as well as for relative days lost are found positive and significant.

This simple correlation analysis suggests the lack of the hypothesised in the literature negative correlation between strike activity and arbitration and provides only a rough indication of their relationship as other sources of influence upon strike activity are not controlled for. Can the positive correlation between arbitration and strike activity be considered as indicative of smooth operation of the Greek system of compulsory arbitration? This might be the case if strikes precede arbitration procedures and, therefore, strike activity causes and explains the use of arbitration. But it is rather the case of a causal relationship running the other way round. We not only on the fact that arbitration base this inference proceedings were mainly invoked by trade-union leaders, but indeed, upon the observation that a high percentage of strikes did not take place within the formal system of collective bargaining.

Strike activity did not cause the extensive use of arbitration because it largely evolved out of both the tertiary and the secondary level of the bargaining structure, i.e. the bargaining

levels at which disputes can be referred to compulsory arbitration. Over the period 1976-80, for which analytical data are available in the Ministry of Labour, less than 10 per cent of the total number of strikes took place within the secondary and the tertiary level of the formal system. These findings are presented in Table 3.4.3. The number of strikes undertaken within and above the secondary level of the trade-union structure, indicates that over the period 1976-81 not every secondary level organisation, which was entitled to contract formal collective agreements or to invoke arbitration proceedings, resorted to industrial action during the annual bargaining rounds. In other words most of the conflicts recorded by strike activity statistics were not part of formal bargaining processes.

As Paldam (1989:63-64) observed "the number of industrial conflicts always works better in economic models than the variables referring to the number of workers involved (the size of conflicts) and days lost, as these two series are heavily influenced by a few large conflicts between the organisations, while the sheer number of conflicts is a grass-roots measure of labour market tensions". Furthermore, Paldam reported that "the series of number of conflicts for different countries have different, and often dramatic, long-run trends, which are causally related to institutional factors, notably to the relative size and degree of centralisation of the trade-unions in the country".

Of course, even if strike activity and arbitration are normally considered as alternative impasse resolution procedures, arbitration is not a perfect form of no-strike insurance. It does not protect from "wildcat" strikes or stoppages over issues and at levels outside the scope and the structure of the formal system of collective bargaining. But the existence in 1976-81 of such a great number of industrial conflicts at the lowest level of the bargaining structure along with its positive correlation with the use of arbitration cannot be considered as a simple imperfection of the system of compulsory arbitration.

Allowing for statistical errors in recording strike activity, we can say that, since the higher the organisational level of industrial action the easier to record it, the defects of strike statistics may underestimate strike activity undertaken and especially industrial conflicts occurring, at the plant level. From this point of view the number of strikes indicating the number of open conflicts suggests the existence of more bargaining units in the bargaining structure than those officially registered through formal collective agreements and arbitration awards. These bargaining units were probably active out of the formal system of collective bargaining and associated with the trend towards decentralisation observed even within the formal bargaining structure.

This deviant behaviour on behalf of salary and wage earners suggests that in analysing the Greek system of industrial relations the distinction between leaders and rank-and-file should be taken into account as a crucial factor of developments

in industrial relations. As trade-union leaders acting within the formal system were less dependent on their rank-and-file and the latter became, especially in 1975-81, rather strike-prone, the system of industrial relations ended up in a vicious circle of fragmented bargaining and lack confidence between governments, employers, union leaders and union rank-and-file.

The failure of compulsory arbitration incorporates and intensifies the lack of confidence of the participants, or at least of the substantial part of them, which is essential for a system of arbitration to be effective. Given that the arbitration was never independent and autonomous the main obstacle to its effectiveness, i.e. the influence exerted by by the Ministry of Labour, has been long-lasting. This has affected industrial relations in general as the attitudes of the parties, and especially those of salary and wage earners, are crucial for the function not only of a system of arbitration but for the system of industrial relations as a whole. Strike activity did not simply try to pressurise arbitral bodies. To the extent that it was neither part of formal bargaining processes nor of compulsory arbitration, it consisted the only way available for setting up new bargaining relationships out of the formal system of collective bargaining and its bargaining structure.

To summarise, the strike activity record in Greece over the postwar period indicates that compulsory arbitration has been unsuccessful in controlling labour disputes. This is consistent with the idiosynchratic character of the system of arbitration

which has been mainly used as machinery for incomes policy implementation. Compulsory arbitration has been popular among trade-union leaders but not among their rank-and-file. Strike activity should not be simply considered as indicating the malfunction of the system of compulsory arbitration but indeed, as a mean for the creation of bargaining units out of the formal system of collective bargaining. It seems that this growth of collective bargaining out of the formal system occurred far from any collaboration and consensus, but on the basis of inter-union and intra-union conflicts.

<u>3.4.2. Strike Activity and Informal Collective Bargaining in</u> <u>Manufacturing.</u>

Strike activity in Greek manufacturing, according to the indices presented in Table 3.4.4, follows the trend observed at the aggregate level of analysis. It had its peak during the 1975-81 period and then remained at similar high levels. Moreover, strike activity in manufacturing is strongly and positively associated with the use of compulsory arbitration as well as the use of other procedures of regulation of collective bargaining provided by Law 3239.

Table 3.4.5 provides the estimated correlation coefficients relating sets of variables for arbitration and regulation of bargaining, within the formal system, with three variables for strike activity undertaken by wage and salary earners in manufacturing. Nearly all the specifications on the use of

compulsory arbitration and other methods of regulation are found with positive, significant and, indeed, strong correlation coefficients. These results suggest that the use of compulsory arbitration and other methods of regulation failed to minimise industrial conflict in manufacturing.

As the number of conflicts exceeds overwhelmingly the number of formal bargaining units registered in manufacturing it seems reasonable to infer that it is not strike activity which precedes the use of arbitration and regulation, but, mainly arbitration and regulation of collective bargaining are followed by strike activity. This means that largely strikes are not part of the arbitration process but they rather are protest strikes, at least, or strikes attempting to set up new bargaining relationships out of the formal system which is regulated by the Ministry of Labour directly or indirectly via the arbitration machinery.

The index for strikes frequency in manufacturing, presented in Table 3.4.4, compared to that referring to all sectors, presented in Table 3.4.1, indicates a higher than average frequency of strikes in manufacturing. In other words, more strikes per 100000 employees occur in manufacturing than in the rest of the economy, and therefore it seems that conflicts in manufacturing are relatively more decentralised.

The high frequency of industrial conflicts in manufacturing is the main feature of the period 1975-81 as far as strike activity is concerned. In 1976, in particular, the number of industrial

conflicts was the highest ever recorded in Greece. The record number of 184 strikes per every 100000 salary and wage earners in manufacturing were recorded. In 1975 the relevant number may have been high as well but the available strike activity statistics cover only the period from September to December of that year. In 1976, the then Conservative government adopted new legislation, namely Law 330, aiming to control "wildcat" and "informal" strikes at the company level which dominated manufacturing industrial relations after the fall of the military regime in July 1974. Presumably between the adoption of Law 330 and the high frequency of strikes at plant level in 1976 there is a causal relationship running both directions as the new legislation faced a wave of strikes expressing the employees disagreement and discontent.

In the literature varying definitions of informal aspects of industrial relations have been discussed. As far as strike activity is concerned Clarke (1987b:394), for instance, distinguished "wildcat" strikes, as those taking place contrary to procedure agreements for dispute resolution, from the "unofficial" strikes i.e. those taking place outside the rules of their own union. According to Clegg (1976:82) "plant bargaining leads to a relatively large number of official strikes, and industry or regional bargaining to a smaller number of larger official strikes. ... The number of strikes (most of them small and where the distinctions are relevant also unofficial and unconstitutional) is likely to be high where disputes procedures are absent or defective, and where the

distinctions between unofficial and official strikes and/or unconstitutional and constitutional strikes are blurred or non existent".

In Greek manufacturing after 1975, strike activity evolving at the primary level of the bargaining structure and the tradeunions structure was dominant. In 1975-77 most manufacturing disputes were combined struggles for the right to organise at the company level and for higher pay. The early enterprise unions were often involved in clashes with state agencies and employers who tried to suppress them. Typically the full recognition of union rights came in 1982. Over the period 1975-81 along with the plant level strike activity, the structure of unionism underwent substantial changes. Employees got rid of the occupation-based unionism and created a factory unionism which organised the labour force at the plant level regardless of the occupational and technical division of labour (Kasimatis, 1978, Manicas, 1984, Ioannou, 1986).

Table 3.4.6 examines the extent to which industrial conflicts expanded at the plant level during the 1975-81 period and favours the argument about decentralisation as the ratio of strikes per 100 establishment more than doubled in 1976-81 compared to 1962-66. Furthermore, its dispersion over the period 1976-81 is narrower than in 1962-66 and this implies that in 1976-81 the expansion of strike activity at the plant level was less asymmetric than in 1962-66. This observation reinforces the argument regarding the decentralisation of the bargaining process through plant level strike activity. Indeed, in 1976-81 the ratio

of strikes per 100 establishments is higher than in 1962-66 despite the fact that more labour force was concentrated in large-scale industries.

The ratio of strikes per 100 establishments in each two-digit manufacturing industrial branch, presented in Table 3.4.7, indicates the existence of large asymmetries among them, which are mainly associated with the varying number and size of establishments in each branch. In heavily concentrated industries as in branch 34: Basic Metal Industries, in 1976-81 occurred on average four strikes at every ten establishments, while in branch 29: Leather Manufacturing, which is dominated by handicraft plants occurred less than two strikes per 100 establishments. However, the test on stability of the dispersion of industrial conflicts across the twenty two-digit industrial branches, presented in Table 3.4.6, denotes the existence of stability over time, which is probably related to structural characteristics of manufacturing production and employment and behavioural characteristics of the labour force in each branch.

During the period 1976-81, the plant level strike activity was not equally developed across the manufacturing sector. The index of strikers density, which describes the number of strikers per 1000 employees, indicates that employees in certain industrial branches were more strike-prone and active from average. The set of active industrial branches in terms of strikers density over the period 1976-81 include the following: i) 34: Basic Metal Industries, ii) 27: Paper Industry iii) 37: Electrical Machinery

iv) 28: Printing and Publishing v) 38: Transport Equipment (Ioannou, 1988:106). The ranking in descending order of the same data-set on strikers density across the twenty two-digit industrial branches, presented in Table 3.4.8, has a 0.55 coefficient of concordance. This coefficient denotes a rather unstable hierarchy in terms of strike-prone industrial branches in the Greek manufacturing sector over the period 1976-81. Data unavailability on the number of strikers by industrial branch does not allow us to calculate the same coefficient for the whole period 1975-88.

An alternative set of data, the index of days lost per 1000 employees, is used instead. The ranking in descending order of the twenty industrial branches according to the index of days lost per 1000 employees, which is presented in Table 3.4.9 has a 0.25 coefficient of concordance. This value implies that, over the period 1975-88, the hierarchy of industrial branches as far as strike activity is concerned is quite unstable. Statistic evidence on the inter-industry variation in strike activity is summarised in Table 3.4.10. It appears that, during the period 1975-88 in the Greek manufacturing sector, there were no clearly distinguished industrial branches which were leading the plant level strike activity and branches either following the leaders or remaining away of strike activity at the plant level.

The evidence on the expansion and the dispersion of the plant level strike activity indicates the bargaining aggressiveness and the militancy of employees across the manufacturing sector. But by no means should it be perceived as indicator of their

effectiveness. There are no data recorded on the effectiveness of the plant level strike activity. The only evidence available refers to the total number of strikes and strikers over the period 1976-81. This evidence suggests that only 20 per cent of the total number of strikes were reported as being of full or partial success, and only 6 per cent of the total number of strikers were reported as being fully or partially successful (Ioannou, 1988:102).

The difference of percentages between successful strikes and successful strikers may suggest that strikes undertaken at the lower level of the bargaining and trade-unions structure, by relatively small groups of employees, were more successful as far as their objectives are concerned. But, in fact, the effectiveness of the strike activity cannot be properly recorded directly by the strike statistics. Moreover, strike activity is not a good indicator of bargaining power. Groups of employees and whole industrial branches without any significant bargaining power may go out to strike and may be strike-prone.

In any case the plant level unionism aimed at ad hoc single employer arrangements concerning wage determination and job regulation, thus, initiating an informal system of collective bargaining. Such bargaining processes may imply the weakening of the centralised formal system of collective bargaining. Both the formal and the informal system can be considered as parts of a dual system via which primary unions and informal groups of employees tried to "beat" the formal annual collective agreements

and compulsory arbitration awards. As compulsory arbitration has been used to implement incomes policies, it seems that its extensive use caused the rise of informal bargaining procedures by those who disagree with both procedural and substantial aspects of the governmental policies.

To summarise, despite the restrictions imposed upon wage determination through collective bargaining at the plant level, after 1975, successive waves of strike activity developed at the plant level in manufacturing industries This strike activity evolved out of the formal system of collective bargaining as well as out of, and in conflict with, the official structure of the Greek trade-unionism. It resulted in the creation of an informal system of wage determination and created rather erratic wage rounds in the Greek manufacturing sector. Thus, despite the long lasting characteristics of undeveloped and over-regulated institutions and processes of formal collective bargaining, plant level strike activity and informal collective bargaining led to decentralisation and fragmentation of the manufacturing bargaining structure and can be considered as a potential source of influence with regard to wage developments in the Greek manufacturing sector.

3.5. CONCLUSIONS.

Third party intervention, as expressed in systems of arbitration, can be compatible with collective bargaining if arbitrators are and remain independent and, therefore, arbitration procedures work as less costly procedures, than strike activity, for impasse resolution in collective bargaining.

The system of arbitration is a unique aspect of the Greek system of industrial relations. Implemented in 1955 along with the formal system of collective bargaining, as part of it, and dominated by the Ministry of Labour, it rather consists a part of the governmental machinery for incomes policy implementation, rather than an independent arbitration service. By granting legal recognition to unions lacking representativity, compulsory arbitration became the "raison d' etre" of a certain category of trade-union leaders and thus it has been also a machinery for the preservation of the formal bargaining structure. Not surprisingly, it has been used extensively, especially after 1975 when, in the context of the new parliamentary democracy in Greece, collective bargaining was re-established and direct methods of regulation of collective bargaining were abandoned.

Within the formal system of collective bargaining, over the period 1955-88, the parties' dependence upon compulsory arbitration increased over time. This indicates a "freezing" or "chilling" effect upon collective bargaining caused by compulsory

arbitration. It seems that the main source of this effect is neither the compulsory aspect of arbitration nor arbitration per se. It is the combination of its compulsory character with the lack of the essential characteristics of arbitration procedures, i.e. independent and autonomous arbitral bodies. In other words, the Greek system of compulsory arbitration is found incompatible with collective bargaining not because it is compulsory but because it is not a system of arbitration.

However, salary and wage earners developed, especially in 1975-81, strike activity at record levels. Strike activity, which is found positively correlated to the use of compulsory arbitration, indicates the failure of the system of compulsory arbitration in controlling industrial conflicts. Moreover, as strike activity developed largely out of the formal bargaining levels and units, reflects initiatives of bargaining units located out of the formal bargaining structure for decentralised collective bargaining. This development is mainly associated with collective bargaining in manufacturing industries where legislation excludes plant level collective agreements. In 1975-81 strike activity at the plant level initiated an informal system of decentralised collective bargaining, parallel to the formal industry-wide or occupation-based bargaining at the national level. Thus, a dual structure of manufacturing wage determination became at work.

The rise of strike activity and the creation of the informal system of collective bargaining indicate that since 1975, mainly in manufacturing, there was a switch to an alternative route of wage determination initiated by strike activity at the plant

level. This can be mainly attributed to the legalistic character of the formal system of collective bargaining as well as to the ossified bargaining structure imposed upon salary and wage earners.

1

Of course, strike activity is influenced by substantial issues regarding wage formation. And these very substantial issues have been directly affected by the settlements imposed through compulsory arbitration. But, as already argued in Chapter 2, the procedural arrangements provided by the formal system of collective bargaining, which favours centralised bargaining and excludes decentralised bargaining at the company and plant level, contributed to the fragmentation of the real bargaining structure, especially in the Greek manufacturing sector.

Table 3.3.1

1964–1988								
	Number	of Bar	gaining	Units ^a	Density of	Bargaining	Units ^b	
Year	AC	Ca	Both	C/Both	A	С	Both	
190	64: 97	93	190	48.95	7.67	7.35	15.02	
19	65 : 121	93	214	43.46	9.45	7.27	16.72	
19	66: 105	82	187	43.85	8.11	6.33	14.44	
19	67 : 63	79	142	55.63	4.81	6.03	10.84	
19	68: 63	93	156	59.62	4.76	7.02	11.77	
19	69 : 50	100	150	66.67	3.73	7.46	11.20	
19	70: 53	67	120	55.83	3.91	4.95	8.86	
19	71: 46	43	89	48.31	3.36	3.14	6.50	
19	72: 88	38	126	30.16	6.26	2.70	8.96	
19	73: 88	60	148	40.54	10.96	4.16	15.12	
19	74: 70	85	155	54.84	4.74	5.75	10.49	
19	75: 131	143	274	52.19	8.65	9.45	18.10	
19	76: 144	120	264	45.45	9.29	7.74	17.03	
19	77: 146	179	325	55.08	9.20	11.29	20.49	
19	78: 166	214	380	56.32	10.23	13.19	23.43	
19	79: 173	247	420	58.81	10.43	14.90	25.33	
193	80: 220	299	519	57.61	12.99	17.65	30.63	
198	81: 233	330	563	58.61	13.47	19.07	32.54	
193	82: 284	232	516	44.97	16.57	13.53	30.10	
19	83: 57	80	137	58.39	3.36	4.71	8.07	
19	84: 252	264	516	51.16	14.67	15.37	30.03	
19	85: 175	167	342	48.83	9.89	9.44	19.32	
19	86: 44	82	126	65.08	2.48	4.62	7.10	
19	87: 76	84	160	52.50	4.24	4.68	8.92	
19	88: 210	83	293	28.33	11.64	4.60	16.24	
vera	ge							
L964-	66: 108	89	197	45.42	8.41	6.98	15.39	
1967-	74: 65	71	136	51.45	5.32	5.15	10.47	
1975-	81: 173	219	392	54.87	10.61	13.33	23.94	
1982-	88 <mark>:</mark> 157	142	299	49.89	8.98	8.14	17.12	

COLLECTIVE AGREEMENTS AND COMPULSORY ARBITRATION AWARDS

Notes: a. The number of bargaining units refers to both collective agreements and compulsory arbitration awards.

- b. Density of Bargaining Units = Number of Units per 100000 employees in employment.
- c. Collective agreements.d. Compulsory arbitration awards.
- e. Share of compulsory arbitration awards in the total number of both agreements and awards.

Source: Ministry of Labour.

Table 3.3.2

<u> </u>	<u>PERCENTAGE</u>	<u>of both</u>	<u>AGREEME</u> <u>ALL BECTO</u> 1973-198	<u>NTS AND AW</u> <u>RS</u> 8	ARDS BY	<u>LEVEL</u>
Level	General Nati	Occupat onal	ional Local	Special	A11	
Year	ક	ક	ક	ક્ષ	ક	
1973	-	34.72	21.74	54.90	40.54	
1974	-	67.69	44.44	46.03	54.84	
1975	0.00	66.29	73.47	36.09	52.19	
1976	100.00	63.41	52.94	30.77	45.45	
1977	0.00	73.39	65.91	40.94	55.08	
1978	100.00	63.73	54.84	53.25	56.32	
1979	100.00	69.63	60.98	52.07	58.81	
1980	100.00	67.47	51.92	53.00	57.61	
1981	-	70.65	53.57	52.63	58.61	
1982	100.00	66.82	47.50	23.21	44.96	
1983	_	83.64	52.94	38.46	58.39	
1984	0.00	78.34	40.54	30.27	51.16	
1985	0.00	66.88	39.58	32.37	48.83	
1986	0.00	70.49	76.67	47.06	65.08	
1987	100.00	66.13	67.65	30.16	52.50	
1988	0.00	34.13	44.00	15.52	28.33	
Average						
1973-74	-	51.21	33.09	52.72	47.69	
1975-81	-	67.78	59.09	45.54	54.87	
1982-88	-	66.63	52.70	31.01	49.89	
1982-87	-	72.05	54.15	33.59	53.49	

COMPULSORY ARBITRATION AWARDS

Source: Ministry of Labour

Index	Density ^a	Frequency ^b	Days Lost ^C
Year			
1953	79	18	108
1954	45	16	36
1955	45	19	62
1956	n.a	n.a	n.a
1957	100	15	124
1958	78	10	94
1959	35	8	50
1960	46	11	68
1961	43	9	72
1962	46	15	104
1963	80	18	265
1964	130	32	274
1965	200	34	355
1966	269	47	550
1967 ⁴	69	7	87
1975	31	9	145
1976	196	62	501
1977	357	36	769
1978	295	38	579
1979	773	36	938
1980	845	44	15/1
1981	236	27	419
1982	204	52	296
1903	132	40	200
1095	90 612	27	244 601
1985	961	10	715
1987	1213	21	1308
1988	2/9	30	462
Average	247	50	402
1953-62	57	13	80
1963-66	170	33	361
1976-81	450	40	796
1982-88	480	33	615
Notes: a.	Number of Strike	ers per 1000	employees.
b.	Number of Strike	s per 100000	employees.
C.	Number of Workir	ng Days Lost	per 1000 employees.
d.	From January to	April.	
e.	From September t	o December.	

STRIKERS DENSITY, FREQUENCY OF STRIKES AND RELATIVE DAYS LOST IN GREECE, ALL SECTORS 1953-1988

Source: Ministry of Labour.

<u>B</u> RELATING	<u>IVARIATE</u> <u>COMPULSO</u> <u>IN</u> <u>G</u>	<u>CORRELATION</u> RY <u>ARBITRATION</u> REECE <u>All SEC</u> 1953-1988	<u>COEFFICIENTS</u> I <u>AND STRIKE ACTIVIT</u> TORS	Y
Variables	Strike Activity			
Compulsory Arbitration	Density ^d	Frequency ^e I	Days Lost ^f	
Number of Awards ^a	0.30639 ^g 0.1363 ^h 25 ¹	0.49074 0.0127 25	0.51885 0.0079 25	
Degree of Intervention ^b	0.22957 0.2696 25	-0.10737 0.6095 25	0.14279 0.4959 25	
Density of Awards ^C 	0.24542 0.2370 25	0.47537 0.0163 25	0.47533 0.0163 25	

Notes: a. Number of Compulsory Arbitration Awards. b. Percentage of Awards in both Agreements and Awards. c. Number of Awards per 100000 employees. d. Number of Strikers per 1000 employees. e. Number of Strikes per 100000 employees. f. Number of Working Days Lost per 1000 employees. g. Correlation Coefficients

h. Probability of |R| under Ho: Rho=0

i. Number of Observations

NUMBER OF STRIKES ACCORDING TO ORGANISATIONAL LEVEL 1976-1981

Federations		Level of Federations	of Strikes	of Bargaining Units
Year				
 1976	19	2	947	264
1977	21 ^a	1	563	325
1978	30	13	616	380
1979	41	13	588	420
1980	55	17	726	519
1981	n.a	3	466	563

Note: a. From March to December 1977.

Source: Ministry of Labour.

	<u>IN GREEK MANUFACTURING</u> 1959–1988								
Index	Index Density ^a Frequency ^b Days Lost ^C								
Year									
1959	25	69	152						
1960	34	78	212						
1961	30	68	100						
1962	40	76	337						
1963	118	52	393						
1964	250	110	680						
1965	264	81	806						
1966]	254	86	738						
1967 ^a	13	9	34						
1975 ^e	54	26	325						
1976	283	184	964						
1977	371	107	1355						
1978	134	89	594						
1979	226	102	744						
1980	187	124	699						
1981	185	81	764						
1982	452	205	1640						
1983	n.a	n.a	1554						
1984	n.a	n.a	886						
1985	n.a	n.a	530						
1986	171	27	188						
1987	n.a	n.a	286						
1988	n.a	n.a	839						
Average									
1959-60	30	74	182						
1961-66	159	79	509						
1976-81	231	115	853						
1982-88	n.a	n.a	846						

STRIKERS DENSITY, FREQUENCY OF STRIKES AND RELATIVE DAYS LOST

Notes: a. Number of Strikers per 1000 employees. b. Number of Strikes per 100000 employees. c. Number of Working Days Lost per 1000 employees.

- d. From January to April.
- e. From September to December.
- n.a = Not available.

Source: Ministry of Labour.

١

Variables	Strikes			
Regulation	Density ^g	Frequency ^h	Days Lost ⁱ	
Number	0.79124j	0.71842	0.72261	
of	0.0001k	0.0001	0.0001	
Awards ^a	241	24	25	
Degree	0.61855	0.57058	0.73660	
of	0.0013	0.0036	0.0001	
Intervention ^b	24	24	25	
Density	0.85801	0.88579	0.68836	
of	0.0001	0.0001	0.0006	
Awards ^C	20	20	21	
Number of	0.83732	0.83609	0.68861	
Regulated	0.0001	0.0001	0.0001	
Settlements ^d	24	24	25	
Degree	0.29255	0.36748	0.47712	
of	0.1654	0.0773	0.0159	
Regulation ^e	24	24	25	
Density of	0.81155	0.76725	0.52276	
Regulated	0.0001	0.0001	0.0150	
Settlements ^f	20	20	21	

BIVARIATE CORRELATION COEFFICIENTS RELATING BARGAINING REGULATION WITH STRIKE ACTIVITY IN MANUFACTURING 1959-1983

Notes:

a. Number of Compulsory Arbitration Awards.

b. Compulsory Arbitration Awards as percentage of both Agreements and Awards.

c. Number of Compulsory Arbitration Awards per 100000 employees in large-scale Manufacturing.

d. Arbitration Awards and Agreements Amended, Corrected, Rejected and Extended by the Ministry of Labour.

e. Regulated Agreements and Awards as percentage of voluntary Collective Agreements.

f. Number of Regulated Agreements and Awards per 100000 employees in large-scale Manufacturing.

g. Number of Strikers per 1000 employees.

h. Number of Strikes per 100000 employees.

i. Number of Working Days Lost per 1000 employees.

g. Correlation Coefficients

k. Probability |R| under Ho: Rho=0

1. Number of Observations

	FREQUENCY	DISPERSION AND 1962-1981	<u>STABILITY</u>	
Year	Strikes per 100 establishments ^a	Coefficient of Variation ^b	R-square ^C	
1962 1963 1964 1965 1966 1967 1976 1977 1978 1978 1978 1980	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	293.76 255.74 139.23 177.42 155.73 248.28 245.11 117.12 216.20 111.26 214.66 124.54 133.32	1.000 .839 .746 .776 .822 .917 .969 .892 .839 1.000 .948 .884 .692 .695 .959 .910 .719 .783 .759 .782	
Ave: 1962 1976	rage 2-66 2.28 5-81 4.73	196.87 151.44	1.000 .878	

STRIKES PER 100 MANUFACTURING ESTABLISHMENTS: FREQUENCY DISPERSION AND STABILITY

Notes: a. Average of 20 two-digit industrial branches.

- b. Variability across twenty two-digit industrial branches.
- c. Coefficient of stability is measured by correlating relative strike performance of industries among pairs of years, one of which is the base year. N=20 (two-digit manufacturing industrial branches).
- d. From January to April.
- e. From September to December.

Table	3.	4	•	7
-------	----	---	---	---

STRIKES	<u>PER</u>	100 MANUI	FACTURING	<u>estabi</u>	ISHMENTS	BY BRA	ANCH, 19	62-1981
Year	1962	1963	1964	1965	1966	1967	Average	
Branch				2200	2700	1001	1962-66	
20	1.70	2.52	1.78	2.77	3.01	0.75	2.36	
21	0.82	0.00	0.87	0.85	0.00	0.00	0.51	
22	1.74	1.26	8.54	7.04	7.19	0.00	5.16	
23	1.57	1.30	2.74	2.34	3.41	0.47	2.27	
24	1.14	0.90	1.03	1.97	2.05	0.15	1.42	
25	0.34	0.00	1.33	2.21	2.49	0.30	1.27	
26	0.00	0.50	0.00	1.79	0.44	0.00	0.55	
27	5.28	1.96	3.85	4.76	6.36	0.00	4.44	
28	0.89	2.97	1.66	3.29	4.82	0.00	2.73	
29	4.02	1.92	1.44	0.94	1.33	0.00	1.93	
30	0.00	0.00	1.97	1.88	1.18	0.00	1.01	
31	3.15	8.43	3.53	12.44	9.74	0.97	7.46	
32	0.00	25.00	8.82	22.22	15.79	0.00	14.37	
33	0.93	2.91	1.37	2.39	1.95	0.16	1.91	
34	43.75	64.29	20.00	47.37	40.00	4.76	43.08	
35	0.29	1.08	1.96	4.31	6.05	0.47	2.74	
36	0.00	0.68	0.83	0.96	0.62	0.00	0.62	
37	0.00	0.00	2.39	2.67	1.59	0.51	1.33	
38	0.00	0.44	1.06	1.23	2.06	0.00	0.96	
39	0.00	0.00	0.60	0.00	4.26	0.00	0.97	
A11	1.41	1.87	1.96	2.97	3.18	0.33	2.28	
Year 1	 1975	1976	1977	 1978	1979	 1980	 1981	Average
Branch								1976-81
20	0.34	3.14	1.82	1.67	1.51	1.61	0.76	5 1.75
21	1.09	4.26	5.82	11.90	3.54	15.35	9.05	8.32
22	2.15	14.29	4.35	0.00	7.20	5.06	8.75	6.61
23	0.89	10.37	2.93	3.36	2.80	7.47	4.98	5.32
24	0.48	2.93	2.46	1.55	1.07	1.44	2.09	1.92
25	0.91	2.42	5.74	3.40	2.21	4.19	1.31	3.21
26	0.00	4.46	1.96	1.35	0.52	0.75	0.00	1.50
27	4.76	20.61	17.42	11.57	8.64	4.80	4.00) 11.17
28	3.58	12.50	7.83	4.25	7.39	8.04	2.47	7.08
29	0.88	0.83	0.00	0.40	0.00	0.00	0.75	0. 33
30	1.28	10.06	1.81	3.86	6.15	2.31	10.20	5.73
31	1.07	12.37	5.21	6.99	7.04	6.46	1.24	6.55
32	0.00	17.95	0.00	2.40	6.62	14.29	3.92	7.53
33	0.66	11.80	3.25	6.41	6.48	5.46	2.34	5.96
34 3	30.77	65.00	92.50	25.78	90.44	43.75	29.41	57.81
35	1.02	9.67	4.60	2.50	6.25	5.65	3.21	5.31
36	0.24	3.30	2.80	6.18	4.35	4.91	2.66	5 4.03
37	2.77	13.17	10.78	5.88	7.23	5.11	4.78	7.82
38	1.89	12.53	13.37	6.38	8.31	16.63	5.41	10.44
39	0.00	2.98	1.19	0.00	2.64	2.10	0.00	1.48
A11	1.12	7.68	4.52	3.74	4.26	5.03	3.16	5 4.73

Note: Index calculated as [(number of strikes in branch i / number of establishments in branch i) ± 100].

				(0	lesce 1	ndi 976	ng o -198	rder) 1			
Indus Brar	strial nch	L									
Year	20	21	22	23	24		25	26	27	28	29
1976 1977	16 12	18 11	8.5 16.0	8.5 8.0	15.0 3.0	1	4.0 6.0	13.0 15.0	1 2	6 4	20.0 19.5
1978 1979	10 18	7 16	19.5 5.0	14.0	9.0 13.0	1	3.0	15.0 19.0	2 6	6 8	17.0 20.0
1980	18 18	11 6	13.0	6.0 8.0	16.0		2.0	19.0	14 5	4 16	13.5
Indus Brar	stria: nch	1									
Year	30	31	. :	32	33	34	35	36	37	38	39
1976 1977 1978	5 18 13	11 10 8	17 19 19 18	.0 1 .5 9	0.0 9.0 4.0	2 1 1	7 13 16	12.0 17.0 12.0	3 5 5	4 7 11	19.0 14.0 19.5
1979 1980 1981	4 15 2	12 7 17	2 17 7 8 7 10	.0 .0 .0	2.0 9.0 9.0	1 1 1	15 5 15	7.0 17.0 13.5	3 10 7	14 3 11	9.0 12.0 19.5

RANKING OF INDUSTRIAL BRANCHES ACCORDING TO STRIKERS DENSITY
Table 3.4.9

			RANKII	NG C	<u>)F</u>]	INDU	JSTF	IAI	L BI	RANC	CHES	
	ACC	ORD	ING TO	<u>) R</u>	ELA!	<u>rivi</u>	<u>e d</u> z	<u>Ys</u>	LOE	<u>3T]</u>	IN STRIE	<u>(ES</u>
				(de	esce	endi	ing	ord	ler))		
					-	1975	5-19	88				
Trductrie												•
Branch	11											
Vear	20	1 21	22	22	21	25	2	6	27	28	20	
			<i></i>		27 						<i></i>	-
1975	16.0) 8	14.0	12	13	6	19.	0	1	7	2.0	
1976	12.0) 19	17.0	10	13	14	7.	0	1	4	20.0	
1977	14.0) 10	13.0	12	7	5	6.	0	1	2	19.5	
1978	12.0) 3	19.5	13	8	2	14.	0	1	4	17.0	
1979	13.0) 11	5.0	14	17	18	16.	0	12	3	20.0	
1980	18.0) 15	14.0	1	17	16	19.	0	12	4	20.0	
1981	17.0) 3	10.0	2	6	5	19.	5	1	13	18.0	
1982	10.0) 14	16.0	6	9	4	12.	0	1	17	19.0	
1983	9.0) 12	5.0	11	4	20	17.	0	2	7	18.0	
1984	11.0) 3	10.0	4	15	17	12.	0	9	19	14.0	
1985	16.5	52	10.0	9	6	1	20.	0	15	19	8.0	
1986	11.0) 2	6.0	12	14	20	13.	0	16	9	17.0	
1987	12.0) 1	2.0	14	19	17	16.	0	10	13	18.0	
1988	15.0) 3	4.0	7	8	12	19.	0	11	17	13.0	
و هيا هوا خله وله خله باله جو خله											*	-
												_
Industria	 al											-
Branch												
Year	30	31	32	33	34	35	36	37		38	39	
												-
1975	4	15	19.0	9	5	11	17	3	10	0.0	19.0	
1976	6	9	16.0	11	2	5	15	3	8	3.0	18.0	
1977	16	11	19.5	8	3	15	17	4	9	9.0	18.0	
1978	9	11	18.0	5	7	16	15	6	10	0.0	19.5	
1979	4	7	19.0	9	2	15	6	1	10	0.0	8.0	
1980	13	10	9.0	6	5	7	8	11	:	3.0	2.0	
1981	7	12	16.0	14	4	11	8	9	19	5.0	19.5	
1982	15	11	18.0	5	2	8	3	7	1:	3.0	20.0	
1983	16	3	6.0	10	8	15	1	13	14	1.0	19.0	
1984	18	13	1.0	16	8	2	7	6	5	5.0	20.0	
1985	7	12	5.0	11	4	3	18	14	10	5.5	13.0	
1986	5	18	7.0	8	1	4	10	15	19	9.0	3.0	
1987	5	6	4.0	9	7	11	15	3	8	3.0	20.0	
1988	14	16	2.0	10	1	6	18 	9		5.0	20.0	_

Table 3.4.10

INTER-INDUSTRY VARIATION IN STRIKE ACTIVITY 1975-1988

Coefficient of Variation in:

Year	Strikers Density	Relative Days Lost
1975	n.a	267.98
1976	115.6	159.98
1977	164.6	175.10
1978	106.0	140.81
1979	118.1	130.19
1980	100.7	104.93
1981	136.5	215.75
1982	n.a	122.20
1983	n.a	85.65
1984	n.a	175.01
1985	n.a	96.95
1986	n.a	155.78
1987	n.a	194.91
1988	n.a	141.09
Coefficient		
of	0.55	0.25
Concordance		

Notes:

Coefficient of Variation = Standard deviation / mean, n=20.

Coefficient of Concordance = Variance of rank sums / maximum possible variance of rank sums. Called also Kendall's W statistic. By definition cannot be negative and its maximum value is 1. For further information see Hays, W. L. 1973: <u>Statistics for Social Sciences</u>, Holt, Rinehart and Winston Inc., New York, pp. 801-803.

CHAPTER 4. WAGE AND EMPLOYMENT REGULATION IN GREECE.

4.1 INTRODUCTION AND SUMMARY.

In the context of the formal system of collective bargaining, wage formation has never been a process independent of direct and indirect state intervention. The extensively used mechanism of compulsory arbitration, along with the legal right of the Ministry of Labour to reject and modify any collective agreement or compulsory arbitration award if considered as not being in line with the governmental policy, dominated wage determination processes, throughout the post-war period.

This chapter examines the types, the characteristics and the rationales of policies pursued on the basis of the extensive procedural regulation of industrial relations in Greece. These policies, which consist the substantial regulation of the Greek labour market, refer to wage setting as well as to employment adjustment, and thus set the scene for flexibilities or rigidities in the Greek manufacturing sector.

Section 4.2 discusses both the rationales and the mechanics of policies adopted in Greece over the period under examination. Section 4.3 examines wage developments and the variability in the Greek manufacturing wage structure and discusses it in the context of the question of flexibility in relative wages. Section 4.4 presents the existing legislation and the mechanics for employment regulation and examines the evolution of employment in

the Greek manufacturing sector.

Our analysis suggests that after 1975 wage determination was under the influence of two types of incomes policy. First, in 1975-81 there was the norm-based incomes policy and then, in 1982-88, indexation policies followed. Indexation policies involved relatively more compulsion. In this context of wage regulation, the wage structure, although stable as far as the distinction between high-wage and low-wage industries is concerned, became wider in 1975-81 and narrowed again in 1982-88 and, overall, variability in relative wages is observed throughout the period 1965-88.

Variability in relative wages is coupled with noticeable changes in manufacturing employment. Throughout the period 1965-88 employment in Greek manufacturing was expanding, with the most rapid expansion observed in 1967-74. But employment adjustments were regulated by the Ministry of Labour on the basis of the legislation setting limits on collective dismissals. Despite wage and employment regulation, variability in relative wages and expansion in manufacturing employment set a scene in which contradictory forces conducive either to flexibility or rigidity operated.

4.2. INCOMES POLICY IN GREECE.

4.2.1. Types and Characteristics of Incomes Policy.

In the literature, in Braun (1986) and Artis (1981) for instance, incomes policy is normally defined in terms of antiinflation measures aiming to improve the trade-off between inflation and unemployment. Measures concerned with income distribution, i.e. inequality, protection of low paid, reduction of poverty, are not comprehended by the term. But in practice such measures are often adopted as part of incomes policies. As Blackaby, quoted in Braun (1986:4), noticed "one of the general comments on the experience of incomes policy in Western Europe is that it has been impossible to evade the distributional aspect of even the simplest kind of incomes policy". In another occasion he pointed out that incomes policies "inevitably have effects on the distribution of incomes; they affect the broad distribution between employment and profit incomes, and more narrowly, the differentials between pay in different occupations and industries" (Blackaby, 1971:34).

Incomes policies vary considerably in design, sophistication, objectives and rigour. Various classifications have been proposed, but wage freezes, guidelines or norms and indexation are the types of policy most frequently used. However, even within these categories, specifications vary too. The reviews on incomes policies implemented in industrialised market economies

suggest that norms and guidelines were mostly used in the late 1950s and early 1960s, while indexation was widely used in the late 1960s and the 1970s. But, in the late 1970s, there was a general shift away from incomes policies (Blackaby, 1971, Brown, 1986, Flanagan et al, 1983).

Suzuki (1980) examined the extent of use of wage indexation in the industrialised market economies and divided the relevant cases in four categories, that is those with widespread use of indexation (Belgium, Denmark, Israel, Italy, Ireland, Netherlands, Luxemburg), those where it applied to a part of the labour force (Canada, France, Switzerland, U.S.), those where has been practically non-existent (Austria, West Germany, Japan, Sweden) and those where it was introduced and abandoned (Finland, Norway, France, Australia, U.K. Netherlands, Ireland). In a more recent work Chan-Lee et al (1987) after observing that prices and incomes policies played a comparatively minor role in the 1980s, argued that "the single most important institutional change in many European countries in the 1980s has been the weakening of explicit or implicit indexation provisions". Exception to this trend have been France and Spain, where incomes policies were adopted within a framework of ex ante indexation based on inflation targets.

In Greece, governmental involvement in wage determination has been the rule over the whole post-war period. But formal incomes policy was adopted only after 1975. Incomes policies undertaken since then fall into two categories. First, were the normbased policy of the 1975-81 period. Second, was the policy based

on partial indexation which was in operation throughout the period 1982-88. In other words, Greece too followed the exceptional pattern of adopting indexation policy in the 1980s as in France and Spain. Presumably, there must be some association with the fact that over that period Socialist parties were in power in these countries.

To classify the rationale of successive incomes policies the distinction between nominal and real policies, as proposed by Flanagan, Soskice and Ulman (1983:22), is a fruitful starting point. As nominal policies are perceived those designed to reduce the rate of inflation without restraining real wages or affecting unemployment, by attempting, for instance, to alter expectations directly. Real incomes policies are considered those which aim to restrain real wages, to bolster profitability, to improve international competitiveness or to enable unemployment to be reduced. While there is no real dichotomy between nominal and real incomes policies, their explicit or implicit objectives, as well as the criteria they are based on, suffice to classify them in one category or the other.

From this point of view, since 1975 in Greece there has been a sequence of nominal incomes policies interrupted by real ones which, in the event of severe deteriorations of the balance of payments, were aiming to create breathing space for the economy. Both schemes of policy, the norm-based and the partial indexation, were at first conceived, designed and implemented so that to reduce the rate of inflation without restraining real

wages. But both had to be radically changed to restrain real wages as well.

As far as the wage structure is concerned, the nominal policies which were pursued through the norm-based scheme were not especially concerned with the wage structure and pay relativities. Even when the governmental concern about the low paid was being announced, it was not clearly shown in incomes policy making. But, real incomes policies pursued through either the norm or the partial indexation were clearly tuned to compress the wage structure. Indeed, the partial indexation policy, even during its nominal stages, that is when it was not aiming to restrain real wages, was clearly in favour of the low paid and tuned to compress the dispersion of relative wages.

Overall, the course of incomes policy in Greece indicates the existence of a political cycle. Newly elected governments after taking-over used to award increases in order to compensate wage earners for income losses suffered during the period of their predecessors. They also used to adopt loose incomes policy in their way to re-election for a second term. And, then, after been re-elected, they used to discover the need to control real wages and cost-push inflation. It seems that, during the periods of formal incomes policy, wage determination remained part of the political game as it was throughout the post-war period. It is also noteworthy that a trend in favour of compressing pay relativities was in work and it was explicitly taken into account in incomes policy making since the late 1970s.

4.2.2. The Benchmark Period 1967-74.

The 1967-74 period, which is used as benchmark, should not be perceived as one free of wage controls. In fact, during those incomes policy promoted. Wages years there was no formal were largely being set unilaterally by the government. The right to strike was suspended by the military governments of the period. One strike of the drivers and the ticket collectors of the electric urban buses in Athens in 1971 is considered as an event of historical importance. No other strike activity was recorded. Probably, due to the political conditions, industrial conflict was not overt but covered. The massive migration abroad, while the level of unemployment was at its lowest level ever recorded in post-war Greece, suggests that an "exit" mechanism might have been at work.

The last National General collective agreement of the 1960s was signed in February 1967. As all such agreements referred to minimum wage rates, which were set at 20 per cent above those prescribed in the precedent general national collective agreement signed in December 1965. Two thirds of the increase were payable from the 1st February while the remainder was due from the 1st July 1967.

In April 1967 the military government seized the power and statutory regulation of minimum wages became the rule for wage determination. Throughout the period, the common rule in use for wage setting was based upon the growth of economy-wide

productivity in order to keep unit labour costs stable. Till 1971 the continuation of relatively loose labour market conditions helped the authorities in their aim of keeping aggregate wage increases in line with the rate of aggregate productivity advance. The floor of the wage structure was being set by minimum wages determined by the government. Collective bargaining was typical, but not essential. It was taking place between employers and industry-wide or occupation-based unions which were rather weak and had no strong links with the labour force they were supposed to represent.

Apart from the political conditions, developments in the official structure of trade-unions caused the collective bargaining processes to become rather typical and sterile. Trade-union officials were mainly governmental appointees. Indeed, collective agreements were subject to review by the Ministry of Labour either through legislation entitling the government to reject and/or modify any collective agreement and compulsory arbitration award or through the direct control of the compulsory arbitration courts. In fact, the government was influencing wage developments in two ways. First, through statutory adjustment of the minimum wage rates. Second, through dictating these adjustments as guidelines which had to be followed by the Ministry of Labour, the compulsory arbitration courts and the appointed trade-union leaders.

In those conditions, after the general national collective agreement signed in February 1967, the legal minimum wage was increased by 7 per cent on 1st October 1968 and by 8 per cent on

1st May 1969, with no further adjustment in 1970. In the early 1970s, an OECD (1971:14) report pointed out that "avoidance of wage inflation partly explains Greece's success in containing price inflation". In the same report, it was intuitively noted that "the possibility that the causal connection runs the other way cannot be discarded". A year later similar comments praised the Greek performance in the field of wage and price inflation. "Greece's favourable inflationary performance throughout the recent phase of generalised inflation abroad no doubt owes much to the policies pursued on the wage/price front. In respect of wages the aim has been to contain average nominal increases to the rate of productivity growth. Success in this field has been greater that in most member countries for various reasons -an important one having to do with the relative strength or weakness of trade-unions in different countries" OECD (1972:23).

While labour prices were controlled, labour quantities were not. Large scale immigration of Greek labour into Western Europe in the late 1960s and the early 1970s can be considered as a response of the labour force to such a statutory regulation of wage adjustments. Despite the declared rationale that wage rates should follow changes in productivity, they were in 1968-69 and 1972 falling behind rises in productivity, creating thus the higher ever recorded rates of profits in the Greek manufacturing sector. Presumably, political conditions along with favourable job prospects in European labour markets for migrant labour, mainly in West Germany where the Greek labour force had created a long tradition since the late 1950s, encouraged skilled workers

to quit the country.

Labour's option for exit caused the employers to bid up pay, thus increasing average wage rates. But the tightening of the labour market also forced the military governments to speed up adjustments in minimum wage rates. During 1973 three increases in minimum wages were awarded. In May the legal minimum wage for male workers was increased by 8.7 per cent; in September it was raised by 10.4 per cent; and in December a further 20 per cent increase was awarded. Nevertheless, labour shortages were not avoided. During the 1971-74 period, shortages of skilled and semi-skilled labour were being reported. The 1967-74 period ends with the restoration of a civilian government in July 1974. In September legal minimum wages for male workers were raised by 10 per cent so that employees be compensated for the accelerating inflation and income losses suffered during the period of the military governments.

To summarise, this period was one with neither free operation of the labour market nor formal incomes policy at work, with the government regulating directly wage determination. Indeed, there were no representative trade-unions in the labour market and no essential collective bargaining processes.

4.2.3. The Period 1975-81: Norm-based Policy.

Throughout the period 1975-81, with conservative governments in office a norm-based incomes policy was in operation. Successive governments aiming at decelerating wage inflation used to announce in the beginning of each year, when new annual collective agreements were to be contracted, a norm or guideline for nominal wage increases in two semestrial instalments.

As far as wage determination processes are concerned, the legal right of the Ministry of Labour to reject and/or modify collective agreements and compulsory arbitration awards was abandoned and the right of collective bargaining was reestablished on the basis of the legislation observed in the mid-1960s, before the military regime. Minimum wage rates became again negotiable. They were being determined either by the General Confederation of Labour and the Employer's Associations or by the compulsory arbitration courts. The minimum wage rates were being established before any other rates and were serving as guideline for subsequent negotiations. Nevertheless, new appointees were during that period leading the General Confederation of Labour and the official trade-union structure.

The pattern of annual increases through two instalments was followed in minimum wage setting throughout the 1975-81 period. The governmental influence on wages determination was substantial for two main reasons. First, since the early 1970s, the public sector in its wide sense (i.e. including civil

Table 4.2.1

THE NORM-BASED INCOMES POLICY 1975-1981

Year	Inflation Jan to Jan change	Norm for Increases in Minimum Wages	Special Characte- ristics	Implemented via	
1975	13.4%	12% :1 March 8% :1 Aug.		General National Coll. Agr.	
1976	13.3%	10% :13 Feb. 7% :1 July		Compulsory Arbitration	
1977	12.2%	10% :15 Feb. 5% :1 July		General National Coll. Agr.	
1978	12.5%	16% :1 Feb. 6% :1 July		Compulsory Arbitration.	
1979	19.0%	15% :l Feb.	Only the low paid to get the full 15%	Compulsory Arbitration.	
1980	24.9%	15% :1 Feb.	Only the low paid to get the full 15%	Compulsory Arbitration.	
1981	24.5%	19% :1 Jan. 7% :1 July	Threshold clause for inflation higher than 20%	The norm for public sector used as guide- line in private sector.	

service and local government, public corporations and entities, the central bank, commercial banks and companies under state ownership), has been employing more than 35 per cent of employees in employment. In 1987, according to the annual Labour Survey, public sector employment amounted to 41 per cent. Second, in the private sector, formal collective bargaining was restricted at the secondary and tertiary level of the official trade-union structure. Indeed, the mechanism of compulsory arbitration was supposed to control any deviation from the norm for annual increases in wages.

During the 1975-77 period, in contrast to many other European and OECD countries, Greece was pursuing expansionary demand management aimed at achieving high levels of economic activity and employment. In technical terms, there were two main criteria for setting the norm : changes in consumer prices and productivity growth. In fact, as an advisor of the government at the time argued, the norm was ad hoc dominated by a rationale favouring the maintenance of real wages and salaries above their early 1970s levels (Economou, 1981:207). The economic criteria for setting the norm were becoming vague on its way from conception and design to announcement, as political criteria were taking priority.

In 1975, the new civilian government moved to what they considered as a cautious expansionary policy aimed to overcome the 1973-74 recession. In 1975 the minimum wage was increased by about 20 per cent, of which 12 per cent was effective on 1st March and the rest 8 per cent was scheduled for 1st August. In

1976 the norm for wage increases was ratified by the compulsory arbitration court which awarded a 17 per cent increase in minimum wage rates, again in two instalments. A 10 per cent increase was effective on 13th February with a 7 per cent becoming effective on 1st July. For 1977, in view of inflationary risks, the authorities decided to slow down increases in wage rates to a range below 15 per cent. Only the low paid were supposed to receive the full 15 per cent increase. Thus, minimum wages increased by 10 per cent on 15th February and 5 per cent on 1st July 1977. Favouring the low paid in order to reduce gradually any large disparities in pay became more emphasised in 1977 but was already declared as an aim of the wage policy in 1975.

In 1978 the view was adopted that the most important immediate task of economic policy was to bring down appreciably the rate of inflation. Domestic factors, notably the substantial annual nominal rises of wages and salaries were blamed. As the inflation rate remained comparatively high and accelerating in the first half of 1978, the government changed its stance and introduced more restrictive fiscal, monetary and incomes policy.

The anti-inflationary policy placed the emphasis on the moderation of labour cost increases. The norm for the annual increase in the minimum wage rates in 1979 was announced nearly seven months before the discussions for the next pay round were to begin. The pro-governmental leadership of the General Confederation of Labour was reluctant to sign a general national collective agreement involving the norm. The case was referred to

compulsory arbitration, which awarded an increase in line with the norm. In 1979, instead of the usual increase in two semestrial instalments, an increase of 15 per cent was granted from the beginning and for the whole year. Under conditions of accelerating inflation an identical incomes policy was adopted in 1980. The norm was set for a 15 per cent increase from the 15th February. In 1979 and 1980 the government made explicit the aim to stop the norm for increases in minimum wages from going automatically into every pay settlement as if the norm was only the floor for further collective bargaining. The emphasis was set so that to compress the wage structure.

Incomes policy in 1981, year of elections crucial for the conservative governments in office since 1974, was designed with two rather contradictory aims: first, to combat inflation, and second, to do it without leading to a further sharp reduction in real earnings as it had happened in 1979 and 1980. On the basis of a forecast for an increase in consumer prices during the year lower than 20 per cent, the norm was set for a 12.5 per cent increase in January and a further 7.5 per cent increase effective on 1st July. A threshold clause was included in this year's policy. The authorities announced that there would be an additional increase by the end of the year if inflation were to exceed the 20 per cent forecast. In the October 1981 election the conservative government was defeated and thus ended the period of norm-based incomes policy.

To summarise, this period was one with a formal incomes policy based upon the announcement and the promotion of the governmental

norm for annual increases in nominal wages. Successive governments were trying to influence wage determination rather indirectly in comparison with the previous period. The compression of the wage structure became an explicit target of the norm-based incomes policy only after 1978.

4.2.4. The Period 1982-88: Partial Indexation Policy.

In the period 1982-88, with socialist governments in office, systems of partial indexation of nominal wages to the consumer prices index on a trimestrial basis were in operation. Nevertheless, in the same period, there were two wage freezes, in 1983 and 1985, and important amendments of both the plateau and the index used for the automatic and partial indexation.

In 1982, the new socialist government, to offset the erosion of real earnings of employees suffered in the 1979-81 period, and to fulfil electoral pledges, awarded substantial increases mainly for the lower paid wage and salary earners which took effect on 1st January 1982. Minimum wage rates were increased by about 32 per cent by fiat. Moreover the new government changed its stance with respect to incomes policy's rationales and mechanics. To preserve real wages, it introduced an indexation system for fourmonthly adjustments on the basis of the consumer prices rises in the preceding four months. The indexation policy was also tuned in favour of the lower paid. Full indexation was provided for average monthly earnings up to Dr.35000. The band of earnings between Dr.35000 and Dr.55000 was indexed only to half of the

consumer prices increase. The next band, between Dr.55000 and Dr.80000 was entitled a 25 per cent indexation, while no indexation was provided for the band of monthly earnings above Dr.80000. This system of formal and partial indexation took effect in January 1982 with the first four-monthly adjustment due on 1st May.

As the wage increases awarded in 1982 along with deterioration in productivity performance fuelled the wage-price inflationary spiral, in 1983 the government was obliged to modify, only for 1983, the indexation scheme introduced the very previous year. It was decided to postpone the automatic adjustment due on 1st January 1983 for the rate of inflation over the previous four months. Only one half of this adjustment, namely 4.5 per cent, was awarded on 1st January with the second half scheduled for the 1st of May. The automatic adjustment due on 1st September was to compensate for consumer price increases during the first eight months of 1983. There was no modification of the scales for partial indexation. Indeed, those arrangements became, by law, compulsory for both the private and the public sector. Penalties were to be applied to those firms exceeding the guidelines for partial and delayed indexation. But there was no real mechanism to control for compliance with and evasion from the policy. This is why no real sanctions were imposed. Sanctions were speculative to ease the implementation of the wage freeze.

In 1984 and 1985 the indexation scheme was followed in its original version as it was conceived and adopted in 1982.

Table 4.2.2

THE INDEXATION INCOMES POLICY 1982-1988

/			_ ~ ~ ~ ~ ~ ~	\
Year	Annual Inflation	Automatic Increases in Wages Freezes	Scales of Indexation (in drachmas)	Implemented via
		rieezes	(In drachmas)	
_				
1982	21.0%	5000dr:1 Jan 7.3%:1 May	100% :<35000	Decree. Compulsory
		2.0%:1 Sep	50% :35000 - 55000	Arbitration.
1983	20.2%	8.8%:1 Jan		Compulsory
		Wages Freeze for 8 months	25% :55000 - 80000	by Decree.
		10.2%:1 Sep	0% :>80000	
1984	18.5%	8.9%:1 Jan		General National
		6.9%:1 May		Coll. Agr.
		2.2%:1 Sep	100% :<50000	
1985	19 38	8 28.1 Tan		Conoral National
1905	17.00	6 68.1 May	508 .50000 -	Coll Agr
		2.18.1 Sen	75000	coll. Agr.
		Wage Freeze	75000	Compulsory
		for 3 months	25% :75000 -	by Decree.
		Index	10000	Dy Deerce.
		Modified	100000	
1986	23 08	4 59.1 Tan	09	Conoral National
1900	23.04	1 49.1 May	03 ./100000	
i		4 78.1 May		COII. AGI.
		4./%.1 Sep	~	
1987	16.4%	4.1%:1 Jan	100% :<60000	Compulsory
		1.0%:1 May		Arbitration.
		4.5%:1 Sep	50% :60000 - 75000	
1988	13.5%	8.5%:1 Jan		General National
		1.0%:1 May 6.5%:1 Sep	25% :75000 - 100000	Coll. Agr.
			0% :>100000	

Automatic pay adjustments were provided every four months. Partial indexation remained strongly in favour of the lower paid, with slightly modified scales, as full indexation was offered to monthly earnings up to Dr.50000. The next band covered earnings between Dr.50000 and Dr.75000. The band of earnings entitled a 25 per cent indexation was set between Dr.75000 and Dr.100000.

In 1982-85, unit labour costs were increasing rapidly, leading to a severe deterioration in the current account. It was then realised that wage rises every four months in line with the inflation performed during the past four months were not allowing inflationary expectations to decelerate and, indeed, were not allowing sufficient time for any productivity gains to exert any significant downward pressure on prices before the next rise in wages was to take place. But political calculus rather than economic considerations led the government to maintain the system of indexation in its original version. The government was then trying to compensate for rising unit labour costs through slow devaluation of the national currency.

After the socialist government was re-elected in the June 1985 election, an economic stabilisation programme was prepared. It was introduced in October 1985. One of its four main measures was a wage freeze for the last quarter of 1985 along with a modification of the indexation scheme. The rationale for the modified indexation scheme was to decelerate increases in unit labour costs. The indexation scheme underwent two-fold modification. First, the impact of import prices was excluded from indexation. Second, instead of adjusting wages on the basis

of the past price rises, the adjustment was, from January 1986, to be on the basis of the projected inflation. Corrective adjustments were announced for the end of 1986 and 1987 if the increase in consumer prices exceeded the projected inflation. This modified indexation scheme was also made, by law, compulsory for both the public and the private sector up to the end of 1987. Penalties were to be imposed to firms exceeding the guidelines for indexation. But these penalties were again notional due to the lack of effective mechanism for controlling compliance with and evasion from the guidelines for partial indexation. Scales for partial indexation remained stable in 1985 and 1986. In 1987 the limit for full indexation was raised to monthly earnings not exceeding the amount of Dr.60000, with the average level of monthly wage and salaries being, in 1987, just over Dr.70000.

In 1988, after the end of the stabilisation programme, the modified system of indexation was maintained and was used as the formal and official incomes policy. Wages were adjusted to projected inflation and a corrective percentage increase was due at the end of the year. The main elements in the package announced in November 1987 were : a 4.5 per cent increase in pay in the public sector to compensate for loss of purchasing power in 1987 (which was expected to produce a similar concession in the private sector where all restrictions on bargaining were lifted); freedom for employers to give productivity bonuses at the end of 1988 based on performance throughout the year; continuation of the indexation system with "imported inflation" being included in the calculations for pay up to Dr. 60000 and a

payment of the first automatic indexation increase from 1 January 1988. The incomes policy's emphasis remained on favouring the low paid.

The level of the adjustment which was to be made at the end of 1987 became a matter of political debate. The package agreed and announced by Mr. Simitis, the Greek Minister of National Economy, provided for that first increase to be delayed until May 1988, but Prime Minister Papandreou reversed the cabinet decision and this prompted Mr. Simitis' resignation as the government adopted a loose policy by awarding substantial increases in the beginning of 1988.

To summarise, during this period of formal indexation, incomes policy was at first tuned so that to preserve automatically real wages. Indeed, it was clearly oriented in favour of the low paid. But falls in productivity and rises in unit labour costs obliged the authorities to impose two short-term wage freezes and, then, to modify the indexation scheme and impose it through statutory regulation. Thus, the socialist governments which were claiming that they would protect real wages, mainly of the low paid, had to recourse to direct intervention in wage determination to implement incomes policies aimed to restrain real wages.

4.3. WAGE DEVELOPMENTS IN GREEK MANUFACTURING.

<u>4.3.1. Wages and the Wage Structure across Manufacturing</u> <u>Industries.</u>

Given the policies described in Section 4.2, Tables 4.3.1 and 4.3.2 illustrate the labour market conditions prevailing over the period 1965-88 and indicate the basic developments in the Greek manufacturing labour market as far as wages and unemployment are concerned. Developments regarding the nexus nominal wagesproductivity-unit labour costs indicate the distinct characteristics over the periods under consideration. Productivity growth slowed down in 1975-81 and became negative in 1982-88. Unit labour costs soared in 1975-81, following the increases in nominal wages, and were slightly moderated during the 1985-87 stabilisation programme as increases in nominal wages were also restrained.

As far as real wages are concerned, while in 1975-81, compared to 1967-74, productivity growth slowed down, real wages rise at a rate twice that of productivity. In 1982-88, along with the negative productivity performance, real wages remained stable as the real incomes policy of the 1985-87 stabilisation programme counter-balanced the impressive increases awarded in 1982 and 1984. In other words, while both periods of norm-based and indexation policy are characterised by soaring unit labour costs, during the norm-based policy increases in real wages

exceeded productivity growth and during indexation policy negative productivity growth led to stagnating real wages. However, the course of real wage during the indexation period is less smooth than in earlier periods as is characterised by regressive moves. Distinct unemployment levels characterise also the periods under consideration. In 1967-74 and 1974-81 low levels of unemployment are observed, with the labour market being relatively tighter in 1974-81. But the period 1982-88 is characterised by relatively high unemployment.

The minimum wage as percentage of average wage played an important role in 1967-74, as it represented the 70 per cent of but its ratio decreased in 1975-81 and in the average wage, 1982-88 to levels around 55 per cent. These levels and trends are comparable to those prevailing in most industrialised market economies. In the late 1970s the relevant ratio was in the U.S. 1978:50 per cent, in Canada 1977:51 per cent, in Japan 1975-77:50 per cent, in Australia 1978:52 per cent, in France 1977:63 per cent and in the Netherlands 1977:75 per cent (Starr, 1981:42-47). But in the 1980s statutory minimum wages as a percentage of average earnings in the 1980s play a reduced role, though they remained highly significant in the Netherlands, Spain and France, where in 1984 they were equivalent to respectively 65 per cent, 41 per cent and 37 per cent of average manufacturing earnings It appears that in Greece, compared to OECD (1987:41). countries such as Spain and France, with socialist governments in office following indexation policies, minimum wages played a relatively more important role.

At the disaggregated level one interesting feature of the manufacturing wage structure is the high degree of stability over time in the hierarchy of relative wages in the twenty twodigit industrial branches. The coefficient of concordance in the ranking of the manufacturing wage structure for male workers in the period 1965-1988 is 0.82. This value means that male employees of certain high-wage industrial branches in Greek manufacturing maintained, throughout this period, their higher position in the wage structure.

The ranking in descending order of the average pay for male workers, as presented in Table 4.3.4, shows clearly that the composition of the top five high-wage industrial branches is very stable. Membership is restricted to the following branches : i) 34: Basic Metal Industries, ii) 32: Products of Petroleum and Coal iii) 28: Printing and Publishing, iv) 31: Chemicals, v) 38: Transport Equipment and vi) 33: Non-Metallic Mineral Products. In parallel, the club of the bottom five low-wage industrial branches has also stable composition and permanent members such as : i) 26: Furniture, ii) 25: Wood and Cork iii) 39: Miscellaneous Manufacturing iv) 36: Non-Electrical Machinery and v) 24: Clothing and Footwear.

This stability at the top and the bottom of the wage structure hierarchy is not a Greek-specific phenomenon. Marsden (1981) and Saunders and Marsden (1981) showed that in the 1970s there was an overall stability of inter-industry ranking in West European countries. Roughly the same industries were found among the

high-wage and among the low-wage industries in each country. The "energy" sector and printing dominated the high-wage industries, and textiles, clothing, and footwear were the most common among the low-wage. A more recent study by OECD (1985:92) reported findings consistent with previous research as at the lower end of the earnings distribution there was considerable uniformity across countries: textiles, clothing, and food and beverage industries dominated the low end. Moreover, while there was less uniformity across countries among the high-wage industries there was a tendency for chemical, primary metals, petroleum-refining and transportation equipment industries to be consistently in the high-wage sector.

Marsden (1983) reviewed the evidence on the factors accounting for different pay levels across industries. These factors include the relative skill composition of the industry labour force (skilled labour is paid better), the working conditions (wage differentials compensate for worse conditions), the industrial concentration in relatively few large firms (large firms pay better) and the sex composition of the labour force (female labour is paid less than male).

These factors account largely for different pay levels observed across the twenty two-digit industrial branches in Greek manufacturing. Table 4.3.5 presents estimates of the determinants of the manufacturing wage structure, based on cross-section regression analysis. Four variables, namely productivity, size, skill (as proxied by the share of salary earners to both salary and wage earners) and sex composition, are used as independent

variables in order to explain the manufacturing wage structure across twenty two-digit industrial branches. Size, skill and sex composition were found significant in 1981 and, indeed, their influence was stronger than 1976. In other words, factors such as size, sex and skill showed themselves up over the period 1976-81, during which the norm-based incomes policy was in use, strike activity soared and the informal system of collective bargaining became at work.

The inter-branch variance of hourly earnings can be described by the coefficient of variation. When the coefficient of variation is used as a summary indicator of wage differentials, the manufacturing wage structure in Greece, compared with other OECD countries, is classified in the category of those countries with considerable fluctuation -upward and downward- in the dispersion of inter-industry wages (OECD, 1985: 85). The OECD study arrived to this conclusion by examining the variability in wages for both male and female workers over the period 1975-83. Our analysis, presented in Table 4.3.6, arrived to the same conclusion by using wages of both sexes, of male workers and of female workers for a longer period that the OECD study, namely for 1965-88. However, throughout the period 1965-88, variability of wages across manufacturing was far greater in the case of male workers than in the case of female workers. Overall, the wage structure was rather narrow in 1967-74, became wider in 1975-81 and was compressed again in 1982-88 near its 1967-74 level.

In other European countries over the 1970s, as Saunders and

Marsden (1981:126) observed, the size of the differentials between industrial branches changed towards a very moderate narrowing. Only in Italy, where the differentials were rather wide in the early 1970s, was there a substantial narrowing, which appears to reflect the equalising policy of the unions. And only in Germany, where the differentials were relatively narrow, was there a widening. Thus in Greek manufacturing during the 1970s, and especially over the period 1975-81, a rather different course than in other European countries was followed. The widening of the wage structure in 1975-81 coincides with the use of the normbased incomes policy as well as with the rise of strike activity and the creation of the informal system of collective bargaining.

4.3.2. Indices of Flexibility in Manufacturing Relative Wages.

When changes in relative wage ranking are used to measure the stability of the manufacturing wage structure, it appears that industry wage ranking in the Greek manufacturing is, as in the case of other European countries, relatively less stable, compared with non-European OECD members (OECD, 1985:85). The OECD study examined the ranking in industry wages for workers of both sexes. Our analysis of industry wages for all three categories of wages, presented in Table 4.3.6, found that the inclusion of wages of both sexes provides higher coefficients of variation indicating higher variability, as well as higher coefficients of determination indicating less unstable ranking. This implies that the existing variability does not affect the hierarchy as far as ranking is concerned, but evolves within the limits of the

existing hierarchy.

A different view on variability in industry wages is taken by focusing upon the evolution of the relative wage in each industrial branch separately, over the period 1966-81 and the sub-periods of distinct incomes policies. From Table 4.3.7, which presents the variability of relative wages over time in each of the twenty two-digit industrial branches, we can infer that there was no great variability in branch relative wages over the period 1966-88. According to Table 4.3.7, over the period 1966-88 only one industrial branch had a coefficient of variation exceeding 10 per cent. Six branches had a coefficient of variation exceeding 5 per cent. The majority, 13 branches, had a coefficient of variation which did not exceed 5 per cent.

The annual coefficients of variation in relative wages across manufacturing branches denote the existence of a rather flexible wage structure. But the branch coefficients of variation in relative wages over time indicate the existence of less variability in industry relative wages. This slight mismatch is of course due to methodological reasons as not exactly the same kind of variability is examined in each case. Annual inter-branch coefficients of variation refer to the positions of industrial branches relative to the average during a certain year. Branch coefficients of variation refer to the variability of industry relative wages over time with regard to their all-period average relative wage. Therefore the findings in Tables 4.3.6 and 4.3.7 indicate that while variability exists, does not result in

important changes in ranking as high-wage branches maintain their top positions and low-wage branches remain at the bottom of the hierarchy even if the gaps between them widen or narrow.

The examination of three sub-periods, selected according to the type of wage policy in force, leads to the identification of distinct levels of variability in the wage structure. At the aggregate level from Table 4.3.6 we infer that wage dispersion was wider in 1975-81 than in 1967-74 and 1982-88 for wages of both sexes and wages of male workers as well. At the disaggregated level, Table 4.3.7 where coefficients of variation estimated for the three sub-periods of distinct wage policies are compared, indicates that the sub-period 1968-74 is characterised by higher over time variability in relative wages of the majority of industrial branches. It is during that period that, in the context of rapidly expanding manufacturing employment, most changes in the relative position of branches occurred. During the sub-period 1982-88 relatively lower variability is observed, whatever happened in the precedent sub-periods. This suggests that, even if over the full period 1966-88 there was some, though small, "restructuring" in the wage structure, in 1982-88, a period during which growth in manufacturing employment slowed down and indexation policies were in use, this variability becomes smaller.

The examination of separate sub-periods suggests the existence of three patterns of over time variability in branch relative wages. First, there is a pattern of variability diminishing period after period. This pattern is mainly followed by low-wage

branches. Second, there is a pattern of variability peaking during 1975-81. This pattern is mainly followed by high-wage industries which display the higher variability in their relative wage during 1975-81. Third, the remaining branches, that is the majority, had the lowest variability in their relative wages during 1975-81. These centrifugal tendencies indicate that, when the evolution of relative wage of each industrial branch is considered separately, for the majority of them, the sub-period 1975-81 appears to be pivotal, in the sense that it is the one of the wider disturbances in the manufacturing wage structure as far as variability across manufacturing and over time are concerned.

Of course we must recognise the limits of indices such as the coefficient of variation in relative wages for assessing any flexibility in the wage structure. For such an assessment to be valid the wage structure should be related to other labour market outcomes such as changes in employment, productivity as well as trade-unions and collective bargaining developments. As the analyses with regard to the Australian wage structure by Keating (1983) and Withers et all (1983) suggest, there was more variation in wage relativities than it was usually alleged, even if the labour force responded to quite small adjustments to industry wage relativities. Indeed, as quantity adjustments (employment) are usually more important than price (wage) adjustments, an important role can be played by small variations in relative wages. From this point of view, simple indices of variability in relative wages when examined alone may understate

the actual flexibility in the wage structure with respect to the labour market outcomes.

To summarise the analysis of variability indicators, both at the aggregate and the disaggregated level, suggests that there is some flexibility in the inter-industry wage structure which should be examined. Moreover, the sub-period 1975-81 seems to hold a pivotal position as far as disturbances in the manufacturing wage structure are concerned. This seems consistent with the norm-based wage policy followed in 1975-81, which incorporated relatively less compulsion, as well as with the rise of the informal system of collective bargaining which took place then.

4.4. EMPLOYMENT REGULATION AND TRENDS IN EMPLOYMENT IN GREEK MANUFACTURING.

<u>4.4.1. Employment Regulation and the Legislation on Collective</u> <u>Dismissals.</u>

Employment protection and growth has been one of the main policy concerns in Greece during the post-war period. In the mid-1950s, in conditions which were considered as conditions of surplus labour in the primary sector of the economy, wage controls were aimed at securing employment growth in the manufacturing sector. Indeed, over the post-war period, while wage and employment policies were aimed at employment growth, stability in employment was considered as a principle which had to be met at any price and against any macro- or micro-economic rationale. The recurrence of the phenomenon of ailing industries nearly after every wave of important disturbances in the internal and external balance of the economy, and the governmental treatment of the recurring redundancies problem, have made clear that politicians, across the board, have been at odds with any alternative policy to subsidising ailing businesses, inefficient managements and insecure jobs -insecure in terms of market conditions.

Nowadays the question of flexibility with regard to employment regulation in the Greek labour market is mainly raised concerning the existing legislation on collective dismissals, on the basis of which labour adjustments have been taking place. The relevant

legislation was updated in 1983 to become in accordance with the EEC directive, adopted in 1975, on collective dismissals. In Law 1387/1983 strict limits are set for the maximum number of employees who could be dismissed per calendar month. According to it the collective dismissal of employees is permitted, for reasons not related to their individual contract, for a maximum of five employees in companies with between 20 and 50 employees, or for a maximum percentage of the employees in companies with more than 50 employees. For a dismissal to be considered as collective this percentage must be at least 2 per cent. The upper limit of permitted dismissals should not exceed the 10 per cent of the total number of employees, and it is subject to a maximum of 30 workers. The percentage of permitted dismissals is revised, after taking into account the situation of the labour market, at six-monthly intervals by Ministerial decision. For a long period stood in the range from 2 to 3 per cent.

Dismissals exceeding the afore-mentioned limits are deemed invalid and termination of the relevant labour contracts is considered as non-existent if, one month after their notification to the Manpower Employment Office, they have not been approved by the Minister of labour. Such approval is granted following an application by the employer which must submitted within 8 days of the notification of dismissals.

The main important change brought about by the updated legislation refers to requirements to inform and consult employee representatives and to tightened up provisions on notification and authorisation of collective dismissals. Authorisation of the

collective dismissal by the Ministry of Labour is essential for it to be lawful allowed and not revoked. According to the earlier legislation, before the introduction of Law 1387/1983, it was permitted, per calendar month and for economic reasons, the collective dismissal of 2 to 10 per cent of the workforce in firms normally employing at least 50 employees. The exact percentage was also determined and revised every six months (on 1 January and 1 July each year) by the Ministry of Labour. But, regardless of changes in legislation, for a long period stood, as already mentioned, between 2 and 3 per cent. This percentage has represented the maximum permitted level by which employers may reduce the size of their workforce, although the Ministry can waive this rule where firms find themselves in "special circumstances".

Nevertheless, for a long period, few permissions for exceeding the legislated maximum has been granted by the High Council of Labour, the advisory body which examines applications for exemption from the legislated limits and proposes to the Minister of Labour whether these applications should be accepted or not. Tables 4.4.1 and 4.4.2 provide information on cases of collective dismissals under the Minister of Labour's and the Perfect's jurisdiction respectively over the period 1983-86. In seven cases, out of the 29 reported, authorisation for exceeding the legislated limits was granted. Three of them, in 1986, involved the complete shut down of companies or plants. But this type of treatment was exceptional.
Indeed, this type of corporate economic adjustment was exceptional either. Ailing firms in need, inter alia, for labour adjustment became under the control of the state-owned and run "Organisation for Companies Rehabilitation" (OAE) set up in 1983, which since then controls firms employing more than 30000 salary and wage earners, i.e. nearly 10 per cent of the total employment of salary and wage earners in large-scale manufacturing industries. Natural wastage has been the only manpower policy used to minimise the excessive labour hoarding in the "problematic" firms.

The legislation on collective dismissals may have caused some kind of rigidity as far as the adjustment of employment at the firm level is concerned. Certainly, firms wanting to adjust employment downwards and cut the size of their workforce may rely on natural wastage. But the legislated limits may have also affected the way via which firms adjust employment upwards, in cases of short-term increases in demand for their products. The restrictive legislation on collective dismissals may have caused firms to channel their higher demand for labour towards higher overtime work rather than towards hiring more employees. It may also have worked in favour of more flexible forms of contracts such as subcontracting, short-term or fixed-term contracts etc. Therefore, the issue of labour adjustment consists the second half of the scissors by which firms adjust their workforce to short-term and long-term changes in the market. Indeed, it seems that in the Greek labour market, employment regulation has been stricter than wage regulation.

<u>4.4.2. Manufacturing Employment: Change, Characteristics and</u> <u>Distribution.</u>

The examination of changes in aggregate manufacturing employment as well as at a disaggregated level of industrial branches provides information about the conditions prevailing in the manufacturing labour market over the period under consideration.

Since the mid-1960s, aggregate employment in the Greek manufacturing sector increased in both absolute and relative terms. From this point of view, Greece, compared to the majority of OECD and EEC countries, is not a typical case as in those countries since the mid 1970s manufacturing employment has been falling. In Tables 4.4.3 and 4.4.4 we observe that since 1973 in the majority of OECD countries the average annual rate of change in manufacturing employment has been negative, and that the share of manufacturing employment in civilian employment has been decreasing. In both cases Greece has been exceptional. The average annual change slowed down but remained positive and the share of manufacturing continued increasing.

A recent OECD (1989:163) study pointed out that the historical experience suggests that, throughout the process of economic development, the share of industry in total employment follows an inverse U pattern. This pattern has been followed by most of the OECD countries as in the 1960s the share of industry employment rose, in the late 1960s peaked, and started declining after the first oil-shock in 1973. As Greece and Turkey experienced

increases in industry's share in total employment, the study argued that this is compatible with the inverse U pattern as Greece and Turkey, countries with relatively lower levels of economic development, they have been, over the 1970s and the 1980s, in the upward side of the inverse U pattern.

The overall increases in manufacturing employment are not to say that employment was constantly expanding throughout the period 1965-88. Behind this increase, distinct periods can be discerned. Table 4.4.5 indicates that high rates of growth in manufacturing employment were observed from 1969 till 1974, in the aftermath of the first oil-shock. Then, till the second oil-shock in 1979, employment growth in manufacturing slowed down. A third phase of falling manufacturing employment followed. This fall was controlled, to a certain extent, by the stabilisation measures adopted in 1983 and in 1985-87.

When three different sub-periods, selected according to the wage policy in force, are examined, it appears that the sub-period 1967-74 was one of strong expansion in employment across the Greek manufacturing sector, clearly distinguished from the rest. After 1969 and till 1974, aggregate manufacturing employment was persistently increasing at the highest rates of the whole period 1965-88. In 1975-81, the increasing trend in aggregate manufacturing employment became weaker. Employment growth in manufacturing gave way to a fragile stability.

To discern whether the impressive increases in manufacturing employment in 1970, 1974 and 1983, came from expanding

establishments or new ones we examined the information provided by the Annual Industrial Surveys with regard to establishments and employment in large-scale manufacturing. Unfortunately the last one published and available refers to the year 1981. Therefore we can only discuss the impressive increases, exceeding 10 per cent, in 1970 and 1974. From 1969 to 1970 the number of large-scale manufacturing establishments (i.e. those employing more than 10 employees) decreased from 6356 to 6039. By contrast, from 1973 to 1974 the number of establishments increased from 6529 to 7767. It appears that while the 1969/70 increases came from existing establishments which expanded, the 1973/74 increases were caused by new establishments.

To trace further the sources of those increases in manufacturing employment, we examined the relevant information at the disaggregated level of two-digit branches. In 1969/70 in ten industrial branches employment increases exceeded the allbranches average increase of 10.76 per cent. In half of them the number of establishments increased either, while in the rest fewer establishments were recorded in 1970 compared to 1969. In 1973/74 in five manufacturing branches employment increases were above the all-branches average increase of 10.44 per cent, and in all of them the number of establishments increased too. Therefore it appears that the impressive expansion in manufacturing employment in 1970 and 1974 was largely due to the creation of new manufacturing establishments rather than to expanding old ones. This indicates that the workforce in those green-field sites was presumably involved in setting up from the very

beginning new industrial relations as neither had a labour tradition and culture, nor was incorporated in already existing bargaining and trade-union structures, and, indeed, this happened in the context of the military regime.

As far as the composition of the manufacturing employment is concerned, Tables 4.4.6 and 4.4.7 suggest that over the period 1965-88 the percentage of salaried employment in the total number of manufacturing employees was steadily increasing. In fact, from 1965 to 1988, it was doubled. It was 17.1 per cent in 1965 and became 34.9 per cent in 1988. However, over the period 1976-88 the share of female labour in total manufacturing employment of salary and wage earners was rather stable around the level of 36 per cent.

At a disaggregated level, from 1965 to 1988, as Table 4.4.8 indicates, 15 out of 20 industrial branches increased their employment of wage earners. However, rates of expansion varied considerably over time and across the twenty two-digit industrial branches. From 1967 to 1974 employment of wage earners increased in 18 industrial branches. As Tables 4.4.5 and 4.4.11 indicate, for every single year during the 1967-74 period, on average, 14 industrial branches were increasing their employment of wage earners.

A different pattern characterises the sub-period 1975-81. From 1975 to 1981 6 industrial branches increased the employment of wage earners. Over the same sub-period and for every single year, on average, 7 industrial branches were expanding employment of

wage earners. As the increasing trend in aggregate manufacturing employment became weaker it seems that, in 1975-81, employment growth in certain industries was counter-balanced by job losses in others.

During 1982-88, employment performance of manufacturing was rather contradictory. From 1982 to 1988 11 industrial branches increased the number of wage earners employed by them. For each single year, on average, 8 industrial branches were expanding their employment of wage earners. However, as negative developments dominated at the level of aggregate manufacturing employment this sub-period, it seems that employment growth in certain expanding industrial branches was outnumbered by job losses in contracting branches.

The distribution of employment in Greek manufacturing throughout the period 1965-88 is highlighted by Tables 4.4.6-4.4.11. Overall, the distribution of manufacturing employment across industrial branches is shaped by the increasing dominance of four industrial branches. From Tables 4.4.9 and 4.4.10 becomes clear that, on average, four industrial branches, namely 20:Food Industry, 23:Textiles, 24:Footwear and Clothing and 38:Transport Equipment, account for more than 50 per cent of wage earners in manufacturing employment. The stability of their share above 50 per cent and its slight increase over time indicate concentration rather than dispersion of employment across the twenty two-digit manufacturing branches.

The common characteristic in these four industrial branches is

that, apart from being leaders in terms of employment shares, the hierarchy among them in terms of ranking in the distribution of employment is stable. Branch 23: Textiles employs the higher percentage of wage earners in manufacturing employment, on average, 19.7 per cent. Branch 20: Food industry employs, on average 12.8 per cent. Branch 24: Footwear and Clothing employs 11.8 per cent. And branch 38: Transport Equipment employs 7.8 per cent of wage earners in manufacturing employment.

Given the lack of statistical information on the occupational composition of employment in the Greek manufacturing sector, the male/female as well as the salary/wage earners composition of employment can be used as a rough indication of the skill mix. While female employment is usually associated with relatively lower skills, the employment of salary earners is associated with relatively higher skills. From this point of view, Table 4.4.7 suggests that in three of the four above mentioned branches, namely in 23: Textiles, 20: Food and 24: Footwear and Clothing, the labour force is composed by relatively high percentages of female wage earners. Only in branch 38: Transport Equipment the labour force is strongly dominated by male wage earners. Furthermore, as Table 4.4.6 suggests, the labour force in the branch 38 is also characterised by above-average employment of salary earners, while in the three other branches of high shares in manufacturing employment, the labour force is characterised by below-average employment of salary earners.

To summarise, the analysis on the evolution of manufacturing

employment at the aggregate and the disaggregated level suggests that, in spite of the overall expansion of manufacturing employment from 1965 to 1988, only the sub-period 1967-74 was one of very strong demand for labour in the manufacturing sector. In 1967-74 along with the rapid expansion in manufacturing employment a new generation of employees who had no labour tradition regarding collective bargaining and trade-unions was brought in largely new manufacturing establishments. Furthermore, the analysis of distribution of manufacturing employment across industrial branches suggests that manufacturing employment has been increasingly concentrated in a small number of industrial branches, actually four out of twenty. It is noteworthy that, as the information provided in Appendix 2 indicates, these observations, which refer to employment of wage earners, are also valid for employment of both salary and wage earners in manufacturing.

4.5. CONCLUSIONS.

In addition to the extensive procedural regulation of collective bargaining, analysed in Chapters 2 and 3, extensive regulation is also observed as far as substantial aspects, namely wages and employment, are concerned. Formal incomes policies were introduced after 1975. In 1975-81 norm-based incomes policies were in use. In 1982-88 indexation policies were adopted. In this context the accelerated growth of real wages observed in 1975-81 slowed down in 1982-88 and became negative during the 1985-87 stabilisation programme.

While the distinction between high-wage and low-wage industrial branches remained very stable throughout the period 1965-88, the wage structure became, compared to 1967-74, wider in 1975-81, to narrow again in 1982-88. The variability in the manufacturing wage structure did not undermine the ranking of industries in the wage structure. However, major disturbances in the manufacturing wage structure are observed in 1975-81, i.e. the period of normbased incomes policy during which strike activity soared and informal collective bargaining procedures at the company and plant level were set up. The observed inter-branch and over time variability in relative wages indicates the existence of some flexibility which must be examined in conjunction with employment developments.

Employment adjustment has been under strict regulation too. The

Ministry of Labour maintained all over the period the right to regulate employment adjustments by setting limits on collective dismissals. In the context of this regulation, over the period 1965-88, contrary to what happened in other industrialised market economies, in Greece manufacturing employment was expanding. The period 1967-74 was the main period of rapid expansion in manufacturing employment, expansion largely associated with the creation of new establishments. In later periods the expansion of manufacturing employment was slower. Throughout the period 1965-88 manufacturing employment was rather concentrated to low-wage industries where relatively higher proportions of female labour are employed.

÷ .

Overall, in the context of extensive regulation of the labour market via incomes policy and legislation on collective dismissals, the observed variability in the wage structure along with the employment growth and restructuring in the Greek manufacturing sector set a scene which prima facie is dominated by rigidities. However a certain variability in wages and is observed. In other words it seems that employment contradictory forces conducive to both rigidities and operate in the labour market of the Greek flexibilities manufacturing.

NOMINAL, REAL, RELATIVE WAGES, PRODUCTIVITY, UNIT LABOUR COSTS, MINIMUM WAGE AS PERCENTAGE OF AVERAGE AND FEMALE AS PERCENTAGE OF MALE WAGE IN GREEK MANUFACTURING INDUSTRIES Industrial and Handicraft Establishments (>10) 1965-1988

	Nominal	Real	Relative	Producti	Unit	Minimum	Female
Year	W	lages		vity	Labour	Costs W	age
		··					
	(a)	(D)	(C) *	(a) chai	(e)	(I) e	(g)
1065	Chang		1000	2 02		67 00	
1965	8.26	5.16	18.06	2.03	3.21	67.80	61.86
1966	12.96	7.96	16.21	10.48	0.82	70.49	62.41
1967	9.47	7.77	12.77	4.09	9.97	74.74	67.59
1968	6.46	6.16	12.32	9.43	-0.19	74.35	68.83
1969	8.13	5.73	12.73	9.07	2.88	74.78	69.23
1970	6.49	3.29	14.12	-0.18	0.68	72.28	68.16
1971	8.85	5.85	13.83	3.04	3.81	66.69	65.99
1972	8.84	4.54	13.77	9.85	-0.55	65.71	67.29
1973	17.67	2.17	14.88	7.62	2.45	59.06	64.96
1974	24.73	-2.17	14.25	-11.35	32.17	71.62	66.98
1975	21.35	7.95	14.49	3.29	18.70	69.66	69.47
1976	31.29	17.99	16.63	5.21	21.32	63.37	70.30
1977	22.34	10.24	17.94	-1.06	22.15	59.76	68.72
1978	23.56	10.96	18.95	6.97	19.73	58.90	69.11
1979	21.85	2.85	17.52	5.88	16.22	55.59	67.99
1980	26.13	1.23	18.67	2.34	25.50	50.31	67.79
1981	26.54	2.04	18.56	-0.05	26.30	49.83	67.18
1982	30.37	9.37	15.58	-3.03	37.78	57.86	73.12
1983	20.49	0.29	14.21	-8.94	18.68	58.18	74.65
1984	25.70	7.20	14.49	2.34	22.24	56.13	76.18
1985	18.56	-0.74	13.59	-1.75	17.50	55.73	77.62
1986	12.64	-10.36	14.63	-0.79	12.50	53.24	76.89
1987	9,93	-6.47	14.04	-0.70	10.75	53.39	77.57
1988	15 29	1 79	13.94	4.09	20.50	56.82	77.71
Average	1 <i>2.23</i>				20.00		
1967-74	-	4.17	13,59	3,95	5.68	69.90	67.38
1075-01	1 24 72	7 61	17 54	2.22	21 /1	58 20	68 65
1982-99	2 1 9 0 0	0 15	14 36	-1 26	19 99	55.91	76.25
1902-00		0.10	14.30	1.20		JJ.JL	

Notes:

- a. Hourly wage for male workers.
- b. Change in nominal hourly wage minus change in consumer prices index.
- c. Coefficient of variation in hourly wages for male workers across twenty two-digit industrial branches.
- f. National minimum wage as percentage of average daily earnings.
- g. Female hourly earnings as percentage of male earnings.

Sources:

- a. NSSG, Monthly Statistical Bulletin, Table 11, Various Issues.
- f. Minimum Wages determined by the National General Collective Agreements.

		1965-198	38	
Year	Inflation ^a	Registered Unemployment	Manufacturin ^D Employment ^C	g Unemployment ^d
	Year to	year Change	(8)	Level
1965	3.1	- 1.50	3.56	4.82
1966	5.0	1.52	3.05	5.01
1967	1.7	29.20	-0.67	5.40
1968	0.3	-11.92	-0.67	5.57
1969	2.4	- 9.84	3.77	5.25
1970	3.2	-26.80	4.66	4.16
1971	3.0	-37.68	6.14	3.14
1972	4.3	-21.37	3.78	2.11
1973	15.5	-10.05	6.22	1.97
1974	26.9	26.28	1.07	2.05
1975	13.4	29.19	0.83	2.29
1976	13.3	-27.04	6.12	1.91
1977	12.1	8.38	4.64	1.69
1978	12.6	11.71	3.09	1.83
1979	19.0	2.10	2.74	1.89
1980	24.9	17.86	1.14	2.75
1981	24.5	14.29	1.13	4.05
1982	21.0	19.05	0.12	5.83
1983	20.2	21.77	-1.18	7.86
1984	18.5	15.56	0.38	8.14
1985	19.3	24.93	-1.19	7.81
1986	23.0	19.47	0.06	7.38
1987	16.4	6.74	-1.26	7.40
1988	13.5	-4.51	1.02	7.74
Average				
1967-74	7.2	-7.77	3.04	3.71
1975-81	14.7	8.07	2.81	2.34
1982-88	18.8	14.72	-0.03	7.45

INFLATION AND LABOUR MARKET CONDITIONS IN GREECE 1965-1988

Table 4.3.2

Notes:

- a. January to January percent change in consumer prices index.
- b. Change in the annual average index for registered unemployment. c. Change in the annual average employment of salary and wage
- earners in manufacturing.
- d. Number of unemployed as percentage of the total labour force.

Sources:

a. OECD, Historical Statistics, 1960-1988, Table 8.11, page 83.

- b. NSSG, Monthly Statistical Bulletin, Table 9, page 13, various issues.
- c. NSSG, Monthly Statistical Bulletin, Table 8, page 13, various issues.
- d. OECD, Historical Statistics, 1960-1986, Table 2.15, page 39, and OECD, Economic Outlook, November 1989, Table D4, page 67.

KEYS FOR INDUSTRIAL BRANCHES IN MANUFACTURING SECTOR

Branch Code	Manufacturing Industry Description
20	Food Manufacturing
21	Beverage Industries
22	Tobacco manufactures
23	Manufacture of Textiles
24	Footwear and Clothing
25	Wood and Cork
26	Furniture
27	Manufacture of Paper and Paper Products
28	Printing and Publishing
29	Manufacture of Leather
30	Manufacture of Rubber and Plastic
31	Manufacture of Chemicals and Chemical Products
32	Manufacture of Products of Petroleum and Coal
33	Manufacture of non-metallic Mineral Products
34	Basic Metal Industries
35	Manufacture of Metal Products
36	Machinery and Appliances (non-electrical)
37	Manufacture of Electrical Machinery, Apparatus,
	Appliances and Supplies
38	Manufacture of Transport Equipment
39	Miscellaneous Manufacturing Industries

Note: Classification of branches in the Greek Standard Industrial Classification, according to which data and indices are compiled and published by the National Statistical Service in Greece (NSSG) is based upon and corresponds to the "International Standard Industrial Classification of All Economic Activities" (ISIC-United Nations, Statistical Papers, Ser. M, No 4, Rev. 2).

	RANKING OI	<u>INDI</u>	<u>JSTRI</u> (de	AL BRZ escend 196	ANCI ding 55-1	H <u>ES</u> <u>I1</u> g orde 1988	<u>N THE</u> er)	WAGE	<u>sti</u>	RUCTURE	2
	Coefficient of Concordance ^a = 0.82										
In	dustrial Branch ^b										
Year	20	21	22	23	24	25	26	27	28	29	
1965 1966 1967 1968 1969 1970 1971 1972 1973 1974 1975 1976 1977 1978 1979	10.5 14.5 12.5 11.0 10.0 13.0 15.0 14.0 16.0 14.0 13.0 15.0 15.0 15.0 15.0 15.0 15.0	9.0 8.5 14.0 17.0 17.0 17.0 17.5 10.5 12.5 17.0 17.0 14.0 14.0 14.0 15.0	7.5 4.5 8.0 14.0 15.0 15.0 17.5 10.5 16.0 11.5 10.0 9.0 8.0 13.0 13.0	$ \begin{array}{c} 10.5 \\ 10.0 \\ 6.0 \\ 5.5 \\ 7.5 \\ 5.0 \\ 7.0 \\ 9.0 \\ 7.0 \\ 7.0 \\ 7.5 \\ 11.0 \\ 10.0 \\ 10.0 \\ 10.0 \\ \end{array} $	14 16 11 14 10 10 4 6 18 17 16 9 17 16 17 14	20.0 20.0 19.0 19.5 20.0 20.0 19.0 19.0 19.0 15.0 16.0 18.0 18.0 16.0	19.0 19.0 20.0 19.5 19.0 16.0 13.0 15.0 19.0 20.0 19.0 20.0 19.0 20.0 19.0	$ \begin{array}{r} 16.5\\ 14.5\\ 17.0\\ 12.0\\ 13.0\\ 11.0\\ 12.0\\ 9.0\\ 17.0\\ 9.5\\ 9.0\\ 7.5\\ 7.5\\ 10.0\\ 8.0\\ 8.0\\ \end{array} $	$1 \\ 1 \\ 2 \\ 1 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ $	12.0 12.0 9.0 8.0 7.5 11.0 12.0 14.0 15.0 11.5 13.0 12.0 12.0 12.0	
1981 1982 1983 1984 1985 1986 1987 1988	17.0 15.0 14.0 14.0 16.0 15.0 15.0 16.0	12.0 12.0 11.0 9.0 10.0 10.0 12.0 11.0	11.0 8.5 7.0 6.0 9.0 7.0 7.0 8.0	10.0 8.5 9.0 11.0 7.0 8.0 8.0 7.0	15 16 17 18 18 19 17 17	18.0 14.0 16.0 15.0 19.0 18.0 19.0 18.0	19.0 19.0 20.0 20.0 20.0 20.0 20.0 20.0	9.0 10.0 8.0 8.0 9.0 11.0 12.0	2 4 6 7 5 6 6 6	14.0 17.0 18.0 17.0 13.5 14.0 14.0 15.0	
											•

Notes:

a. Coefficient of Concordance = Variance of rank sums / maximum possible variance of rank sums. Called also Kendall's W statistic. By definition cannot be negative and its maximum value is 1. For further information see Hays, W. L. 1973: <u>Statistics for Social Sciences</u>, Holt, Rinehart and Winston Inc., New York, pp. 801-803.

b. For keys see Table 4.3.3.

Table 4.3.4 (continued)

RANKING OF INDUSTRIAL BRANCHES IN THE WAGE STRUCTURE (descending order) 1965-1988

Industrial Branch^p 30 31 32 33 34 35 36 37 38 39 Year 15.0 4.0 3 5.5 2 13.0 18.0 16.5 5.5 7.5 1965 1966 11.0 6.5 3 4.5 2 13.0 18.0 17.0 6.5 8.5 9.0 4.0 3 5.0 1 15.5 18.0 15.5 12.5 7.0 1967

 7.0
 4.0
 3
 5.0
 2
 14.0
 18.0
 16.0
 14.0
 10.0

 7.0
 5.5
 3
 4.0
 1
 10.0
 18.0
 16.0
 12.0
 15.0

 5.0
 6.0
 3
 4.0
 1
 12.0
 19.0
 14.0
 9.0
 18.0

 1968 1969 1970 1971 7.0 9.0 3 8.0 1 10.0 18.0 15.0 6.0 20.0 1972 4.0 5.0 3 8.0 1 11.0 16.0 13.0 10.0 20.0 1973 7.5 5.0 4 7.5 1 6.0 12.0 13.0 3.0 15.0 1974 6.0 5.0 4 9.5 1 8.0 11.0 12.5 3.0 18.0 10.0 5.0 3 6.0 1 8.0 20.0 14.0 4.0 18.0 1975

 11.0
 4.5
 3
 6.0
 1
 12.0
 18.0
 14.0
 4.5
 19.0

 7.5
 5.0
 3
 6.0
 1
 10.0
 18.0
 13.0
 4.0
 19.0

 9.0
 5.0
 3
 6.0
 1
 7.0
 17.0
 12.0
 4.0
 20.0

 1976 1977 1978 9.0 5.0 3 6.0 1 7.0 16.0 11.0 4.0 19.0 1979 1980 9.0 5.0 3 6.0 1 7.0 17.0 11.0 4.0 20.0 1981 8.0 5.0 3 6.0 1 7.0 16.0 13.0 4.0 20.0 1982 7.0 2.0 3 6.0 1 11.0 18.0 13.0 5.0 20.0 10.0 4.0 2 5.0 1 12.0 15.0 13.0 3.0 20.0 1983

 10.0
 3.0
 2
 5.0
 1
 12.0
 15.0
 13.0
 3.0
 20.0

 10.0
 3.0
 2
 5.0
 1
 12.0
 16.0
 13.0
 4.0
 19.0

 12.0
 4.0
 2
 6.0
 1
 11.0
 13.5
 15.0
 3.0
 17.0

 12.0
 5.0
 2
 4.0
 1
 11.0
 13.0
 16.0
 3.0
 17.0

 10.0
 5.0
 2
 4.0
 1
 9.0
 13.0
 16.0
 3.0
 18.0

 1984 1985 1986 1987 1988 9.0 4.0 2 5.0 1 10.0 13.0 14.0 3.0 19.0 ~~~~~~~~~~~~~~

Note: b. For keys see Table 4.3.3.

DETERMINANTS OF THE WAGE STRUCTURE IN GREEK MANUFACTURING 1976, 1981

Dependent Number of	variable: observations	Manufacturin = 20 (two-d	g Wages. igit industr	ial branches).
Y	ear 1976		1981	
Constant	41.21 (11.72)	[0.00]	125.05 (8.61)	[0.00]
Productivi	ty 0.01 (0.73)	[0.20]	-0.03 (-1.10)	[-0.34]
Size	0.11 ^{**} (2.66)	* [0.57]	0.50 ^{***} (3.40)	** [0.73]
Skill	-0.002 (-0.02)	[-0.003]	0.76 [*] (1.86)	[0.45]
Female	-0.16 ^{**} (-2.96)	**[-0.36]	-0.55 ^{***} (-2.74)	* [-0.39]
R ² R ² -adjuste F-value	.83 d .79 18.42		.77 .71 12.68	

Notes:

Wages = Hourly wage of workers of both sexes in branch i. Productivity = Value added per employee in branch i. Size = Number of employees per establishment in branch i. Skill = Because of data unavailability we use as a proxy the share of salary earners in employment of salary and wage earners in branch i. Female = Share of female wage earners in wage earners of both sexes in branch i.

Levels of significance: *=0.10, ***=0.025, ****=0.01, ****=0.005. t-statistics in parentheses. Standardised estimates in brackets.

			1965	-1988		
Index	Coefficie	ent of Va	ariation ^a	R-s	quare ^b	
E Year	oth Sexes	Male	Female	Both Sexes	Male	Female
1965	21.29	18.06	11.47	1.00	1.00	1.00
1966	19.35	16.23	7.10	•97	.95	.58
1967	15.47	12.77	6.48	.89	.86	.56
1968	14.43	12.34	5.75	.86	.82	.32
1969	14.90	12.74	4.68	.82	.78	07
1970	16.99	14.08	5.39	.81	.74	.15
1971	16.72	13.86	6.44	.78	.70	.21
1972	16.38	13.77	8.13	.74	.67	.25
1973	18.47	14.88	8.42	.85	.85	19
1974	17.04	14.25	5.67	.78	.71	14
1975	17.13	14.49	5.97	.83	.81	.35
1976	19.60	16.63	7.88	.79	.75	.17
1977	20.80	17.95	8.51	.84	.79	.19
1978	21.36	18.95	8.72	.86	.81	.03
1979	20.49	17.52	8.38	.87	.84	.19
1980	21.14	18.67	10.23	.84	.81	04
1981	20.58	18.56	12.41	.82	.79	49
1982	17.23	15.59	9.13	.71	.62	11
1983	16.32	14.21	8.84	.68	•58	02
1984	16.47	14.49	8.47	.67	.57	.29
1985	15.40	13.59	7.50	.71	.66	.48
1986	16.72	14.63	8.40	.70	.61	.51
1987	15.82	14.04	8.02	.68	.60	.51
1988	15.68	13.94	7.76	.67	.60	.36
Averag	e					
1965-6	6 20.32	17.15	9.29			
1967-7	4 16.30	13.59	6.37			
1975-8	1 20.16	17.54	8.87			
1982-8	8 16.23	14.36	8.30			

INTER-BRANCH VARIABILITY IN THE GREEK MANUFACTURING WAGE STRUCTURE

Notes: a. Coefficient of Variation= Standard Deviation/Mean, n=20 (two-digit industrial branches).

b. R-square is used as indicator of stability in the wage structure by correlating relative wages of industries among pairs of years, one of which is the base year. In this case 1965 is used as the base year.

Selected Periods								
	lent of var		er100					
1900-88	1968-74	19/5-81	1982-88					
3.2	3.2	2.0	1.1					
4.9	4.1	3.2	2.7					
8.0	2.6	3.5	1.6					
4.0	1.4	0.8	1.9					
7.6	7.1	3.9	2.1					
3.2	3.8	1.5	1.7					
6.0	4.2	1.9	2.6					
4.4	3.7	1.5	2.4					
10.5	2.4	5.5	2.6					
4.6	3.6	2.0	2.4					
3.7	2.6	1.9	1.5					
3.6	3.4	2.1	2.7					
5.7	3.1	4.6	3.9					
3.6	4.3	1.6	2.0					
4.7	3.4	3.6	3.0					
1.6	1.3	1.8	1.1					
3.3	4.0	2.4	1.9					
2.4	2.5	2.8	2.0					
6.0	7.9	2.1	2.4					
7.9	5.9	4.0	4.7					
	Coeffic 1966-88 3.2 4.9 8.0 4.0 7.6 3.2 6.0 4.4 10.5 4.6 3.7 3.6 5.7 3.6 4.7 1.6 3.3 2.4 6.0 7.9	Select Coefficient of Var 1966-88 1968-74 3.2 3.2 4.9 4.1 8.0 2.6 4.0 1.4 7.6 7.1 3.2 3.8 6.0 4.2 4.4 3.7 10.5 2.4 4.6 3.6 3.7 2.6 3.6 3.4 5.7 3.1 3.6 4.3 4.7 3.4 1.6 1.3 3.3 4.0 2.4 2.5 6.0 7.9 7.9 5.9	Selected PeriodsCoefficient of Variation in P1966-881968-741975-81 3.2 3.2 2.0 4.9 4.1 3.2 8.0 2.6 3.5 4.0 1.4 0.8 7.6 7.1 3.9 3.2 3.8 1.5 6.0 4.2 1.9 4.4 3.7 1.5 10.5 2.4 5.5 4.6 3.6 2.0 3.7 2.6 1.9 3.6 3.4 2.1 5.7 3.1 4.6 3.6 4.3 1.6 4.7 3.4 3.6 1.6 1.3 1.8 3.3 4.0 2.4 2.4 2.5 2.8 6.0 7.9 2.1 7.9 5.9 4.0	Selected PeriodsSelected PeriodsCoefficient of Variation in Period1966-881968-741975-811982-883.23.22.01.14.94.13.22.78.02.63.51.64.01.40.81.97.67.13.92.13.23.81.51.76.04.21.92.64.43.71.52.410.52.45.52.64.63.62.02.43.72.61.91.53.63.42.12.75.73.14.63.93.64.31.62.04.73.43.63.01.61.31.81.13.34.02.41.92.42.52.82.06.07.92.12.47.95.94.04.7				

OVER TIME VARIABILITY OF BRANCH RELATIVE WAGES IN GREEK MANUFACTURING

Note:

Wages refer to hourly wage of workers of both sexes. Relative wage in branch i is compiled by dividing the wage in branch i by the average wage of all branches. Variability is calculated by estimating the coefficient of variation (standard deviation / mean) in each branch relative wage over the selected periods. The number of observations in each period is 1966-88:23, 1968-74:7, 1975-81:7, 1982-88:7. For keys see Table 4.3.3.

CASES OF COLLECTIVE DISMISSALS UNDER MINISTER OF LABOUR JURISDICTION 1983-1986 TotalNumber ofNumberEmployeesofPlanned toPeriodEmployeesBeDismissedDecision 1983^a Firm 1 Firm 2 Firm 3 Firm 4 23096300096802910049 230 Authorisation Denied Agreement after Consultation Authorisation Denied Agreement after Consultation Authorisation Denied 3000 1984 15312183496773351002287514075 Authorisation Granted Authorisation Denied Authorisation Denied Authorisation Denied Authorisation Denied Authorisation Denied Firm 5 Firm 6 Firm 7 Firm 8 Firm 9 Firm 10 1985 No Cases --1986^b .. 10 61 31 297 297 142 39 Firm 11 Agreement after Consultation Authorisation Granted Authorisation Granted Authorisation Denied Authorisation Denied Firm 12 Firm 13 39 109 225 Firm 14 142 Firm 15 289 Firm 16906225Authorisation GrantedFirm 172350200Authorisation Granted _____

Table 4.4.1

Notes: a. September-December.

b. January-October.

Source: Sabethai (1986:40).

TANATA JIJIN	Table	e	٤.	4.	•	2
---------------------	-------	---	----	----	---	---

<u>CASES</u>	<u>of</u>	COLLECTI	VE <u>DISMISSALS</u> 1983-19	UNDER 986	PREFECTS!	JURISDICTI	<u>0N</u>
Period		Total Number of Employees	Number of Employees Planned to Be Dismissed	 M	inisterial Decision		
1983 ^a Firm 1 Firm 2	· 	••	28 ••	Autho Autho	risation 1 risation 1	Denied Denied	-
1984 Firm 3 Firm 4 Firm 5 Firm 6 Firm 7		179 185 140 	179 51 26 5% 30	Autho Autho Autho Autho Autho	risation risation risation G risation risation	Denied Denied ranted Denied Denied	
1985 Firm 8	•			Autho	risation 1	Denied	
1986~ Firm 9 Firm 10 Firm 11 Firm 12) - -	 40 150	 15 75	Autho Autho Autho Autho	risation risation risation risation	Denied Denied ranted Denied	_

Notes: a. September-December. b. January-October.

Source: Sabethai (1986:40).

RATE OF CHAI	NGE IN MAI	Selected	<u>NG</u> <u>EMPLOY</u> Periods	<u>Ment in O</u>	ECD COUNTRIES
Period	1960-68	1968-73	1973-79	1979-87	1960-87
Country					
U.S.	2.3	0.2	1.1	-0.9	0.7
Japan	4.1	2.0	-1.3	0.8	1.5
Germany	0.3	0.9	-1.4	-1.0	-0.3
France	0.5	2.1	-0.9	-2.3	-0.3
U.K.	-0.3	-1.0	-1.3	-4.2	-1.8
Italy	0.8	0.4	0.2	-1.8	-0.2
Canada	2.8	2.1	1.5	-0.1	1.5
Austria	-0.7	1.8	-0.7	-1.5	-0.5
Belgium	0.6	0.2	-3.3	-2.5	1.3
Denmark	0.1	1.5	-1.5	0.5	0.1
Finland	1.0	4.8	-0.1	-1.0	0.8
Greece	1.9	5.0	1.6	1.4	2.2
Iceland	-0.5	5.1	3.3	1.4	1.9
Ireland	1.6	1.8	1.2	-2.0	0.5
Luxemburg	0.7		-2.2	-1.5	
Netherlands	0.3	-0.9	-1.9	-0.7	-0.7
Norway	1.0		-0.2	-1.1	
Portugal	0.6	0.9			
Spain	2.5			-2.0	
Sweden	0.1	-1.3	-0.6		
Switzerland	1.4	0.1	-2.6	-0.4	-0.3
Turkey	3.6	3.4	1.7	3.1	3.0
Australia	1.1	1.6	-1.7	-0.6	0.0
New Zealand	2.7	0.8	0.7	_	
Total EEC	0.5	0.6	-0.9	-1.9	-0.5
Total OECD	1.6	0.8	-0.4	-0.9	0.3

Note: Average annual percentage rates of change between periods indicated.

Source: OECD 1988, Labour Force Statistics, Table 1.10, page 28.

Selected Periods								
Period Country	1960-67	1968-73	1974-79	1980-87	1960-87			
U.S. Japan Germany France U.K. Italy	26.8 23.7 35.3 27.6 37.4	25.8 26.9 36.6 27.6 36.1	23.0 25.4 35.1 27.2 32.6	20.2 24.7 32.7 23.9 26.3	23.9 25.0 34.8 26.4 32.9			
Canada Austria Belgium Denmark Finland	25.0 25.0 31.7 39.1 24.6 26.8	27.9 23.8 32.1 36.5 28.5	27.5 21.7 31.2 31.5 28.0	24.5 19.7 28.7 26.0 17.9 25.2	26.3 22.5 30.8 33.2 26.9			
Greece Iceland Ireland Luxemburg Netherlands	13.1 24.0 18.3 33.5 29.4	16.5 22.4 20.3 27.0	19.1 23.0 21.2 30.4 23.8	19.4 23.0 19.9 24.8 20.0	16.9 23.2 19.8			
Norway Portugal Spain Sweden Switzerland Turkey Australia New Zealand Total EEC	26.1 23.3 25.1 31.4 38.5 8.3 29.2 27.4 29.7	24.1 26.8 28.3 36.5 10.1 26.4 26.9 30.1	22.6 25.1 26.2 26.4 33.0 10.9 23.1 24.9 28.7	18.3 25.5 23.8 22.8 30.3 11.5 19.9 25.2	25.3 27.2 34.6 10.2 24.6 28.3			
Total OECD	26.9	27.2	25.3	22.6	25.4			

<u>SHARE OF MANUFACTURING EMPLOYMENT IN OECD COUNTRIES</u> Selected Periods

Note: Employment in manufacturing as percentage of civilian employment.

Source: OECD 1988, Labour Force Statistics, Table 2.11, page 37.

			1965-1988	3		
Year	Manufactu in Total : All ^a Em (share	ring Share Employment ployees ^b %)	Agg Manufa Emplo SWE ^C (annual	regate acturing oyment WEd change %)	Branch Expand Employm SWE (out of	ies ling bent WE 20)
1965 1966 1967 1968 1969 1970 1971 1972 1973 1974 1975 1976 1977 1978 1977 1978 1979 1980 1981 1982 1983 1984 1985 1986 1987 1988 Average 1967-74	17.14 ^e 26.68 27.59 27.66 27.28 27.11 26.43 19.29 19.26 19.19 19.11 18.91 19.93 19.90 e 4 - 1 26.01	27.35 ^e 29.03 29.83 29.84 29.35 29.30 29.27 28.52 27.56 27.47 27.21 26.59 28.20 27.44	5.93 3.77 -1.45 -1.96 2.10 11.06 6.56 5.21 7.64 10.50 1.19 5.09 2.58 0.57 0.22 -1.27 -1.23 -2.18 9.27 -0.07 3.10 0.10 -1.30 1.03 6.03 0.85 0.77 0.22	5.18 2.26 -2.88 -3.82 1.51 10.76 6.42 4.70 6.41 10.44 0.64 0.50 1.84 -0.56 -0.41 -2.54 -2.84 -3.90 12.48 -0.51 0.07 -0.62 -2.77 -0.68 5.20 -0.57 -0.57	19 17 12 10 12 19 17 18 19 20 9 14 14 14 8 12 6 4 4 4 15 9 13 7 7 9 13 7 7 9	18 16 8 9 11 17 15 17 17 18 8 11 11 5 9 5 2 1 16 6 11 7 6 7 14 7 2
<pre>1982-88 19.38 27.41 1.77 1.07 9 8 Notes: a. All types of employment included, i.e. employers, employees, self-employed and unpaid family workers. b. Only salary and wage earners included. c. Salary and Wage Earners. d. Wage Earners. Sources:a. NSSG, Annual Labour Force Survey, Table 7 (for 1975-83 1985-88) and Table 8 (for 1984). b. As for a. c. Table A4.4.11. d. Table 4.4.11. e. Census 1971, Vol. 3, Table 8, page 225.</pre>						

SHARE OF AND CHANGE IN MANUFACTURING EMPLOYMENT IN GREECE

201

Έ.,

			1965	-1988			
Ye	ar 1965	1970	1975	1980	1985	1988	
Branc	:h		share p	er cent			
20	23.78	24.88	23.99	26.21	31.25	32.85	
22	7.67	13.01	13.49	18.53	21.94	24.67	
23 24	10.63 8.61	13.50 9.74	14.60 8.47	20.13 8.00	18.49 11.77	20.98 12.33	
25 26	8.16 12.08	11.19 10.63	15.90 11.47	19.40 18.52	30.40 17.84	33.65 17.53	
27 28	15.16 36.92	25.78 33.55	27.70 36.36	32.97 39.98	38.04 54.56	50.42 54.69	
29 30	6.93 17.45	9.51 20.97	9.01 22.55	14.82 20.99	16.77	16.65	
31 32	32.71	44.76	50.10	60.47	56.76	59.15	
33	14.98	23.09	29.97	32.57	39.34	42.38	
35	12.21	16.39	20.16	26.88	27.94	43.56	
36 37	11.70 23.87	15.87 27.46	20.78 32.03	27.21 36.80	33.05 45.98	37.99 47.78	
38 39	22.20 12.19	23.08 13.68	26.24 14.16	45.63 19.23	45.38 28.68	48.37 30.66	
AII 	1/.13 	21.71	23.55	29.61	32.32	34.93	

SHARE OF SALARY EARNERS IN MANUFACTURING EMPLOYMENT BY BRANCH

Note: For keys see Table 4.3.3.

Sources: 1965-1975, NSSG, Annual Industrial Survey, Table 6. 1976-1988, NSSG, Labour Force Statistics, Table 3.

<u>SHARE</u>	<u>of</u>	FEMALE	EMPLOY	<u>ees</u>	<u>IN</u> 197	<u>MANUF7</u> 6-1988	CT	URING	EMPLO	YMENT	<u>by</u>	BRANCH
Ye	ear		1976		1980	1	.98	4	1988			-
Branch				sł	nare	per d	en	t				
20			3.65	44	1.59	43	.04	 4	42.47			-
21		2	25.31	25	5.23	23	1:	3	23.71			
22		5	57.30	55	5.76	56	5.8	1	50.39			
23		5	58.66	57	7.18	55	5.4	1	53.14			
24		7	76.57	83	L.40	74	.9	3	76.01			
25		2	22.36	23	3.83	28	.8	0	19.76			
26			8.65	10	0.83	13	. 8	9	11.11			
27		2	27.96	26	5.68	23	.2	6	23.75			
28		2	27.97	27	7.43	34		6	31.23			
29		3	30.70	34	1.80	39	.6	6	39.48			
30		3	38.75	40).99	31	5	1	26.89			
31		3	35.41	36	5.12	33	.8	6	33.93			
32		1	L0.82	12	2.53	10	.34	4	9.75			
33		1	17.16	14	1.06	12	. 4	0	16.50			
34			4.21	4	1.97	e	5.5	0	5.36			
35		2	21.22	22	2.30	17	.3	9	18.47			
36			7.10	-	7.03	7	. 82	2	9.10			
37		2	25.22	30	.48	26	5.3	4	22.62			
38			6.11	(5.25	. 7	.28	8	6.47			
39		5	51.39	50	.56	39	. 68	8	38.75			
A11		3	85.26	36	5.48	36	.8	3	35.62			
Note:	Shar sala	e of f ary and	female wage e	emp	loy	ees i	n 1	total	emplo	yment	of	both

Source: NSSG, Labour Force Statistics, Table 3.

203

..

• .

Period	1967/74	1975/81	1982/88	1965/88	
Branch		change	per cent		
20	38.12	-14.62	42.63	75.52	
21	47.90	-16.33	14.48	69.41	
22	-31.21	8.36	-15.63	-54.65	
23	26.08	-5.18	6.93	25.54	
24	131.44	9.54	29.91	282.65	
25	52.57	-16.66	-9.04	1.54	
26	59.78	-19.70	4.19	19.33	
27	-0.86	-8.64	11.35	-1.24	
28	18.93	-16.37	-11.13	-19.48	
29	2.34	-12.82	33.50	-0.20	
30	39.61	24.31	-19.41	76.55	
31	28.99	-13.31	31.76	23.12	
32	102.25	20.46	-13.02	82.03	
33	11.20	-14.01	21.24	1.86	
34	83.58	-22.41	23.57	113.31	
35	45.50	1.23	-7.85	20.33	
36	33.52	-12.83	-28.46	-21.31	
37	79.05	-16.96	-4.44	28.26	
38	122.04	24.69	-28.38	70.79	
39	80.83	-34.01	35.73	70.01	
All	41.58	-4.02	7.47	40.28	
Number of Branches with Expanding					
Employment	18	6	11	15	

<u>CHANGES IN EMPLOYMENT OF WAGE EARNERS IN MANUFACTURING BY BRANCH</u> Selected Periods

Sources: 1967-1975, NSSG, Annual Industrial Survey, Table 6. 1976-1988, NSSG, Labour Force Statistics, Table 3.

			1902-1900	5		
Year	1965	1970	1975	1980	1985	1988
Branch			per cent	t		
20	12.04	13.73	12.94	11.56	13.76	15.07
21	1.52	2.05	2.03	1.91	2.12	1.83
22	7.48	4.38	2.71	3.08	2.50	2.42
23	21.85	19.31	19.84	20.17	19.43	19.55
24	5.92	7.28	11.26	12.52	15.58	16.14
25	2.77	3.10	2.83	2.59	2.15	2.00
26	2.17	2.47	2.37	1.97	1.82	1.84
27	2.83	2.50	2.06	1.99	2.25	1.99
28	2.55	2.69	2.19	2.16	1.49	1.46
29	2.03	1.65	1.29	1.20	1.64	1.45
30	2.99	3.30	3.69	4.65	4.01	3.76
31	4.60	4.04	3.67	3.24	4.12	4.03
32	0.31	0.38	0.38	0.54	0.36	0.40
33	8.08	7.28	5.97	5.60	6.11	5.87
34	1.50	2.31	2.56	2.24	2.63	2.28
35	7.06	6.91	6.83	6.79	6.21	6.06
36	3.58	3.67	3.27	2.99	2.27	2.01
37	4.06	5.15	4.69	4.19	3.83	3.71
38	5.85	6.67	8.22	9.60	6.80	7.12
39	0.82	1.14	1.20	0.98	0.91	1.00
A11	100	100	100	100	100	100
(20 22						
24, 38)	45.66	46.99	52.26	53.85	55.57	57.88
Total	181430	194846	256752	253616	265184	254502

DISTRIBUTION OF WAGE EARNERS ACROSS MANUFACTURING INDUSTRIES 1965-1988

Sources: 1965-1975, NSSG, Annual Industrial Survey, Table 6. 1976-1988, NSSG, Labour Force Statistics, Table 3.

	Average 1967-74	1	Average 1975-81		Aver age 1982-88		Average 1966-88	
Bran	nch	Share %		Share %		Share %		Share %
20	26603	13.02	 29953	11.65	34549	13.38	29895	12.67
21	4232	2.07	4965	1.93	5054	1.96	4663	1.98
22	8561	4.19	7840	3.05	6589	2.55	7926	3.36
23	40390	19.76	50331	19.58	50291	19.47	46426	19.68
24	15982	7.82	31277	12.17	40567	15.71	27908	11.83
25	6339	3.10	6534	2.54	5687	2.20	6150	2.61
26	4734	2.32	5801	2.26	4987	1.93	5103	2.16
27	5113	2.50	5255	2.04	5713	2.21	5352	2.27
28	5135	2.51	5726	2.23	4244	1.64	5029	2.13
29	3172	1.55	3217	1.25	3866	1.50	3410	1.45
30	7110	3.48	11271	4.39	10846	4.20	9463	4.01
31	8333	4.08	8487	3.30	10844	4.20	9150	3.88
32	635	0.31	1476	0.57	1031	0.40	1002	0.42
33	14538	7.11	14987	5.83	15156	5.87	14880	6.31
34	4730	2.31	5761	2.24	6057	2.35	5388	2.28
35	14023	6.86	17385	6.76	16470	6.38	15737	6.67
36	7290	3.57	8770	3.41	5880	2.28	7283	3.09
37	10967	5.37	11235	4.37	9708	3.76	10499	4.45
38	14254	6.98	23860	9.28	18424	7.13	18288	7.75
39	2213	1.08	2890	1.12	2269	0.88	2409	1.02
A11	204353	100	257021	100	258232	100	235962	100
(20	23							
24	38)	47.85		55.16		57.58		53.53

AVERAGE EMPLOYMENT OF WAGE EARNERS IN MANUFACTURING INDUSTRIES Selected Periods

Sources: 1967-1975, NSSG, Annual Industrial Survey, Table 6. 1976-1988, NSSG, Labour Force Statistics, Table 3.

Year	1965	1966	1967	1968	1969	1970
Branch					1909	2370
20	1.59	6.47	-0.56	-3,13	3.76	15.06
21	9.63	18.62	12.06	-7.95	-2.88	21 72
22	7.68	-5.66	-15.10	-8.91	-13 84	0 02
23	0.80	1.72	-4.68	-7 22	-2 51	Q 25
24	4 51	3 60	3 94	0 10	-0 /3	22 16
25	15 35	3 37	1 75	-4 28	-0.4J 5.02	12 66
26	1 95	1 07	-0 18	-4.20	-2.95	12.00
20	15 75	5 45	-1 26	-1 90	-3.90	23.37
27	-0 70	2.45	-1.20	-1.60	-3.90	-3.39
20	-0.79	J./6 _5.74	-0.21	-4.51	0.28	13.84
29	-2.12	-5.74	-0.37	-0.17	-2.34	0.94
30	2.92	9.61	2.88	2.55	13.21	-9.43
31	19.46	1.45	-10.26	2.66	3.59	-2.44
32	2.97	-23.78	5.20	6.29	-0.42	59.02
33	14.76	1.78	-3.21	-6.94	-2.04	7.68
34	24.20	23.49	-0.30	0.98	20.60	10.18
35	1.89	-0.15	-6.35	0.44	1.70	10.01
36	4.74	2.46	-0.26	-6.18	7.02	7.31
37	1.32	-3.22	9.84	0.18	10.05	16.19
38	5.27	0.08	-9.27	-5.05	14.73	23.86
39	13.47	6.84	2.51	0.55	0.37	34.34
A11	5.18	2.26	-2.88	-3.82	1.51	10.76
Year	1971	1972	1973	1974	1975	1976
Branch						
20	5.01	0.47	2.89	10.01	3.99	-8.01
21	12.62	5.55	3.84	10.10	-3.77	-5.31
22	-2.74	-10.04	-1.38	1.55	-6.91	14.87
23	6.41	6.03	7.66	6.01	5.15	-0.62
24	9.80	11.01	12.09	38.02	8.04	-4.33
25	5.45	9.59	8.94	6.08	-9.78	-9.93
26	0.10	4.22	2.09	23.77	-4.02	10.21
27	2.40	1.24	-0.42	5.32	-0.32	5.71
28	-0.04	1.94	2.86	4.07	-1.73	12.24
29	-0.65	-13.34	0.54	27.81	-6.31	-1.77
30	6.94	9.53	6.64	6.30	10.92	13.62
31	-0.75	9.29	7.48	6.64	-3.76	-4.63
32	-10.81	3.89	-2.59	33.14	9.78	73.18
33	5.54	-0.79	2.42	5.62	-4.57	3.28
34	7.04	11.08	15.36	-0.26	6.64	-9.94
35	3.78	8.07	5.96	8.95	0 58	-5,78
36	1 66	1 1 2	11 10	8 31	-5 31	12 20
37	10 16	10 16	10 5/	_7 05	-14 21	12.00
38	20.57	7 11	6 71	-2.05	-1 3/	4.20
20	20.57	/•II	11 20	17.4U 6 34	-1.34 A 27	1.0/
-כ ווג	J./8 6 40	4 70	Z 11	0.J4 10 AA	4.0/	0.8J 0.50
	0.42	4./0	U.41 	LV.44	U.04	0.30

ANNUAL CHANGES IN MANUFACTURING EMPLOYMENT OF WAGE EARNERS BY BRANCH 1965-1988

(continued)

Table 4.3.11 (continued)

Ve	ar 1977	1978	1979	1980	1081	1082
Branch	ur 1977	1770	1979	1900	1901	1902
20	-1.08	-0.99	-6.33	4.61	-3 30	-5 22
21	2.45	3.47	-2.44	-4.91	-10.13	-6.52
22	4.22	-2.84	0.01	-3.46	-3.52	-3 29
23	0.85	-1.56	-0.61	2.37	-5.55	-3 66
24	10.46	7.65	8.06	-10.67	-0.25	-0 13
25	1.27	-4.38	-0.24	3,91	-7.82	-7.48
26	-4.48	-6.41	-7.60	-9.55	-2.47	-7.82
27	-1.54	-3,11	-2.41	-2.84	-4.46	-5.66
28	-3.05	-3.52	0.55	-7.78	-14.10	-10.87
29	2.78	2.26	-6.59	-5.31	-4.53	-4.77
30	1.54	6.93	6.96	-5.53	-0.28	0.91
31	-7.96	-1.49	0.36	0.61	-0.72	-4.64
32	4.61	-9.07	1.54	-16.79	-13.45	-2.42
33	-0.31	-0.97	-4.26	-4.99	-7.28	-6.54
34	-3.06	-1.98	1.53	-0.32	-10.41	-7.69
35	5,96	0.53	0.01	-2.10	3.02	-5.74
36	4.62	-0.80	-9.09	-15.11	-3,53	-2.36
37	-8.39	-1.77	-5.31	-0.63	-5.89	-1.02
38	9,18	-5.60	3.02	0.77	8.04	-3.85
39	-2.14	-7.74	-3.95	-14.59	-18,13	-8.44
A11	1.84	-0.56	-0.41	-2.54	-2.84	-3.90
 Ye	ar 1983	1984	 1985	1986	 1987	 1988
Ye Branch	ar 1983	1984	1985	1986	1987	1988
Ye Branch 20	ar 1983 19.89	1984 10.50	1985 2.47	1986 -1.74	1987 1.61	1988 5.24
Ye Branch 20 21	ar 1983 19.89 28.59	1984 10.50 5.67	1985 2.47 1.41	1986 -1.74 -5.76	1987 1.61 -6.58	1988 5.24 -5.63
Ye Branch 20 21 22	ar 1983 19.89 28.59 -4.84	1984 10.50 5.67 -8.22	1985 2.47 1.41 3.92	1986 -1.74 -5.76 9.71	1987 1.61 -6.58 -25.05	1988 5.24 -5.63 13.05
Ye Branch 20 21 22 23	ar 1983 19.89 28.59 -4.84 5.85	1984 10.50 5.67 -8.22 0.28	1985 2.47 1.41 3.92 4.29	1986 -1.74 -5.76 9.71 3.48	1987 1.61 -6.58 -25.05 -2.00	1988 5.24 -5.63 13.05 -4.76
Ye Branch 20 21 22 23 24	ar 1983 19.89 28.59 -4.84 5.85 40.22	1984 10.50 5.67 -8.22 0.28 -0.37	1985 2.47 1.41 3.92 4.29 -6.47	1986 -1.74 -5.76 9.71 3.48 -1.23	1987 1.61 -6.58 -25.05 -2.00 -0.48	1988 5.24 -5.63 13.05 -4.76 1.14
Ye Branch 20 21 22 23 24 25	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28	1985 2.47 1.41 3.92 4.29 -6.47 -3.83	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13
Ye Branch 20 21 22 23 24 25 26	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40
Ye Branch 20 21 22 23 24 25 26 27	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10
Ye Branch 20 21 22 23 24 25 26 27 28	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52
Ye Branch 20 21 22 23 24 25 26 27 28 29	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74
Ye Branch 20 21 22 23 24 25 26 27 28 29 30	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -9.16 -9.16 -9.16 -9.16 -9.16	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32 33	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03 27.44	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51 -0.48	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32 3.73 3.73	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10 -0.80	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90 -5.18	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09 -2.03
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32 33 34	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03 27.44 27.51	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51 -0.48 4.81	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32 3.73 11.09	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10 -0.80 -3.90	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90 -5.18 -12.07	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09 -2.03 -1.50
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32 33 34 35	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03 27.44 27.51 3.21	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51 -0.48 4.81 -1.38	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32 3.73 11.09 -3.28 -3.28 -3.28 -3.28 -3.28	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10 -0.80 -3.90 0.90 0.90	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90 -5.18 -12.07 -4.89	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09 -2.03 -1.50 -2.45
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32 33 34 35 36	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03 27.44 27.51 3.21 -14.66	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51 -0.48 4.81 -1.38 -0.72	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32 3.73 11.09 -3.28 -0.26 -0.26 -0.26 -0.26 -0.26	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10 -0.80 -3.90 0.90 -7.61	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90 -5.18 -12.07 -4.89 -7.36	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09 -2.03 -1.50 -2.45 -1.08
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32 33 34 35 36 37	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03 27.44 27.51 3.21 -14.66 3.65	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51 -0.48 4.81 -1.38 -0.72 -11.76 -9.15 -0.48 -1.38 -0.72 -11.76 -0.48 -1.38 -0.72 -11.76 -0.48 -1.38 -0.72 -11.76 -0.48 -1.38 -0.72 -11.76 -0.48 -0.72 -11.76 -0.48 -0.72 -0.48 -0.72 -0.48 -0.72 -0.48 -0.72 -0.48 -0.72 -0.48 -0.72 -0.48 -0.72 -0.48 -0.72 -0.48 -0.72 -0.10 -0.48 -0.72 -0.72 -0.48 -0.72 -0.48 -0.72 -0.48 -0.72 -0.72 -0.48 -0.48	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32 3.73 11.09 -3.28 -0.26 12.23	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10 -0.80 -3.90 0.90 -7.61 -2.49	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90 -5.18 -12.07 -4.89 -7.36 -6.44	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09 -2.03 -1.50 -2.45 -1.08 2.05
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32 33 34 35 36 37 38	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03 27.44 27.51 3.21 -14.66 3.65 -30.40	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51 -0.48 4.81 -1.38 -0.72 -11.76 -5.70	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32 3.73 11.09 -3.28 -0.26 12.23 8.56	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10 -0.80 -3.90 0.90 -7.61 -2.49 -8.09 -8.09 -8.09	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90 -5.18 -12.07 -4.89 -7.36 -6.44 1.06	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09 -2.03 -1.50 -2.45 -1.08 2.05 8.22
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32 33 31 32 33 34 35 36 37 38 39	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03 27.44 27.51 3.21 -14.66 3.65 -30.40 10.03	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51 -0.48 4.81 -1.38 -0.72 -11.76 -5.70 2.31	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32 3.73 11.09 -3.28 -0.26 12.23 8.56 15.22	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10 -0.80 -3.90 0.90 -7.61 -2.49 -8.09 0.22 0.22	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90 -5.18 -12.07 -4.89 -7.36 -6.44 1.06 1.73	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09 -2.03 -1.50 -2.45 -1.08 2.05 8.22 2.64
Ye Branch 20 21 22 23 24 25 26 27 28 29 30 31 32 33 34 35 36 37 38 39 All	ar 1983 19.89 28.59 -4.84 5.85 40.22 16.80 32.25 27.02 26.86 34.14 1.39 65.72 -5.03 27.44 27.51 3.21 -14.66 3.65 -30.40 10.03 12.48	1984 10.50 5.67 -8.22 0.28 -0.37 -9.28 -7.97 -1.16 -9.15 11.43 -2.74 -4.67 -5.51 -0.48 4.81 -1.38 -0.72 -11.76 -5.70 2.31 -0.5 -0.5	1985 2.47 1.41 3.92 4.29 -6.47 -3.83 -12.03 4.47 -18.11 5.46 -9.16 -11.28 -8.32 3.73 11.09 -3.28 -0.26 12.23 8.56 15.22 0.07	1986 -1.74 -5.76 9.71 3.48 -1.23 -1.23 -3.39 3.85 -2.56 3.13 -2.86 0.96 -2.10 -0.80 -3.90 0.90 -7.61 -2.49 -8.09 0.22 -0.62	1987 1.61 -6.58 -25.05 -2.00 -0.48 -6.70 3.21 7.72 -0.87 -11.94 -5.72 -3.04 7.90 -5.18 -12.07 -4.89 -7.36 -6.44 1.06 1.73 -2.77	1988 5.24 -5.63 13.05 -4.76 1.14 -3.13 -2.40 -24.10 -2.52 -6.74 -1.77 -3.97 0.09 -2.03 -1.50 -2.45 -1.08 2.05 8.22 2.64 -0.68

ANNUAL CHANGES IN MANUFACTURING EMPLOYMENT OF WAGE EARNERS BY BRANCH 1965-1988

CHAPTER 5: RELATIVE WAGE FLEXIBILITY IN GREEK MANUFACTURING

5.1. INTRODUCTION AND SUMMARY.

Given the extent of wage and employment regulation in Greece, as analysed in earlier chapters, this chapter examines the formation of relative wages at a disaggregated level of analysis. It focuses upon flexibility in relative wages at the level of twodigit industrial branches. In theory there can be three different situations. First, there can be lack of flexibility, second, there can be competitive flexibility and third, non-competitive flexibility may be observed. While the distinction between rigidity and flexibility is rather a matter of degree differences with regard to market outcomes, the distinction between competitive and non-competitive flexibility is, rather, a matter of kind.

This chapter examines the extent to which, the Greek manufacturing wage structure was responsive or not either to competitive influences or to industry-specific conditions. Industry-specific changes in productivity, rates of strike activity and changes in employment are examined as potential sources of non-competitive flexibility in relative wages. Moreover, as the two different types of flexibility may exert opposite influence upon labour allocation and employment growth, this chapter discusses the extent to which the asymmetric response of wages to productivity, if any, was employment

enhancing or not.

Section 5.2 discusses the theoretical aspects of competitive and non-competitive flexibility in industry wages and the relevant findings of the empirical research in this area. Section 5.3 clarifies the hypotheses which are tested in this chapter and presents the methodology adopted. Section 5.4 presents and discusses the findings of the bivariate correlation analysis for testing the competitive hypothesis. Section 5.5 presents and discusses the results of the cross-section regression analysis for industry-specific determinants of relative wages. Finally, Section 5.6 summarises the conclusions on relative wage flexibility in Greek manufacturing.

Our analysis suggests that over the period 1966-88 the manufacturing labour market operated far from the competitive model, with the noticeable exception of the 1967-74 sub-period, during which competitive forces were given a chance and showed themselves up. But throughout the 1966-88 period non-competitive flexibility associated with industry-specific changes in productivity in the long-run, and industry-specific rates of strike activity in the short-run, characterised relative wages in the Greek manufacturing sector. Furthermore, over the period 1977-88, the asymmetric response of industry wages to changes in industry productivity hindered employment growth. Overall, the analysis suggests that wage and employment regulation disfavoured competitive flexibility and provided an environment conducive to non-competitive flexibility.

5.2. RELATIVE WAGE FLEXIBILITY AND RELATIVE EMPLOYMENT.

5.2.1. The Flexibility Debate and Relative Wages.

Most of the theoretical issues underlying the labour market flexibility debate are not new. But only since the early 1980s, have they become in the centre of the interest of economists, industrial relation specialists and politicians. The flexibility debate mounted to address the failure of the traditional mechanisms of wage and employment regulation to allow for necessary adjustments in face of changing international economic conditions in the 1970s. Since then the boundaries of the flexibility debate, both in theory and in policy making, have not been stable.

In a bibliographical essay on the labour market flexibility literature Gaudier (1987) aimed to show the state of the discussion of labour market flexibility under way in industrialised countries. On the central question what is flexibility she concluded that "interpretation depends on cultural, scientific and organisational factors that broaden the range of meanings ascribed to it and of the circumstances it covers". However, from the analysis it is made clear that labour costs, conditions of employment and hours of work are the mostly discussed issues. Moreover, Gaudier (1987:44) pointed that "wages are in the forefront of the battle" as the most frequent criticisms are of wage determination and indexing systems,

centralised wage bargaining, rigid opposition to reduction of wages and increase of wages without any counterpart in increased productivity.

Developments in labour markets with regard to flexibility lead to similarly varied conclusions. The comparative anthology by Sarfati and Cobrin (1988) illustrated some of the fundamental developments in labour market flexibility and summarised the evidence on different forms of flexibility. On the latter issue, Sarfati and Cobrin argued that "there are as many forms of flexibility as there are components in the employment relationship" and that "they all form the subject of compromises which reflect a number of parameters specific to the country, the sector or the enterprise in question". In the same line of argument Sarfati and Cobrin (1988:39) stress that "there is a general move towards greater flexibility, but ... the methods of achieving it vary according to the economic, social and institutional context in each country".

However, despite of the wide variety of approaches and issues raised in the flexibility debate, the review of the literature as well as of policy developments suggests that in several countries and in particular those with more centralised bargaining systems and indexation policies, such as Italy, Spain, Belgium, Denmark and Australia, the flexibility debate is mainly focused upon policies for wage moderation and de-indexation. Furthermore, from the technical report of OECD (1986) and Metcalf' s (1987) survey of evidence on labour market flexibility , we can infer that, in the broad area of issues raised by the flexibility

debate, the issue of wages remains central. Indeed, the issue of relative wages can be considered as one key issue of wage flexibility.

With regard to relative wage flexibility three situations can be defined. In theory, these three situations may imply varying consequences as far as labour allocation and employment growth are concerned. First, the situation of wage rigidity refers to wages lacking responsiveness to either unfavourable macro- or micro-economic changes. In such a situation, the labour market adjusts through quantities, that is employment and unemployment. Second, the competitive flexibility situation in which wage flexibility, by definition, allocates employment efficiently and, thus, enhances employment growth. Third, the non-competitive flexibility situation when the causes of and gains from flexibility are related to industry-specific factors such as productivity and unionisation. In such a case, the employment consequences of wage flexibility may be either positive or negative.

5.2.2. Competitive Flexibility in Relative Wages.

In the ideal world of competitive and clearing labour markets there is not much room for unemployment and inefficient allocation of labour. Relative wages adjust any imbalances and, thus, after all, they only reflect skill differences. Competitive flexibility in relative wages delivers, by definition, efficient allocation of labour and employment growth.

According to the competitive model, given the short-run inelasticity in labour supply, relative wages play an essential allocative role. They are used to attract workers in expanding sectors. Therefore, there should be a positive relation between wage and employment changes. In the context of either structural changes or business cycle fluctuations, greater wage growth will in expanding compared to contracting sectors. As be observed already mentioned, according to the competitive model the relationship of greater wage growth in expanding sectors should be observed only in the short-run, because, only then is labour supply considered to be relatively inelastic. Overall, recent and past evidence, summarised in OECD (1985:93-94) and Metcalf (1987:62-63), shows that there is no strong, and sometimes not even positive, significant correlation between employment and wage changes in the short-run.

But the lack of strong evidence supporting the competitive model with regard to the positive relationship between changes in employment and changes in wages is not considered as invalidating the competitive theory itself. In the literature two types of explanation have been offered. They both call for more sophisticated technical elaboration of the prevailing labour market conditions and focus on the conditions required for a positive correlation between changes in wages and employment to be observed.

In the first type of explanation is stressed that the short-run elasticity of labour supply is also affected by the state of the
labour market. If there is high unemployment then the short-run labour supply may be relatively more elastic than in a tight labour market. This implies that in periods of either loose or tight labour market the role of wage differentials may vary considerably. From the existing evidence, which overall does not strongly support the competitive model, and in particular that summarised by Reder (1962:290) and OECD (1985:94), it can be inferred that the little evidence which supports the competitive hypothesis refers mainly to periods of low unemployment and tight labour markets, where the short-run supply of labour is more inelastic.

In the second type of explanation belongs the methodological criticism, developed by Keating (1983) and Withers et al (1983), of the bivariate correlation analysis as inappropriate for testing the competitive model. Both emphasise that the influence of relative wages changes on the distribution of employment can only be adequately assessed if supply effects are distinguished and there is a separate consideration of shifts in the supply of labour from shifts in demand. Though, both authors point out the difficulties, mainly arising from data unavailability, for applying this type of analysis.

If relative wages are not sensitive to increasing employment in the short-run, then what is the alternative method through which firms attempt to attract or retain their labour force? Metcalf (1987:61-62), reiterated Reddaway' s institutional approach to the wage-employment relationship. This approach by Reddaway

(1959) was emphasised as the alternative to the competitive model. While in the competitive model labour allocation works through price signals, in the institutional approach labour allocation works through quantity signals for vacancies. In the institutional approach "the essential characteristic is that the main way in which employment will either be increased or reduced is through direct action by the employers and that only exceptionally will they have to include a change in the relative wage offered in order to secure the desired number of workers". (Reddaway, 1959:33).

.

Thus, relative wage flexibility is not considered as the only way for efficient allocation of labour. Indeed, as Pissarides and McMaster (1984:1-2) pointed out "quantity signals could also signal relative labour requirements and the reallocation of labour could take place regardless of relative wages". This could happen through the higher vacancy rate in the expanding sectors. "Generally a higher vacancy rate implies a lower incidence of unemployment in the sector. Workers interested in maximising incomes over a long horizon would respond positively to such change ... they would move to the expanding sector to take advantage of the improved employment prospects. In this way labour could be reallocated between sectors and so some kind of market mechanism would work, without changes in relative wages".

5.2.3. Non-Competitive Flexibility in Relative Wages.

In a competitive labour market relative wages are not influenced by different rates of productivity growth unless the latter

reflects changes in occupational composition of the labour force. Supply and demand adjust in terms of wages and employment so that the gains from different rates of productivity are equally distributed across the board via employment and wage growth. While relative wage flexibility is not the only way for efficient labour allocation, relative wage flexibility, when observed, does not imply that wages are responsive to competitive forces. Reddaway in his analysis of the institutional approach, where employers do not use the wage (price) signals as the main mechanism for increasing or decreasing employment, discussed also the case of non-competitive flexibility in relative wages. The same case has been recently analysed by Freeman and Bell (1985).

Even when a positive association between change in wages and employment had been observed, Reddaway noticed the possibility that institutional industry-specific factors, unions for instance, might had caused that association. For instance, when demand for labour increases, unions may capture this opportunity and capitalise upon it by increasing their wage premium. In Reddaway's (1959:44) own words, "a higher wage might be necessary to attract workers to an expanding industry, or it might simply reflect the fact that the union concerned has taken advantage of the increasing level of demand to negotiate a higher wage, even though enough workers would in fact have been forthcoming at the old wage in response to offers of employment. We cannot hope to establish that the wage increase was necessary to attract workers, we can only see what in fact happened".

In fact, the rationale described by Reddaway reappears in Bell and Freeman's definition of non-competitive flexibility. According to Freeman (1987:2) non-competitive flexibility refers to the situation "in which changes in wages respond to conditions that are not associated with supply and demand factors and thus do not facilitate employment changes. Our example of noncompetitive flexibility would be wage increases in a highly profitable high-wage firm that is not necessary to attract or retain labour. Another would be union-induced increases for workers already paid above-market rates." The slight difference between Reddaway's and Freeman's definition is that Reddaway put the emphasis upon union influence which, in turn, might have been strengthened in conditions of expanding employment. Bell and Freeman, as they sought a basis for comparisons to the findings of the study by Salter (1960) as well as to past work on the U.S. wage structure, emphasised productivity as the key factor in shaping relative wages, without industry-specific dealing with either other conditions or the processes which allow relative wages to follow changes in productivity.

The recent evidence on the existence of industry-specific noncompetitive flexibility stresses the importance of this type of flexibility. Salter (1960) in his analysis of inter-industry wage structure in the U.K., found that changes in wages among industries were unrelated to industry conditions, and in particular were not related to changes in productivity. He interpreted this as indicating the operation of a competitive

labour market. Bell and Freeman (1985) reported a different result. They found the U.S. wages rising most rapidly in sectors having rapid increases in productivity and/or product prices. An OECD (1985) study found weaker tendencies of industry-specific factors which influence wage changes. For the 1979-84 period in U.K. manufacturing, Freeman (1987) found relations between industry-specific productivity growth and wage growth that were quite strong and in contrast to the earlier findings by Salter. In other words, the evidence suggest that, even if this is not consistent with the competitive hypothesis, changes in wages which reflect industry level conditions consist a form of wage flexibility which sometimes may be found rather important.

5.2.4. Non-Competitive Flexibility and Employment Growth.

As already mentioned, recent evidence suggests that in many cases the existing flexibility is related to industry-specific factors rather than to the competitive working of the labour market. In theory the non-competitive flexibility can boost or lower aggregate employment. Therefore when changes in productivity are the key industry-specific factor influencing relative wages, the question is whether the positive relation between changes in wages and changes in productivity is employment enhancing or employment hindering. Freeman and Bell (1985) analysed this issue on theoretical grounds and tested for the employment consequences of industry wage flexibility in the U.S. According to this analysis the positive relation between growth of industry productivity and growth of industry wages may boost employment if

wages respond more to relative declines in productivity than to relative increases in productivity. Only such an asymmetry in the response of wages to changes in industry productivity may result to a net positive outcome in terms of employment growth. Because more jobs may be "saved" in industries where wages are more responsive to relative declines in productivity, and less jobs may be "forgone" in industries where wages are less responsive to relative increase in productivity.

In practical terms ,the required asymmetry should exist between two broad sectors which are defined in terms of above-average and below-average changes in productivity. As the OECD (1986:31) study summarised the rationale on the asymmetric response "if these two "sectors" are of equal size, have equal elasticities of labour demand, then greater employment than otherwise would have occurred will be observed only if there is significant asymmetry in the wage response. If the asymmetry hypothesis has any explanatory power, the coefficient [for sectors] of below-average changes in productivity should be positive significant and greater than that [for sectors] of above-average changes in productivity."

The evidence on the employment consequences of productivityrelated flexibility in relative wages is contradictory. Freeman and Bell (1985) found that among U.S. industries in the 1970s, industries with rapid productivity growth tended to have lower rather than higher employment growth. This implies that, in that case, flexibility in relative wages was employment hindering. In

other words, they found that kind of wage flexibility inconsistent with the competitive hypothesis as far as employment growth is concerned. Salter had addressed the same issue and had found a strong positive correlation between productivity growth and employment growth in the U.K. An OECD (1986) study found the opposite results for the U.S. than those presented by Bell and Freeman, and argued that the industry wageproductivity system in the U.S. was not employment hindering. It also found results supporting the asymmetrical wage response for Canada and Sweden as well.

These contradictory findings with regard to the employment consequences of industry-specific wage flexibility in the U.S. are mainly due to methodological differences in testing for the existence of asymmetry in the wage response to changes in productivity. Bell and Freeman (1985) in testing if such asymmetry holds for the U.S. allowed for asymmetric responses of wages to above- and below-average productivity increases. As the coefficient for sectors of above-average changes in productivity was found significant and greater than the coefficient for sectors of below-average changes in productivity, wages were found more flexible upward than downward. The OECD study reported, contrary to Bell and Freeman, that the U.S. wage structure did exhibit the required asymmetry. According to the OECD (1986:44) study the reason for these different results is caused by the specification of productivity changes as independent variable in the regression analysis. Bell and Freeman used as independent variable the actual industry wage changes,

while the OECD study used the actual difference between each industry wage change and the average wage change across all industries, which stands for the industry relative wage change.

5.2.5. Summary.

To summarise the analysis of this section, flexibility in relative wages can be either competitive or non-competitive. But even when the positive association between changes in employment and changes in wages is observed, it may be not evidence of competitive flexibility and, indeed, it may be due to industryspecific factors. Non-competitive flexibility may be caused by industry-specific factors such as productivity changes, union bargaining power, high profitability etc. In the relevant literature, productivity changes are considered as the key industry-specific factor influencing relative wages. The consequences of productivity-related wage flexibility on employment growth, given a positive association between changes in industry wages and industry productivity, depend upon the asymmetry of industry-specific wage flexibility with regard to productivity changes. When wages are more flexible upward (downward) than downward (upward), they may cause relative job losses (gains) and, thus, contribute to employment decline (growth).

5.3. HYPOTHESES TO TEST AND THEIR RATIONALES.

5.3.1. Relative Wage Flexibility in the Context of Greek Industrial Relations?

Did any relative wage flexibility, competitive or noncompetitive, exist over the period 1966-88? The overall context of wage and employment regulation in the Greek manufacturing sector suggests that, prima facie, there was not much room for such flexibility. However, as Tables 4.3.6 and 4.3.7 indicate there was some variability in the wage structure. This variability should be explained. It seems reasonable to attribute part of the observed variability to incomes policies themselves, and to consider it as a by-product of them, to the extent that, during the period under consideration, they treated relative wages in varying ways.

Even if the regulation of wages and employment by successive governments have been conducive to rigidities in the labour market, this does not imply that, at the end of the day, the labour market outcomes should be considered as unilaterally determined by governmental policies. Governmental policies should rather be perceived as one source of influence among others. The extent of the influence exerted by market and institutional forces is a matter of evidence. Econometric tests can provide the basis for attributing the observed variability in relative wages either to incomes policies themselves, or to competitive and noncompetitive industry-specific factors lying beyond incomes policy

and out of the range of governmental controls.

Wage regulation either by governmental policies or under collective bargaining between employers and trade-unions is neither a new phenomenon nor a Greek specific characteristic. It has been observed in the labour markets of Western developed economies since the early decades of the century. As Hicks (1932:vi) put it in the introduction of "The Theory of Wages" "the historical fact which dominates the wage history of the present century -both in Britain and in other countries- is the growth of Trade Unions power and the development of state regulation of wages...". But, according to Hicks " ... the same forces which determine wages in a free market are still present under regulation; they only work rather differently." Or, as he put it when dealing with the working of competition and the conception of the wages of labour as a "system", following Clay' s approach, "the forces elucidated by equilibrium analysis are the forces which, in nearly every case, cause wages to change." (Hicks, 1932:86).

This theoretical argument, according to which even in regulated labour markets, that is markets under conditions of state intervention and trade unions, the main forces of the competitive model should be considered as if they are still at work, provides the basis upon which our own hypotheses are built.

In the context of the U.K. economy, from which Hicks drew empirical observations for theorising, it may be true that competitive forces have been working even under governmental

regulation and widespread collective bargaining. Indeed, what Hicks suggested may be true in the case of most industrialised market economies. The evidence discussed in Section 5.2 indicates that, to a certain extent, competitive forces have been at work regardless of institutional conditions. But whether or not this has happened in the case of Greece is a more open question, because Greece has followed a dissimilar trajectory in economic development as well as in the evolution of industrial relations. In Western developed economies there has been a sequence of developments which occurred during nearly a full century. This sequence is not observed in the case of Greece where developments such as industrialisation, governmental intervention and formation of trade-unionism occurred in a period of less than three decades.

In Western developed economies the industrialisation process was followed by the rise of trade-unionism and, then, its establishment as a functional part of the economy through institutions and processes for collective bargaining. Governments intervened in wage determination to control the bargaining power of trade-unions by adopting incomes policies. Recently, the flexibility drive was adopted in order to modify long-lasting characteristics in the workings of institutionalised labour markets.

In the case of Greece, similar phenomena were developed in a shorter period of time and without following the historical sequence of the Western developed economies. Industrialisation took place during the post-war period and mainly in the 1960s.

Governments have been intervening in wage determination from the very beginning of the post-war industrialisation process. There has been no long tradition of trade-unionism and in manufacturing trade-unions have flourished since 1975, after a seven year period during which free trade-unionism, collective bargaining and strike activity were essentially banned.

In this context, it seems reasonable to assume that there have been important differences with regard to flexibility in relative wages between those national labour markets extensively studied in the flexibility literature, mainly those of the U.S. and the U.K., and the Greek labour market. Such differences may have been caused, first, by the degree of wage and employment regulation and its dominance in the working of the Greek manufacturing labour market, and second, by the incidence of major disturbances in the system of industrial relations in a short period of time. This is not to say that in the case of Greece "economic laws" have not been at work. It rather means that, presumably, they have not been allowed to work properly, or to the extent they have done in the context of other industrialised market economies.

Similar phenomena can probably be observed in other national labour markets and the relevant question about flexibility can be raised concerning every national labour market and its relative position in an international context. Up to the present it appears that there has been a gap in the OECD and EEC literature on labour market flexibility in countries such as Greece, Spain

and Portugal which share some common characteristics as far as economic, political and industrial relations developments are concerned.

The Greek-specific characteristics which may have influenced flexibility in manufacturing relative wages have been discussed in earlier chapters, and include first, the incomes policies and the extensive use of compulsory arbitration, second, the legislation on collective dismissals, third, the conditions of rising manufacturing employment along with waves of migration of Greek labour abroad, fourth, the ban of the right to strike in 1967-74 and the radical changes in manufacturing industrial relations initiated by the rise of strike activity and the creation the informal system of collective bargaining after 1975. This rich combination of factors, in the context of extensive and permanent regulation of wages and employment, may have created both rigidities and flexibilities depending on the exact blend of factors prevailing in distinct periods.

From this point of view, we address the questions whether the wage and employment regulation has buttressed wage rigidity and whether manufacturing wages have been flexible and to what extent. More specifically, we attempt to clarify whether or not forces which work in conditions of competitive labour markets, or forces related to industry-specific conditions have been present in the labour market of the Greek manufacturing sector and, thus, the extent to which there has been any scope for either competitive or non-competitive wage flexibility.

Both types of flexibility are related to the firms and employees behaviour in the context of governmental, procedural and substantial, regulation. Therefore the underlying questions are: Given the governmental wage policies, what has been the role of firms and trade unions in the regulated labour market? Have they acted in line with forces that the competitive theory considers as present even under regulation? Have employers raised the wage rate to attract labour in expanding sectors, enhancing, thus, a better allocation of labour? Have employees shared any industryspecific gains such as rises in productivity? With what consequences upon employment allocation and growth? Have industry-specific factors been related to industrial relations such as high bargaining power or strike activity?

Finally, if any of the above phenomena are observed, do they refer to the full period under examination or they are associated with distinct sub-periods? In testing for the presence of competitive and non-competitive flexibility during four distinguishable periods, namely 1966-88, and the sub-periods 1967-74, 1975-81 and 1982-88, the issue under investigation is also whether there were period differences in the way that relative wages responded to competitive or industry-specific factors. This investigation may shed some light on the flexibility performance during each separate period and, through it, to the flexibility performance of distinct types of wage policy.

To address these questions on competitive and non-competitive flexibility in relative wages bivariate correlation analysis and

cross-section regression analysis are combined. Both types of analysis can provide evidence on the function of relative wages over the period 1966-88 and its distinct sub-periods.

5.3.2. Competitive Flexibility in Relative Wages.

Overall, the long-lasting conditions of wage and employment regulations lead us to the hypothesis that there has been not much room for competitive forces to exert themselves in the context of the Greek manufacturing sector. In other words, the extensive regulation of the labour market may have not been in favour of the use of relative wages as a mean for labour allocation.

Incomes policies, by restraining wage determination processes, presumably do not favour competitive flexibility in relative wages. Labour market conditions of relatively high urban unemployment and the existence of the pool for unskilled labour in the primary sector do not exert pressures upon firms and employers to use higher relative wages as a means for attracting or retaining labour. The legislation on collective dismissals may have caused the firms to expand gradually their employment, at first through increased overtime work and then, when the need for expansion was not seasonal, through hiring new workers in a loose labour market.

However, we may expect episodes during which competitive forces were given a chance and showed themselves up, as there were

important changes in the labour market conditions. The coexistence of high urban unemployment along with concealed unemployment and under-employment in the rural sector had created loose labour market conditions. Unskilled labour has never been in short supply. But in the late 1960s-early 1970s exceptional conditions were created in the labour market, especially in the manufacturing sector. Labour, mainly the skilled and semiskilled, became, partially due to massive migration abroad, in short supply. At the same time, as indicated in Table 4.3.2, the level of unemployment reached its lowest levels ever recorded in post-war Greece.

With regard to competitive flexibility in relative wages, we postulate that during the period 1967-74 were the prevailing conditions favourable for competitive forces to work in the manufacturing labour market, for four reasons. First, it was a period of low unemployment. Second, it was a period during which manufacturing employment was rapidly expanding across the board. Third, there were not trade-union activities and strikes. And fourth, for most of that period, changes in manufacturing wages imposed by the government in office were falling behind average productivity increases, creating thus, a space for manoeuvre concerning industry wage setting for manufacturing employers.

Such preconditions were not observed in the following periods. First, the growth of manufacturing employment slowed down. Second, unemployment soared. Third, collective bargaining came back in the manufacturing sector via waves of wildcat strikes at the plant level. And, fourth, formal incomes policies limited

the space for manoeuvre available to manufacturing employers. Since 1975 the continuous use of formal incomes policy created an environment not favourable for competitive relative wage flexibility. Of course, after 1975 tendencies towards more decentralised wage setting processes developed along with the reestablishment of collective bargaining and the rise of the informal system of collective bargaining in the manufacturing sector. But such developments in the manufacturing bargaining structure may have been a factor causing industry-specific noncompetitive flexibility rather than competitive flexibility in relative wages.

To test for the presence of competitive flexibility in relative wages in the Greek manufacturing sector we examine the bivariate correlation between changes in wages and changes in employment. A positive correlation between changes in employment and changes in wages is considered as evidence of competitive flexibility. However, as Reddaway (1959:44) put it, "any such relationship cannot prove anything but it is worthwhile to see what the data To allow for the case where supply shifts cause a suggest". negative correlation between wage and employment changes, which can be consistent with the competitive model, systematic evidence on the bivariate correlation along with factual support concerning the institutional and economic state of the labour market over the period under examination are considered as necessary for evaluating any results indicating competitive flexibility in relative wages.

5.3.3. Non-competitive Flexibility in Relative Wages.

While both economic conditions and industrial relations prevailing in the Greek manufacturing sector over the period 1966-88 may have not been very conducive for the development and the expression of competitive forces, they may have been relatively more favourable for industry-specific factors to influence the formation of industry relative wages. Probably, regulation does no allow the competitive forces to dominate the workings of the labour market. But it seems reasonable to hypothesise that in the context of first, continuing industrialisation, second, major disturbances in the system of industrial relations and third, the continuous use of formal regulation "favours" non-competitive incomes policies, flexibility in relative wages. Because, in face of successive governmental policies, firms and groups of employees who "beat the gun" and evade from policy restrictions become more flexible as far as wage determination is concerned.

In other words, we presume that under policies of regulation, which by being continuously used they create a climate or culture of regulation in the labour market, flexibility -if anyis probably non-competitive, that is related to industry-specific factors and arising from deviant behaviour against the policies of regulation. Overall, industry-specific changes in productivity and industry-specific bargaining power and strike activity may have been sources of industry-specific non-competitive wage

flexibility in the Greek manufacturing sector. From this point of view, non-competitive flexibility may be more a source of problems for policy implementation rather than a solution to labour market imbalances.

As wage policies were, for the most of the 1966-88 period, mainly targeted to control the employees side and to contain wage increases in line with average productivity growth, employers may have shared with their employees any industry-specific gains arising from changes in productivity, or may have had to increase wage rates in order to retain or attract the more productive skilled labour. Furthermore, the reconstitution of manufacturing trade-unionism at the plant level along with the rise of the informal system of collective bargaining consist another source of potential industry-specific influence upon wage formation. Employers may have been under pressure, arising from collective bargaining power and/or strike activity, to concede above-average increases in nominal wages, justified or not by industry-specific changes in productivity.

Of course, such industry-specific influences vary from period to period. For the 1967-74 period, with neither formal incomes policy in force nor, from April 1967, lawful strike activity and essential collective bargaining processes, we hypothesise that there was not much room for industry-specific factors, such as bargaining power expressed in relatively high unionisation and strike activity, to exert any decisive influence upon industry wages. However, this does not exclude the possibility that firms may had shared industry productivity gains with their employees

even in the absence of collective bargaining and strike activity procedures.

In 1975-81 the economic and industrial relations environment changed dramatically and under the norm-based wage policy and the rising system of informal collective bargaining in the manufacturing sector, industry-specific influences, related to strike activity and productivity growth, may have affected industry wages. Then, in 1982-88 the wage policy of partial indexation, as it was explicitly designed to compress the wage structure, imposed stricter controls upon industry wages. Thus, while both wage policy and industrial relations became less favourable for industry-specific wage flexibility, the lack of effective controls regarding compliance with mandatory incomes policies as well as the occurrence of waves of strikes indicating the presence of industry-specific factors, may have allowed some room for non-competitive flexibility in relative wages.

To test for the presence of non-competitive wage flexibility in the Greek manufacturing sector we examine the influence of industry-specific factors upon changes in industry wages by using cross-section regression analysis. We test for the influence exerted upon changes in nominal wages by changes in productivity, levels of industry strike activity and changes in employment.

In the literature, productivity is considered as the key industry-specific variable. In Bell and Freeman (1985) and OECD (1986), for instance, the key industry-specific variable is the

value productivity per worker. As the change in productivity reflects technical change, relative price changes and changes in the composition of an industry's employment it is not possible to sign it a priori. If changes in wages and changes in productivity are not related this result should be considered as consistent with the competitive model. A positive relationship indicating that employees have shared industry-specific productivity gains should be considered as consistent with the non-competitive flexibility hypothesis and with the workings of a decentralised and informal system of wage determination. A negative correlation between productivity and wage changes indicating that wages are determined regardless of industry productivity growth, as well as against the variability of productivity growth across industrial should be considered as consistent with the branches implementation of incomes policy via either centralised collective bargaining or the machinery of compulsory arbitration.

In the context of informal collective bargaining in the Greek manufacturing sector after 1975 the industry-specific level of strike activity represents an institutional source of potential industry-specific influence upon industry relative wages. Of course, a straightforward positive relationship should be expected only if strike activity indicates bargaining power. And this is not always the case. Strike activity sometimes indicates only dissatisfaction and militancy rather than bargaining power. Instead of bargaining power can simply reflect failed attempts to influence the process of wage determination. Moreover, even when strike activity does indicate bargaining power, it is possible

that this bargaining power is demonstrated once, of course not for all, but at least once for a period of time which can vary considerably. This demonstrated bargaining power, in turn, may have a threat effect during recurring annual bargaining rounds.

From this point of view, the association between changes in wages and the rate of strike activity is considered as indicator of industry-specific influence upon relative wages exerted by the employees collective action. A positive relationship is expected to indicate that strike activity, ascertaining bargaining power, has led employers to concede further increases in industry wages. A negative association may denote the function of strike activity as a safety valve or as an expression of inefficient militancy and a way of expressing discontent.

Changes in industry employment are usually considered as a source of competitive influence upon changes in nominal wages, in the short-run. But they may also be associated with industryspecific influences as employment growth can affect positively the bargaining power of the workforce already inside industries expanding employment. Therefore, while a positive association . between changes in wages and changes in employment in the shortrun should be considered as evidence in favour of the competitive hypothesis, in the long-run this relationship should not be observed. If it does then this may indicate an industry-specific influence arising from additional, explicit or implicit, employees in bargaining power of industrial branches characterised by long-run expansion.

To test for the presence of these industry-specific influences upon relative wages we use cross-section regression analysis. As Freeman (1987:1) argued, the analysis is better cast in terms of cross-sectional changes for three reasons. First, it allows to "difference-out" period effects. Second, it permits to by-pass the problem of evaluating time-series evidence when all variables move, more or less, together. Third, because it provides greater number of observations. However, Freeman considered as a disadvantage of the cross-section approach that we cannot be certain that factors which affect wage structures have similar impacts on changes over time. For our analysis in order to estimate and evaluate any sub-period effects of industry-specific variables, it is important to develop the analysis, not only for the full period 1966-88, but for its sub-periods as well. Therefore, we also test for the presence of determinants of industry-specific wage flexibility during three distinct subperiods of wage determination processes and incomes policies.

Finally, in the eventual presence of asymmetries in changes in wages and changes in productivity the issue of the employment consequences of this hypothesised asymmetry becomes important. The question is whether this asymmetric responsiveness of wages to productivity at the industry level boosted or lowered aggregate employment. Furthermore, the policy issue is whether policy for employment growth via employment regulation and protection was undermined by industry-specific flexibility in relative wages arising from asymmetrical changes in productivity, and whether the restrictive legislation has been

mistakenly considered as the main way for securing jobs. Indeed, the question is whether, in some cases, the employment and wage regulation measures were contradictory as long as, indexation for instance, caused a certain degree of rigidity in relative wages especially in low-wage low-productivity change industries, which, in turn, caused job losses, both potential and actual.

5.4. THE COMPETITIVE HYPOTHESIS: ANALYSIS OF BIVARIATE CORRELATION BETWEEN WAGE AND EMPLOYMENT CHANGES.

According to the competitive hypothesis there should be a positive correlation between changes in wages and changes in employment unless supply shifts predominate in labour market developments.

To test for the existence of any positive correlation between changes in industry nominal wages and changes in industry employment across the twenty two-digit industrial branches of the Greek manufacturing we use two slightly different approaches. First, we estimate the correlation coefficients relating changes in nominal wages and changes in employment for the whole period 1966-88, as well as for the sub-periods 1967-74, 1975-81, 1982-88, which are characterised by different wage policies and varying conditions with regard to demand for labour in the manufacturing sector. These period correlation coefficients are Second, we estimate the correlation reported in Table 5.4.1. coefficients for each year from 1966 to 1988 separately. These year correlation coefficients are reported in Table 5.4.2. Changes in nominal wages refer to average hourly wage of workers of both sexes and changes in employment refer to wage earners.

In both cases, correlation coefficients are estimated for changes in the short-run and in the long-run. According to the competitive theory, because of the short-run inelasticity in labour supply, significant and positive correlation should be observed only in the short-run. The market mechanism is supposed

to adjust any imbalances in longer periods of time. Therefore we estimate the correlation coefficients for both short- and longrun changes in wages and employment in order to detect any weakening of their association in relatively longer periods of time. For this purpose, a continuum-like set of coefficients, ranging from one-year changes to twenty three-year changes, including two-year, three-year, five-year and ten-year changes, is reported.

Overall, the period correlation coefficients for changes in industry wages and employment in the Greek manufacturing sector provide no systematic evidence in favour of the competitive hypothesis. Correlation between changes in wages and employment is found negative when the full period 1966-1988 is examined. It seems that in the period 1966-88, the manufacturing labour market in Greece was operating far away from the competitive model in the sense of the use of relative wages to retain and attract labour in expanding industrial branches. This is probably due, first, to the extensive regulation in terms of both wage and employment determination and, second, to employment developments in the manufacturing sector.

When the sub-periods characterised by distinct types of wage policy and varying levels of unemployment are examined, the estimated correlation coefficients do not provide systematic evidence supporting the competitive hypothesis either. However, the correlation analysis suggests that competitive flexibility existed only during 1967-74. This result is supported by the

facts concerning the state of the labour market as well as the state of industrial relations at the time. In that period there was no formal incomes policy in force. There was no collective bargaining and strike activity. There was a tight labour market (low levels of unemployment). And, after all, that was the main period of expanding manufacturing employment in Greece. Such an association is observed in the case of one-year changes. Coefficients for two-year and three-year changes are also positive but not statistically significant. In addition, the gradual decrease of the estimated correlation coefficients is consistent with the competitive hypothesis, as the supply of labour becomes less inelastic and, thus, wages have to be raised relatively less than over shorter periods of time.

Correlation coefficients for the sub-period 1975-81 are found positive in the case of one-year, two-year and three-year changes and negative in the case of five-year and ten-year changes. But only the latter are significant. A different pattern of coefficients is estimated for 1982-88. They are found negative in the case of one-year and five-year changes. For the remaining cases, they are positive and, indeed, in the case of two-year and ten-year changes are significant as well.

This variety of estimated correlation coefficients hardly denotes the presence of competitive flexibility in relative wages. Certainly, the important criterion for the validity of the competitive hypothesis refers to the association between wages and employment in the short-run. And in the short-run, as the period correlation coefficients suggest, there was not strong,

positive and significant correlation between changes in employment and changes in nominal wages. The estimated coefficients for one-year changes over the period 1966-88 are negative and significant. But they are positive and significant for 1967-74. In the sub-period 1975-81 the estimated coefficients are positive though nearly zero and not significant, while those of 1982-88 period are found nearly zero and not significant either.

The year correlation coefficients, estimated separately for each year from 1966 to 1988, relating changes in industry nominal wages and industry employment, for both short-run and long-run changes, are reported in Table 5.4.2. They also provide not systematic evidence in favour of the competitive hypothesis. Coefficients of one-year changes are found either positively or negatively signed, but the majority of them are not statistically significant. Only the correlation between annual changes in nominal hourly wages and changes in employment of wage earners 1965/66 and 1983/84 are positive and statistically for significant. Thus, two out of three significant coefficients for one-year changes are positive. When we examine changes over a two-year period, three out of four significant coefficients are found positive. For changes over a three-year period two out of three significant coefficients are found positive. But the majority of the estimated coefficients are neither statistically significant nor stable with regard to their sign.

As an OECD (1985:94) study reiterated, it has been frequently

argued that the expected positive wage and employment change association may be obscured by examining the entire range of industries and not the range of industries with employment increases. To test whether such an asymmetry influences the findings of our correlation analysis, we also examine the correlation between annual changes in wages and annual changes in employment only for the case of industrial branches experiencing increases in employment. For this test industrial branches with negative changes in employment are assigned a zero value. This test, the results of which are reported in Table 5.4.3, apart from a slight decrease of the significance levels and the values of the correlation coefficients, does not provide any differing correlation coefficients as far as their significance, sign and level are concerned.

While the lack of systematic evidence of positive correlation between wage and employment changes is considered as lack of support for the competitive hypothesis, in the sense of the use of relative wages to retain and attract labour in expanding industrial branches, the occurrence of negative and significant correlation raise the question how to interpret it. The negative correlation coefficients for the full period 1966-88 seem to be rather systematic as they are significant when one-year, twoyear, three-year, five-year and ten-year changes are examined. Indeed, they increase gradually the longer becomes the period of change under consideration.

According to Keating (1983:386) and Withers et al (1983:7) a negative correlation between wage changes and employment changes

might be consistent with the competitive model if shifts in the supply side of the labour market require lower wages to clear the market or lead to reduced demand for labour in the case of exogenous increases in wages.

A group of explanations based on supply side factors can be concerning the causes of the negative correlation considered between changes in wages and changes in employment. First, the increase in the supply price of labour above its market clearing imposed either by trade-unions or by compulsory level, arbitration and governmental wage policies may provide a rationale for the negative correlation. Such increases probably occurred during the periods of the norm-based and the indexation incomes policy. During the norm-based policy such wage increases may be due to the militancy of trade-unions at the plant level. During the indexation policy such wage increases were probably caused by the system of automatic indexation itself. From this point of view, the negative correlation implies that jobs were being forgone along with the imposed increases in wages. Of in this explanation we need to take into account the course, role of productivity and product market developments.

Second, higher labour supply, even in conditions of rising manufacturing employment, may have required lower wages to clear the labour market. As already mentioned, the coexistence of urban and concealed unemployment along with the labour pool in the primary sector of the economy, remained a permanent characteristic of the Greek labour market in the post-war period.

However, in the 1960s and especially in 1967-74, the immigration of labour abroad created unusual tight conditions in the labour market of the Greek manufacturing sector. As after 1974 the trends of immigration abroad were reversed and expansion in manufacturing employment slowed down, it seems reasonable to assume that a certain shift in the labour supply took place. Moreover, there is evidence (Karantinos, 1989) that since 1975 the rate of female participation in the labour force increased, thus contributing also to a certain shift of labour supply.

Third, the negative correlation between wage changes and employment may be explained by introducing in the analysis the restructuring process in the manufacturing sector. From this point of view, the negative correlation coefficient may reflect developments similar to those in the British nationalised industries, as steal and coal. Those industries, while undergoing a major restructuring process affecting the levels of employment continued to pay relatively high wages to the remaining employees in order to speed up the restructuring process and to avoid major and continuous industrial conflicts.

In 1975-81 the slow down of the manufacturing employment expansion coincided with the two oil-shocks which imposed severe pressures for restructuring and adjustment in the Greek manufacturing industries. These pressures became more severe in 1982-88. In fact, the restructuring and the rehabilitation of wide segments of the Greek manufacturing sector has been a priority of Greek governments in power since 1982. Regardless of the efficiency and the results of this programme, the indexation

policy, which was followed firmly by the industries under governmental control, insulated wage determination from market developments and especially from the market position of ailing industries which while in need for shedding out labour continued paying relatively high wages.

All the above explanations are to a certain extent plausible and relevant to developments in the Greek manufacturing labour market and probably capture parts of the reasons which lie behind the negative correlation between changes in wages and employment. However, even if we can explain the lack of systematic evidence supporting the competitive hypothesis in the sense of the use of relative wages for attracting and retaining labour in expanding industries, the main conclusions of the analysis are two.

First, over the period 1966-88 the Greek manufacturing labour market does not appear to be operating in line with the competitive hypothesis as far as the use of relative wages as means for retaining and attracting labour is concerned. It appears that increases in the supply price of labour, caused by unions, arbitration awards or incomes policies, resulted at systematic negative association between changes in industry wages and industry employment.

Second, during the sub-period 1967-74 both the results and the prevailing conditions suggest the presence of competitive working in the labour market as firms used relative wage increases for retaining and attracting labour in the context of rapidly expanding manufacturing employment, banned strike activity and

wage policy allowing much room for firms' manoeuvre in wage setting. Finally, it is noteworthy that analysis of correlation between changes in wages and changes in employment of both salary and wage earners leads to similar conclusions. The relevant results are reported in Appendix 3.

5.5. INDUSTRY SPECIFIC DETERMINANTS OF RELATIVE WAGES: CROSS-SECTION REGRESSION ANALYSIS.

5.5.1. Methods and Variables.

Our test for the presence of non-competitive flexibility in relative wages is undertaken with regard to both short-run and long-run industry-specific wage flexibility. Long-run noncompetitive flexibility is captured by focusing upon the baseyear to end-year changes in nominal wages as dependent variable. First, in order to explain changes in industry nominal wages from 1966 to 1988, the full period 1966-88 is examined. Then three sub-periods, namely 1967-74, 1975-81, 1982-88, are examined on the basis of the base-year to end-year changes in industry nominal wages. This analysis also refers to long-run industryspecific wage flexibility. Short-run industry wage flexibility is examined by using the annual changes in industry nominal wages as dependent variable. This analysis is too applied for the full period 1966-88 as well as for the sub-periods selected according to the type of incomes policy in force.

Three variables are considered as potential sources of industryspecific flexibility in manufacturing relative wages. First candidate is the rate of change in productivity. We use the industry production index to calculate an output per worker index for industry productivity. This specification reflects technical change and changes in the composition of the industry's

employment. Second potential source of non-competitive flexibility is the rate of industry strike activity as expressed by the index of hours lost in strike activity per 1000 employees. Third potential source of industry-specific flexibility is the rate of change in each industry's employment.

In addition to these three variables which are tested as potential sources of industry-specific wage flexibility, two more variables are included in the regression specifications. First, the proportion of female employment in industry employment is included, to control for the influence exerted by the fact that, as indicated in Table 4.3.1, female workers are on average paid less than male workers, namely they get 60 to 70 per cent of the average male wage. Second, the initial level of the industry nominal wage is included, to capture the influence exerted by wage policies upon the manufacturing wage structure. The OECD (1986)study included also as an independent variable, the absolute change in the percentage of non-production workers to total employment. But in that study, the dependent variable was the change in (relative) earnings, covering the earnings of both salary and wage earners. In this analysis there is no need to include a variable for change in non-production workers to total employment because dependent variable is the change in hourly wages, which refer to wage earners alone, and not to salary earners who are usually considered as non-production employees.

5.5.2. Long-run Industry-specific Flexibility.

The estimates of industry-specific determinants of long-run changes in nominal wages across manufacturing industries in Greece, are reported in Table 5.5.1. In this specification it is not included the strike activity variable as in such long-run changes variation in bargaining power may be captured better by levels of unionisation rather than by annual rates of strike activity. But data on unionisation levels in Greek manufacturing are not available.

The estimated coefficient for the productivity variable is found strong, positive and significant for the period 1966-88, but not for any of the sub-periods. This strong and positive coefficient for the period 1966-88, implies that the long-run change in industry nominal wages, observed from 1966 to 1988, reflects to a wide extent, vibrant changes to industry-specific productivity, and thus, a lot of the observed flexibility should be attributed to industry-specific changes in productivity.

The estimated coefficients for the variables representing longrun changes in industry employment are found insignificant for all the tested periods. Furthermore, the regression specifications without the employment variable are relatively more significant and with slightly higher coefficients of determination than the specifications which include the industry employment variables. This result implies that long-run changes in industry employment have not influenced changes in industry nominal wages and is consistent with the competitive model.
The coefficients for the initial wage level variable are found negative and significant for the period 1966-88. This result reflects the long-run tendency, over the full period 1966-88, towards a more compressed manufacturing wage structure as well as the dominance of governmental incomes policy in industry wage determination processes. Negative signs characterise the original wage level variable in the sub-period regressions for 1967-74 and 1982-88, but the estimated coefficients are not significant. Nevertheless, the estimated coefficient in the specification for 1975-81 is found significant and positive. This result implies a intermittent tendency towards a widening interindustry wage structure. This is consistent with the norm-based policy in force during that period and with the major disturbances observed during that period in the manufacturing industrial relations caused by the high frequency of decentralised strike activity and the widespread collective bargaining at the plant level.

ά.

Finally, the control variable for female employment in each industry's employment is found negative and significant in the 1977-88 period regression, that is the period for which data on the female participation in industry employment are available. The negative sign simply reflects the fact that, as female workers are, on average, paid less than men workers, the increase of their share in an industry's employment affects negatively average changes in wages of both sexes.

5.5.3. Short-run Industry-specific Flexibility.

The estimates of industry-specific determinants of short-run changes in nominal wages across manufacturing industries in Greece, are reported in Table 5.5.2. We find no positive and significant coefficients for the productivity variable. However, the estimated coefficients for changes in productivity are found negative and significant for the full period 1966-88 as well as for the sub-period 1967-74.

This negative association between short-run, namely annual, changes in nominal wages and changes in industry productivity implies that the dominant pattern over the period under consideration has not been for annual changes in wages to be following changes in industry productivity. This, in turn, may have happened because annual rates of change in industry wages either have been falling behind productivity increases, or wages have been increasing more that allowed by productivity performance. Such developments have been probably caused by the varying incomes policy in force as well as by the major changes in the manufacturing bargaining structure and processes, especially in 1975-81. Both of them insulated wages from average and industry productivity developments. From this point of view, we can infer that productivity changes were not a source of short-run industry-specific flexibility in relative wages.

The inverse results on the influence of industry-specific productivity upon industry wages, as it is found strongly

positive in the long-run and negative in the short-run, do not contradict one other. While at the end of the period under examination the overall change in industry wages reflect the long-run change in industry productivity, it appears that their course through the annual rounds of wage determination over the period 1966-88 has not been smooth and linear because of the alterations in incomes policy and equally major changes in bargaining structures and processes.

The estimated coefficients for the employment variable indicate that annual changes in employment have not been associated with wage flexibility. They are found negative and significant for the period 1966-88 and insignificant for the sub-periods. This finding is consistent with our conclusions drawn from the bivariate correlation analysis of changes in wages and change sin employment over the same period. This result implies a negative relationship between increases in nominal wages and employment growth. As in the cross-section analysis we control for productivity changes, the negative relationship between increases changes in nominal wages and industry employment growth indicates that over the period under consideration the year to year course of industrial pay may has contributed to relative loss of potential jobs by causing reduced demand for labour.

While neither short-run changes in industry productivity nor short-run changes in industry employment are found influential determinants of industry-specific wage flexibility, the estimated coefficients for the strike activity variable indicate that over the period 1966-88 the rate of industry strike activity has

caused a certain degree of short-run industry-specific wage flexibility. The estimated coefficients for the strike activity variable are positive and significant for the full period 1966-88, as well as the sub-periods 1967-74 and 1982-88, except the period 1975-81 for which the regression results are generally poor as the estimated equation itself is insignificant. The strong positive coefficient for 1967-74 refers to the effects of strike activity undertaken during the first fourth months, namely from January to April, of 1967 as only for that period strike activity was lawful. In our data set from 1968 till 1974 the industry strike activity variable has zero values. But, overall the regression results indicate that strike activity has been a source of short-run industry-specific wage flexibility.

The estimated coefficient for the initial wage level variable is found negative and significant for the indexation period 1982-88 and positive and significant for 1967-74. The coefficient for the indexation period is consistent with the type of incomes policy in force, as partial indexation was aimed to compress the wage structure. But the coefficient for the sub-period 1967-74, during which no formal incomes policy was in force, collective bargaining was typical and strike activity banned, due to the political regime of the period, indicates that the forces working for a wider wage structure dominated the year-to-year changes in the manufacturing wage structure.

Finally, the estimated coefficient of the control variable for female employment is found, as expected, negative and

significant, indicating that the increase of the female share in industry employment affects negatively the rate of change in average nominal wages of both sexes the females wage is, on average, nearly 70 per cent of the males wage.

5.5.4. Employment Consequences of the Asymmetric Response of Wages to Productivity.

Given the results suggesting the presence of industry-specific wage flexibility associated with changes in productivity in the long-run, we also test for the employment consequences of the productivity-related wage flexibility. The results from testing the hypothesis of asymmetric response of wages to productivity are given in Tables 5.5.3 and 5.5.4. Table 5.5.3 presents the results from testing the asymmetric response hypothesis in the long-run, that is on the basis of base-year to end-year changes. Table 5.5.4 presents the results from testing it in the shortrun, that is on the basis of year-to-year changes.

The results concerning long-run wage flexibility provide no support in favour of the asymmetric response hypothesis. The estimated coefficients are found insignificant in all regressions specified for the period 1966-88 and its sub-periods. The results of the regression analysis relating changes in relative wages and above- and below-average changes in relative productivity, in the short-run, provide no support of the asymmetric response either. However, the estimated coefficients are found significant in the case of the 1982-88 period, but the variable for changes

in productivity below-average is negatively signed, while the variable for changes in productivity above-average is found with positive sign. This result implies that in 1982-88, that is over the period of indexation policy, wages in the sector of belowaverage productivity changes were not responsive to this relative productivity performance and were rising without being connected to productivity developments. At the same time wages in the sector of above-average changes in productivity were responsive to their better relative productivity performance.

Thus, the test on asymmetric response suggests that the observed productivity-related flexibility was employment hindering. This is caused by the fact that wages in industrial branches with above-average changes in productivity by being more responsive to productivity changes led to relative job losses, while wages in industrial branches with below-average changes in productivity by being de-connected from their relative productivity performance saved less jobs than would have saved by following the relative productivity developments.

This de-connection of industry wages from industry productivity performance seems to be mainly caused by the indexation policy, as indexation by applying automatically across the board insulated year-to-year wage developments from differentials in productivity change. However, results similar to those regarding the sub-period 1982-88 are found for the sub-period 1977-88 too. As indexation policy was introduced in 1982, it seems that, in 1977-81, either the norm-based policies insulated wages from relative productivity performance or collective bargaining

developments had a similar effect.

As far as incomes policy is concerned over the period 1977-88, its common characteristic, despite the differences in type between norm and indexation, was their aim to narrow the wage structure by favouring the low-paid segments of the labour force. As far as collective bargaining developments are concerned, the positive and significant coefficient found for strike activity for period 1966-88 suggests that informal collective bargaining at the plant level contributed to the de-connection of changes in industry wages from changes in industry productivity changes.

To summarise, our analysis on industry-specific determinants of flexibility over the period 1966-88, suggests that in the longrun industry wages were associated with industry-specific productivity changes, while in the short-run strike activity was the main source of industry-specific wage flexibility. Furthermore, it appears that over the period 1977-88 asymmetries in the association between relative changes in productivity and wages at the industry level were employment hindering rather than employment enhancing as wages in industries of relatively poor productivity changes were insulated from this productivity performance. Incomes policy aimed to compress the wage structure seems to be the major cause of this employment hindering asymmetry.

5.6. CONCLUSIONS.

In theory flexibility in relative wages can be either competitive or non-competitive. Competitive is considered the flexibility associated with the use of relative wages for balancing labour demand and supply at the industry level, when, for instance, increased wages are used for retaining and attracting labour in expanding sectors. Non-competitive is considered the flexibility associated with industry-specific factors such as changes in productivity or bargaining power.

In Greek manufacturing over the period 1966-88, although flexibility in manufacturing relative wages was not an overwhelming characteristic, in conditions of extensive regulation of the labour market, that is in conditions which do not resemble very much to the ideal competitive labour market, non-competitive factors played a more important role than the competitive ones. The observed flexibility is mainly of the noncompetitive type as it is caused by industry-specific factors such as productivity changes and industry-specific rates of strike activity.

The lack of competitive flexibility as well as the characteristics of non-competitive flexibility are mainly due to successive incomes policies implemented in the context of permanent employment regulation and to the militancy of trade-unions at the plant level which transformed the manufacturing

bargaining structure. These developments implied that for a number of years the Greek manufacturing sector was characterised by wage rises that were directly associated with industryspecific productivity changes.

The productivity-related flexibility suggests that workers in certain sectors shared industry-specific productivity gains. The strike activity-related flexibility suggests that workers, militancy and informal collective bargaining related to industryspecific characteristics resulted too in higher relative wages. It seems that the overall regulation disfavours competitive flexibility and sets the scene for non-competitive flexibility caused by institutional forces.

While overall, there was not much scope for competitive flexibility, in 1967-74 both the bivariate correlation analysis and the background information suggest that that period can be considered as an episode of competitive flexibility in the Greek manufacturing sector. The sub-periods characterised by different types of wage policy are different as far as flexibility is concerned either. Apart from the episode of competitive flexibility in 1967-74, non-competitive flexibility dominated the relative wages scene, but only for the sub-periods 1982-88 and 1977-88 the rates of industry strike activity are associated with non-competitive flexibility. The results for the sub-period 1975-81 are poor and this may be due to the wide disturbances caused in the system of industrial relations in Greek manufacturing.

As far as the employment consequences of asymmetric responses of

relative wages to industry-specific changes in productivity are concerned, they were found employment hindering for the period 1977-88, that is the period of egalitarian incomes policies of both types, norm-based and indexation. Overall, the extensive and varying policies of wage and employment regulation as well as their repercussions in the area of industrial relations are behind the observed non-competitive flexibility in relative wages and the asymmetries in the wage-productivity-employment nexus.

But the observed variability in relative wages should not be attributed only to these industry-specific factors. Incomes policies themselves may account to a certain extent for changes in relative wages as they treated them in varied ways.

Table 5.4.1

	<u>BIVA</u> <u>NOMINA</u> (T	<u>RIATE CORRELA</u> <u>L WAGES AND</u> ime periods:	<u>FION COEFFI</u> <u>Changes in</u> 1966-88, 19	CIENTS RELATING EMPLOYMENT OF 67-74, 1975-81,	<u>CHANGES IN</u> <u>WAGE EARNERS</u> 1982-88).		
		Full Period:	<u>1966-1988.</u>	<u>Period:</u>	<u>1967-1974.</u>		
Cha	anges of						
1	year	65/66-87/88 n=460	09**	66/67-73/74 n=160	.16**		
2	years	65/67-86/88 n=440	09**	65/67-72/74 n=160	.11		
3	years	65/68-85/88 n=420	16***	65/68-71/74 n=140	.04		
5	years	65/70-83/88 n=380	32***				
10	years	65/75-78/88 n=280	43***				
23	years	66/88 n=20	15				
Period 1975-1981. Period 1982-1988. Changes							
1	year	74/75-80/8 n=140	1.002	81/82-87/88 n=140	01		
2	years	73/75-79/8 n=140	1.12	80/82 - 86/88 n=140	.16*		
3	years	72/75-78/8 n=140	1.03	79/82-85/88 n=140	.13		
5	years	70/75-76/8 n=140	126***	78/82-83/88 n=140	04		
10	years	65/75-71/8 n=140	127***	72/82-78/88 n=140	.21***		
No	 te: L	evels of Sign	ificance:	* = 0.10, ** =	0.05, *** = 0.01.		

Table 5.4.2

BIVA	RIATE COR	RELATION	COEFFICIE	NTS RELAT	ING CHANGE	<u>B IN NOMIN</u>	AL WAGES
	AN	<u> CHANGE</u>	8 IN EMPLO	<u>YMENT</u> OF	WAGE EARNE	RS.	
	(Short-ru	n and Lo	ong-run chai	nges over	the perio	od 1966-88)	
	Dowind					• • • • • • • • • • • • •	,
	l voar		ige:	5 voare	10 years	23 voare	
Voar	i year	z year	.5 5 years	5 years	IU YEAIS	25 years	
							• • • • •
/66	.45**						
/67	.20	.23					
, /68	12	.12	.13				
, /69	.28	02	07				
/70	08	08	29	16			
/71	.24	.21	.20	17			
/72	.36	.37	.13	07			
/73	01	.07	.17	.35			
/74	13	37	32	.02			
/75	12	02	41*	19	21		
/76	24	09,,	16	28	35		
/77	.28	.48^^	.34	.11	04		
/78	03	16	.17	.01	01		
/79	60	51	42	07	12		
/80	.12	.17	.03	.30	09		
/81	.21	.40	.31	.02	22		
/82	.43	.28	.49	.35	15		
/83	33	.01	01	.0002	.11		
/84	.48	10	.14	•11	.09		
/00	.001	.45	UO =**	• 1 /	.15		
/00	13	.10	. 53	.21	10		
/82	.10	16	.00	07	.05	- 15	
/ 00	• • • •	• ±0	• ± ∪	.JJ 	• 17	-•IJ 	

Notes: N = 20 (Two-digit industrial branches of manufacturing). Levels of Significance: * = 0.10, ** = 0.05, *** = 0.01.

Table 5.4.3

BIVARIATE CORRELATION COEFFICIENTS RELATING ANNUAL CHANGES IN NOMINAL WAGES AND ANNUAL CHANGES IN EMPLOYMENT OF WAGE EARNERS. (comparison between coefficients for all branches and coefficients for expanding branches)

Year CHWEi CHWEi>O _____ •45^{**} .25 /66 .18 /67 -.28 .39* /68 -.12 /69 .28 /70 -.08 -.05 .26 .30 .24 /71 .36 /72 .006 /73 -.01 /74 -.13 -.13 /75 -.12 -.01 /76 -.24 -.22 .28 .16 /77 -.03 -.60*** /78 -.37 /79 -.41 /80 .12 -.22 .18 .21 /81 .27 /82 .43 -.37 .41* -.33 .48** /83 /84 .00 .03 /85 -.13 -.04 /86 /87 .10 .08 /88 .08 -.07 ----Notes:

Period of Change: 1 year

N = 20 (Two-digit industrial branches of manufacturing). Levels of Significance: * = 0.10, ** = 0.05, *** = 0.01.

.

ESTIMATES OF THE INDUSTRY SPECIFIC DETERMINANTS OF NOMINAL WAGE CHANGES ACROSS MANUFACTURING INDUSTRIES IN GREECE. (base-year to end-year changes)											
Dependent Variable: CHWi											
Period	СТ	CHWEI C	HPRi	WLi	CHFEI R^2/R^2	F-value					
1966 /88	4,688.68 (11.89)	.07 (.07)	2.29 ^{**} (2.11)	-82.89** (-2.67)	* .45 .35	4.36**					
n=20	4,691.77 (12.36)		2.27 ^{**} (2.27)	-82.84 ^{*;} (-2.75)	* .45 .39	6.95**					
1967 /74	126.43 (4.45)	.03 (.22)	.16 (1.06)	-1.20 (53)	.08 10	.45					
n=20	125.49 (4.59)		.14 (1.12)	-1.01 (50)	.08 03	.69					
1975 /81	212.09 (7.27)	.38 (1.15)	.14 (.94)	1.84 ^{*;} (2.35)	* .29 .16	2.22					
n=20	212.85 (7.23)		.06 (.42)	1.81 ^{**} (2.28)	* .24 .15	2.62**					
1982 /88	192.08 (10.20)	07 (49)	005 (03)	17 (-1.66)	37 .22 (52) .01	1.03					
n=20	191.07 (10.46)		.03 (.23)	16 (-1.67)	41 .20 (59) .01	1.36					
1977 /88	885.81 (15.82)	03 (06)	30 (83)	-2.73 [*] (-2.82)	* -4.64 [*] .61 (-2.00) .50	5.77**					
n=20	885.61 (16.36)		29 (89)	-2.72 [*] (-2.93)	** -4.63 [*] .61 (-2.07).53	8.20***					
Notes:	CHWi = Chan CT = Cons CHWEi= Chan CHPRi= Chan WLi = Init CHFEi= Chan	ge in No tant Ten ge in En ge in Pr ial Leve ge in th	ominal H cm. nploymen coductiv el of Wa ne Share	ourly Wag it of Wag ity in B ge in Br of Fema	ge Paid in Br e Earners in ranch i. anch i. le Employment	anch i. Branch i. in Branch i.					
Levels of Significance: * = 0.10, ** = 0.05, *** = 0.01. t-statistics in parentheses.											

· · ·

<u>estima</u>	<u>TES OF</u> <u>CHANG</u>	THE INDU	<u>ISTRY SPI</u> <u>B MANUFA(</u>	ECIFIC DE CTURING I	TERMIN NDUSTRI	<u>ANTS OF</u> [ES IN GR	NOMIN REECE	IAL WAGE
			(year to	year ch	anges) 			
Depend	ent Var	iable: Cl	łWi					
Period	СТ	CHWEi	CHPRi	SII	WLi	CHFEI	R ²	F-value
1966 -88	17.74 (29.80)	13 ^{**} (-2.88)	**12** (-3.02)	** .20 ^{**} (5.38)	*000		.09	11.55***
n=460		. ,						
1967 -74	-4.34 (-2.58)	09 (-1.55)	14 ^{**} (-2.94)	** 10.45* (2.37)	** .89 [*] (10.15)	***	.44	30.62***
n=160								
1975 -81	23.53	.03 (0.59)	.02	005	.02		.02	.55
n=140	(,	()	(,	()	(/			
1982 -88 n=140	35.51 (20.68)	05 (96)	.06 (1.16)	.18 ^{**} (3.73)(*05 [*] -10.84)	***15 [*]) (-2.38)	**.53	29.93***
1977 -88 n=240	27.71 (35.51)	06 (-1.34)	05 (-1.12)	.05* (1.74)(-	03** 10.50)	**18** (-3.21)	**.35	24.88***
Notes:	 CHWi =	Change i	n Nomina	 l Hourly	Wage Pa			 i.
noces.	CT = CHWEi= CHPRi= SIi = WLi = CHFEi=	Constant Change in Change in Index of Initial Change in	Term. n Employn n Produc Strike Level of n the Sha	ment of W tivity in Activity Wage in are of Fe	age Ea Branch Branch Branch male En	rners in n i. nch i. i. mployment	Brand : in	ch i. Branch i.
	Levels t-stati	of Signi Istics in	ficance: parenth	* = 0.1 eses.	.0, ** =	= 0.05, *	*** =	0.01.

.

ESTIMATES OF THE RELATIONSHIP BETWEEN RELATIVE NOMINAL WAGE CHANGES AND RELATIVE PRODUCTIVITY CHANGE (base-year to end-year change)									
Dependent	Variable:	CHRWi							
Period	СТ	CHPRAi	CHPRBi	WLi	CHFEi	R^2	F-value		
1966/88 n=20	1,119.39 ^{**} (2.19)	1.35 (0.53)	3.14 (1.28)	-89.38** (-2.54)	k	.45	4.45**		
1967/74 n=20	24.27 (0.81)	27 (49)	.40 (1.57)	-1.41 (-0.69)		.15	.92		
1975/81 n=20	-63.26 ^{**} (2.10)	.12 (.49)	04 (11)	1.81 ^{**} (2.23)	*	.24	1.69		
1982/88 n=20	34.43 (1.61)	.05 (.16)	.02 (.08)	16 (1.58)	41 (55)	.20	.96		
1977/88 n=20	185.10 ^{**} (3.33)(*84 -1.91) (60 (-1.01)	-2.99* (-3.37	**-3.63)(-1.66)	.67	7.72***		
<pre>Notes:CHRWi = Change in Relative Wage Paid in Branch i. CT = Constant Term. CHPRAi= Change in Productivity in Branches with Above Average Increase in Productivity. CHPRBi= Change in Productivity in Branches with Below Average Increase in Productivity. WLi = Initial Level of Wage in Branch i. CHFEi= Change in the Share of Female Employment in Branch i. Levels of Significance: * = 0.10 ** = 0.05 *** = 0.01</pre>									
Lev t-s	Levels of Significance: * = 0.10, ** = 0.05, *** = 0.01. t-statistics in parentheses.								

<u>ESTIMATES OF THE RELATIONSHIP BETWEEN RELATIVE NOMINAL WAGE CHANGES</u> <u>AND RELATIVE PRODUCTIVITY CHANGE</u> (year to year change)									
Dependent V	ariable: CH	 RWi			, ap au ao ao ao ao a				
Period CT	CHPRA	i CHPRBi	WLi	CHFEI	R ²	F-value			
1966-88 - n=460 (-	.17 .018 .49) (0.52)	020 (06)	.0004 (.33)		.0001	.17			
1967-74 - n=160 (-	.83049 .60) (59)	.060 (.85)	.046 (.70)		.009	. 47			
1975-81 n=140 (.84023 .88) (40)	.059 (.99)	005 (57)		.009	.39			
1982-88 n=140 (.11 .12 ^{**} .09) (2.22)	*15*** (-2.41)	*003 (96)	11 ^{***} (-2.61)	.099	3.70***			
1977-88 - n=240 (-	.12 .09 [*] .22) (2.09)	*11** (-2.31)	002 (-1.04)	11 ^{***} (-2.95)	.059	3.67***			
Notes:CHRWi CT CHPRA CHPRB WLi CHFEi Level	= Change in = Constant i= Change Average = Change = Initial = Change in s of Signif	n Relative Term. in Product Increase : Increase : Level of N n the Shar icance:	e Wage Pa ctivity : in Produc ctivity : in Produc Wage in B re of Fem * = 0.10,	id in Bran tivity. in Bran tivity. ranch i. ale Empl ** = 0.	ches w ches w ches w loyment	<pre>. vith Above vith Below in Branch * = 0.01.</pre>	i.		

CHAPTER 6. INCOMES POLICY AND RELATIVE WAGES IN GREEK MANUFACTURING.

6.1. INTRODUCTION AND SUMMARY.

Since 1975 there has been no period without formal incomes policy in force. The policies undertaken since then fall into two categories. First, were the norm-based policies of the 1975-81 period. Second, policies based on systems of indexation were in operation throughout the period 1982-88. These policies were undertaken in the context of changing bargaining structures and processes in manufacturing. As argued in Chapter 5 these developments are associated with non-competitive flexibility in relative wages.

Given the high frequency of strikes and the rise of informal collective bargaining at the plant-level, this chapter examines the level and the direction of influence exerted by the normbased and the indexation-based incomes policy upon the interin the Greek manufacturing sector industry wage structure over the period 1975-88. The period 1967-74, during which no explicit incomes policy was enforced, is adopted as benchmark for the comparison and the evaluation of the formal incomes policies' effects upon the inter-industry wage structure. It also examines the level and the direction of economy-wide influences, namely those by inflation and unemployment.

The inter-industry wage structure in Greek manufacturing is described by the coefficient of variation of average hourly pay of male workers across twenty two-digit branches. In

other words, the coefficient of variation and its variability over time are used as summary indicators representing the relative position of the average male worker in each of the twenty industrial branches with respect to hourly wage receipts.

Section 6.2 deals with the theoretical issue of how incomes policies may, if successful, narrow the wage structure. And, then, discusses the set of factors, apart from incomes policy, which may also account for both formation and change over time of the manufacturing wage structure. Section 6.3 elaborates and clarifies the hypotheses concerning changes in relative wages which are to be tested in this chapter. In Section 6.4 the equations' specification for testing on the leverage of incomes policies upon the inter-industry wage structure is presented and the relevant regression results are discussed. Finally, Section 6.5 draws the conclusions of the analysis.

Our analysis suggests the norm-based and the indexation policies are both associated with the widening of the manufacturing wage structure compared to the benchmark period 1967-74. However, the indexation policy widened the wage structure half as the normbased policy. Furthermore, changes in inflation and unemployment are associated with variation in the wage structure in a rather unconventional way. Higher inflation and lower unemployment lead to increased inter-industry wage differentials. Strike activity is found to vary inversely to the coefficient of variation in manufacturing wages. Therefore it cannot be considered as an

indicator of bargaining power. The common cause lying behind these results is the rise and the workings of the informal system of collective bargaining at the plant level and the behaviour of the high-wage and high-bargaining power industrial branches which benefited the most from the decentralisation and the fragmentation of collective bargaining in manufacturing.

6.2. WAGE STRUCTURE, INCOMES POLICY AND BUSINESS CYCLE.

6.2.1. Incomes Policy Rationales.

The use of incomes policy as a tool of macroeconomic management for controlling wage inflation, is of interest to both economic and industrial relations' analysis. Artis (1981) drawing on an analytical as well as historical perspective summarised the main rationales underlying the incomes policies which were adopted during the post-war period. Flanagan, Soskice and Ulman (1983) provided an extensive review of international experiences with incomes policy over the same period. These analyses made clear two points.

First, rationales of incomes policy are influenced by political calculus. Incomes policy design, implementation and effectiveness are, usually, a matter of economic as well as political debate. Under conditions of wage determination through collective bargaining, incomes policy aiming to control wages can be viewed as attempt to control collective bargaining either. The political dimension of incomes policy in conditions of collective bargaining was stressed by the keynesian school and its founder himself. The wage-raising power of unions was perceived as "a political rather than economic problem".

As Brawn (1975) pointed out, incomes policies in the post-war period, when they were not part of corporatist settlements and social contracts, were considered as direct offensive to trade-

unions and free collective bargaining and from this point of view incomes policies have been strongly criticised as unable to cope with wage inflation in an environment of collective bargaining and as causing more problems in the system of industrial relations than they resolve.

According to Clarke (1987:45), in the 1970s incomes policies "even with positive support from trade-union centres frequently had no more than a short term moderating effect, which might be followed by a surge of wage increases when the policy was relaxed". Tarling and Wilkinson (1982), drawing on the U.K. experience with incomes policies in the 1970s, concluded that those policies created anomalies in the general process of earnings changes which in turn led to significant instabilities and disturbances once policies ended.

Second, the views of both policy-makers and economists about the rationales of, and the roles for, incomes policy are closely related to the evolution of theorising about wage determination and wage inflation. Varying theories on wage inflation imply different rationales for the role of incomes policy in macroeconomic management. In an extreme case, the monetarist account of inflation implies no room for incomes policy as a tool for controlling wage inflation. By contrast, in the set of costpush theories of inflation, incomes policy should be adopted in order to control wage costs and, thus, minimise inflationary factors at their origin. In this framework, the issue of relative wages is crucial for controlling wage inflation.

Theories on wage determination in a market environment, as summarised by Artis (1981), fall into four categories: the wage theorem view, the Phillips curve, the augmented Phillips curve and the real wage resistance model. The issue of relative wages is not in the centre of interest in theories dealing with wage determination at the aggregate level. Nevertheless, it is an important issue for theories based upon the wage theorem view. According to the wage theorem view, in its hey-day, "the maintenance of full employment removes the restraint upon tradeunions to press for higher wages ... and inter-union rivalry makes it impossible for the lesson that higher wages mean higher prices to be learnt; rather it encourages an inflationary spiral. (group) money wage rates depend heavily Individual on relativities, with the result that the absolute (average) level of money wages (and prices) is essentially indeterminate" (Artis, 1981:7).

6.2.2. Incomes Policy and Wage Structure.

The analysis of wage determination at the disaggregated level leads to the question whether or not there are any leaders and followers, that is groups of wage earners who set the pace in wage bargaining processes which is then followed by other groups. According to Blackaby (1971:44) leaders in wage bargaining may include: highly capital intensive industries where productivity rises rapidly; highly unionised and militant industries with strong trade-unions; industries employing workers with skills in

scarcity; workers with great importance to export industries; and workers in industries which tend to get left behind in the medium term. As Blackaby notes "it is important to distinguish between the influence of high pay and the influence of the high pay bargain struck on a particular occasion". What mainly matters with regard to wage inflation and its association with developments in the wage structure, is the influence of high pay.

From this point of view, incomes policy should be focusing upon wage determination in industries leading the wage inflation process. These industries, in the absence of such a policy, would achieve higher wage increases. In turn, these increases, could create, through pay comparability and partial spillover effects, accelerating wage inflation. Earlier attempts, as by Sargan (1971), to test for the existence of such relationships confronted problems since wage statistics on which the analysis had to be based did not match with negotiating groups. The use of the coefficient of variation as a proxy of variations in the wage structure helps to overcome the problem of matching bargaining groups and wages behaviour.

In testing for the incomes policy's effects upon relative wages and the resulting wage structure the assumption is made that the labour market is divided in sectors with specific characteristics as far as wage determination and the sectors position in the wage structure are concerned. Wachter (1974) distinguishes between the competitive and the non-competitive sector, the latter being the large oligopolistic unionised industries. These unionised industries are considered as the home of cost-push inflation.

Metcalf (1977), in the same line of argument, distinguishes between sectors covered and uncovered by collective bargaining. In this case the covered sector behaves as the non-competitive sector in Wachter's definition. The origins of these divisions can be traced in earlier attempts to analyse the wage inflation at a disaggregated level. Bowen (1960:74-81), for instance, divided the industries into two groups. First, a market power sector consisting of industries that are both highly concentrated and highly unionised. Second, a competitive sector in which the industries have the reverse characteristics.

On the basis of the distinction between competitive or noncovered sector and non-competitive or covered sectors, given the positive correlation between non-competitive or covered industries from the one side, and wages from the other side, both Wachter and Metcalf assume that, in the case of cost-push inflation, a successful incomes policy should stabilise or reduce the width of the wage structure. Certainly, the assumption that a successful incomes policy narrows the wage structure by setting restrictions on the rate of wage increases in the highwage non-competitive sector holds true when the pressure upon wages comes mainly from the high-wage highly unionised and covered by collective agreements industries.

But a different, although temporary, effect upon the wage structure may be caused when the pressure comes from the lowwage industries. In such a case, low wage industries may contribute in narrowing, temporarily, the wage structure.

Nevertheless, if pay comparability is a principle upon which wage claims are formed, any improvement in the relative pay of the low-wage industries may cause the high-wage industries to react in order to re-establish their relative position in the wage structure. But, typically, wage inflation is not created by lowwage pushers which oblige high-wage industries to react. This might be only a temporary case which has rather to do with the second side of the distinction made by Blackaby, that is the influence of the high pay bargain struck in a particular occasion. In any case, a successful incomes policy should be able to control wage inflation caused by wage spirals initiated by either the high-wage or the low-wage industries and, at least, not allow the wage structure to become wider.

6.2.3. Business Cycle and Wage Structure.

Incomes policy is not the only determinant of variation in the wage structure. Whatever the specific rationale of an incomes policy is with respect to the wage structure, the latter is also influenced by structural and cyclical changes in the economy. Factors which may account for the inter-industry wage differentials include product market conditions, conditions related to capital and technology, and human capital characteristics of the industry's labour force. Moreover, include labour conditions and industrial relations market characteristics such as unionisation, militancy and bargaining power.

Wachter (1970) provided a view on factors determining cyclical variation in relative wages. According to his approach, cyclical variation in relative wages can be analysed in terms of in industry labour supply curves. These shifts in shifts industry labour supply are caused by changes in aggregate unemployment. In his model, the labour market is divided into two sectors, the high-wage non-competitive sector and the low-wage competitive sector. Each sector is assigned a sector-specific labour supply curve. If the labour market tightens, low-wage industries have to improve their pay in order to maintain а certain level of labour supply. If unemployment rises, lowwage industries may adjust downwards their relative wage.

High-wage industries, in turn, adopt wage policies that permit them to maintain their relative position in the wage structure and they may, according to Wachter, allow their relative wages vary inversely with the business cycle. This inverse variation is due to the trade-unions' interest in wage developments within high-wage groups rather than outside them. It is also due to their interest for a smoother variation in wages than in line with the business cycle variation. It can be also explained by the smoothing effect that the periodic collective bargaining of contracts has upon cyclical variations. This rationale led Wachter to the, then corroborated, hypothesis that the size of the wage differentials is a positive function of the level of unemployment, with the variable of unemployment representing excess demand conditions in the labour market.

Moreover, Wachter found that a cost of living variable may also

affect variations in the wage structure, as industries differ speed with which they adjust wages to changes in in the consumer prices. In the context of the U.S. wage formation processes, with three-year long contracts, wage rounds and institutional lags in wage determination, high-wage industries rapidly during inflation responded more slow and less rapidly during rapid inflation, than low-wage industries.

Ξ.

According to the model proposed and tested by Wachter, it appears that there is a negative association between inflation and the wage structure and a positive association between unemployment and the wage structure, so that a decrease in unemployment and an increase in inflation may both act to narrow the wage structure over the business cycle. Metcalf (1977) arrived to similar results with respect to the influence of unemployment and inflation upon variation in the U.K. wage structure.

Apart from the incomes policy influence, three other sets of factors may account for formation and evolution of the manufacturing wage structure over time. The first set refers to differences in the competitive structure in the product markets, which determine differences in capital and productivity. The second refers to human capital characteristics which also imply differences in labour productivity. The third set refers to the business cycle influence upon the manufacturing wage structure. To summarise, on the business cycle influence upon variations in the wage structure, there is evidence concerning the variation in

the U.S. and the U.K. wage structures that unemployment is positively and inflation is negatively associated with it.

•

6.3. HYPOTHESES TO TEST AND THEIR RATIONALES.

6.3.1. High-wage and Low-wage Sectors.

Despite the long-lasting characteristics of undeveloped formal institutions and processes of collective bargaining, in the context of strike activity at the plant level and informal collective bargaining, the model of high-wage highly concentrated highly unionised and low-wage less concentrated less unionised industries is substantiated and becomes relevant to developments in the Greek manufacturing sector.

As far as relative wages in the twenty two-digit industrial branches are concerned the coefficient of concordance in the ranking of the manufacturing wage structure for male workers in the period 1965-1988, presented in Table 4.3.4, is 0.82. This value means that male employees of certain high-wage industrial branches in Greek manufacturing maintained, throughout this period, their higher position in the wage structure. The composition of the top five high-wage industrial branches is very stable and the club of the bottom-five low-wage industrial branches has also stable composition and permanent members.

This stability of the wage structure hierarchy over the period 1965-88 and the central role of the size variable in the determination of the manufacturing wage structure, indicated in Table 4.3.5, implies that the model proposed and tested by Wachter and Metcalf may capture the essential workings of the



STRIKE ACTIVITY RANKS

INDUSTRIES IN WAGE & STRIKE STRUCTURE



Note: Numbers from 20 to 39 refer to keys for manufacturing industries.

Greek labour market with respect to wage determination in the manufacturing sector. However, while the high-wage and low-wage sectors are clearly distinguishable, the evidence on the ranking of industrial branches according to their strike-proneness may not reduce the validity of the model. In the case of the highwage and low-wage industrial branches in Greek manufacturing the combination of ranking of the twenty industrial branches according to wages and strike activity, as presented in Figure 6.3.1 indicates that some high-wage industrial branches hold also high ranks in the strike activity hierarchy while other high-wage branches hold relatively low ranks with respect to strike activity. The same mixed picture is also observed in the case of the low-wage industrial branches. Some of them are rather passive while others hold high ranks in the strike activity hierarchy.

Given the theoretical and empirical considerations discussed in Section 6.2 and in earlier chapters respectively, it seems reasonable to assume that country- and period- specific characteristics are crucial for the analysis of variation in the Greek manufacturing wage structure. They may bring about alterations in the explanation of rationales with regard to determination of relative wages. These country and period specific characteristics are taken into account in formulating the testable hypotheses of this chapter. These hypotheses refer to three sets of influence exerted upon the Greek manufacturing wage structure over the years 1967-88, that is the influence of incomes policy, business cycle and strike activity respectively.

6.3.2. The Influence of Incomes Policy.

Our hypotheses concerning the influence of incomes policies are slightly modified when compared with those in earlier work by Wachter and Metcalf. This modification is due to the fact that in the U.S. and the U.K. case there were both incomes policy-on and incomes policy-off periods in conditions of wage determination under collective bargaining. In the Greek case, there have been incomes policy-on periods with collective bargaining emerging in parallel. Thus, in this chapter different periods of incomes policy are compared with a period during which neither incomes policy nor essential collective bargaining were at work.

During the period 1967-74, with essential collective bargaining suspended, strike activity banned and statutory regulation of wages by the government, the ability of high-wage high-bargaining power industries to maintain and to increase their wage premium was taken away. This may have created relatively better conditions for the low-wage less powerful industries which faced no severe change with respect to wage determination conditions.

In the period 1975-81, with the re-establishment of the right of collective bargaining and the adoption of norm-based incomes policies the wage structure is expected to become wider than that of the period 1967-74, for the following reasons. First, the norm-based incomes policy may have failed to provide a low level of uncertainty with respect to inflationary erosion of wages. The policy's pattern was to award a substantial increase in the beginning of the year and a less important increase in its second

semester regardless of any accelerating rises in consumer prices. Thus, high-wage high-bargaining power industries may have been pre-active and aimed at higher increases to avoid erosion of their real wage. In the same time, low-wage low-bargaining power industries may have been reactive and failed to keep up with rising inflation.

Second, the rise of the informal system of collective bargaining at the plant level in the Greek manufacturing industries may have enabled the high-wage highly-unionised industries to recover, reestablish and increase their wage premium, by initiating a bargaining behaviour which considered the norm as the floor for additional increases in nominal wages. In the same time, low-wage low-bargaining power industries may have had no alternative but to follow the norm. The evidence on the wide dispersion of the strike activity across the manufacturing sector given in Tables 3.4.7-3.4.10, along with the existence of a clearly identified group of high-wage highly active, in terms of strike activity behaviour, industries, as denoted by Figure 6.3.1, suggest that a wage premium may have been extracted by the high-wage highbargaining power industries via the informal system of collective bargaining.

In the period 1982-88, the indexation policy is expected to cause the wage structure to narrow, in comparison to that of the norm-based policy period, for three reasons. First, the system of indexation was securing an automatic adjustment of wages to consumer prices increases, and thus minimised the level of

uncertainty with respect to wage determination in conditions of double-digit inflation. In other words, this policy permitted, by design, to both high-wage and low-wage industries to adjust regularly and with the same speed their wage rates. Thus, the automatic adjustment of wages would stabilise the wage structure. Second, the system of partial indexation was clearly in favour of the low-wage industries and it seems reasonable to assume that, throughout the period 1982-88, the system of partial indexation was compressing the relative position of the high-wage industries.

Third, this period was also a period of high unemployment and job insecurity for a large number of ailing manufacturing industries employing nearly 10 per cent of manufacturing employees. These developments may have affected negatively the bargaining power of the unionised labour force. Presumably, the high-wage highbargaining power industries, which were exerting their power to extract and maintain a wage premium, saw it being eroded. In conditions of higher unemployment, the low-wage and less powerful industries, perhaps, found themselves in a relatively better position as the level of unemployment did not erode their bargaining power which they never dispossessed and which never was a main determinant of their relative position. In addition, it may have been the case that the system of indexation was offering to the low-wage low-bargaining power industries more than they could get by any means of collective action or by the competitive workings of the loose labour market. Thus, with the low-wage industries in the greenhouse and the high-wage

industries exposed to market and governmental pressures, the wage structure is expected to narrow.

However, the indexation policy is expected to experience a wage structure wider than that of the benchmark period. During the period of indexation the means for collective action and collective bargaining were available. And if used, as the recorded strike activity partly indicates, they may have affected positively the variation in the wage structure. In the benchmark period, collective bargaining was suspended and any means for collective action were taken away from the high-wage highbargaining power industries. From this point of view, during the indexation policy, at least during the sub-periods when it was not made compulsory by law, the high-wage high-bargaining power industries dispossessed collective bargaining as a means for extracting a, more or less curtailed, wage premium.

To summarise, the hypotheses with respect to incomes policy are that both the norm-based and the indexation policies will both widen the variation in the wage structure, with the norm-based policy exerting a relatively stronger influence than the indexation policy.

6.3.3. The Influence of Business Cycle.

Can the assumptions made by Wachter and its model be applied in the case of the Greek labour market with respect to wage formation and the evolution of the manufacturing wage structure throughout the period 1967-88? Does the fluctuation
of the coefficient of variation of average hourly pay for male workers across twenty two-digit industrial branches, reflect influences exerted by unemployment and inflation? Is their leverage expected to be positive or negative?

Our hypotheses concerning the influence of the business cycle are also affected by the way the system of wage determination has evolved in the Greek manufacturing sector during the last two decades. Contrary to the evidence provided by Wachter and Metcalf, in the context of the informal system of collective bargaining in Greece we contend that business cycle factors work in an unconventional way as pay of the high-wage and highbargaining power branches reflect more inflation and unemployment influences than pay of the low-wage low-bargaining power branches.

In the context of the dual system of wage determination, which includes both formal and informal collective bargaining, it seems reasonable to hypothesise, contrary to Wachter's and Metcalf's evidence, that rising consumer prices are positively associated with variation in the wage structure. This means that an increase in inflation causes the wage structure to become wider. This may have happened if employees in the less powerful industrial branches, failed to compensate themselves for inflation, when employees in industrial branches with bargaining power were able to keep up with inflation, to get higher increases in nominal wages and, thus, secure a higher wage premium.

Obviously, this case is quite different from that of the U.S. and the U.K. where the existence of formal annual or two to three year collective contracts causes a certain stickiness in wage changes of the high-wage and highly unionised industrial branches. In the case of the Greek manufacturing sector, if the informal and erratic mechanism for wage determination was in operation, the outcome may have been that rising inflation caused the wage structure to widen rather than to narrow. It is likely that rising inflation led high-wage and high-bargaining power to widen the wage structure via the workings of the informal system of collective bargaining.

In the same context, contrary to Wachter's and Metcalf's conclusions, unemployment may have been negatively associated with variations in the manufacturing wage structure. In other words, an increase in aggregate unemployment causes the wage structure to narrow, while a decrease causes a wider wage structure. This may have been the case when, in the context of informal collective bargaining, rising unemployment mainly affects adversely the employees in the high-wage high-bargaining power industrial branches, rather than the low-wage lowbargaining power branches. Likewise, a tighter labour market reinforces more the bargaining power of the high-wage highbargaining power than that of the low-wage low-bargaining power industrial branches.

In both cases, a negative association between unemployment and the wage structure may be observed, as with rising unemployment

the low-wage and less powerful industrial branches become in a relatively worse bargaining position, while with tightening labour markets the high-wage high-bargaining power industrial branches become in a relatively better bargaining position. If unemployment affects in such a way the relative bargaining power of high-wage and low-wage industrial branches, then a negative association between the level of unemployment and the wage structure may be observed.

Contrary to this hypothesis, it was argued by Wachter that rising unemployment permits low-wage industrial branches to adjust downwards their relative wages, while high-wage industrial branches become under for rising less pressure their relative wages. But, even if rising unemployment undermines to a unions' bargaining power, high-wage industrial certain extent branches cannot dispense with institutional pressure from the trade-unions side, which aim to preserve their relative position in the wage structure. In this case, according to the rationale proposed by Wachter and corroborated by himself and Metcalf, rising unemployment may cause the wage structure to become wider. The difference in the rationale of our hypothesis is caused by our postulation concerning the system of industrial relations and wage determination.

In the U.S. and the U.K. case, rising unemployment is considered as undermining the relative position of employees in the low-wage industries, while institutionalised collective bargaining in high-wage industries causes a stickiness in cyclical variation of wages. In the Greek case, we assume that rising unemployment

undermines the relative position of employees in high-wage industries, because of the lack of institutionalised collective bargaining at the plant level. Thus, rising unemployment makes the informal system of collective bargaining weaker and less effective. In other words, rising unemployment, while in the U.S. and the U.K. does not undermine formal collective bargaining arrangements, in Greece may have undermined the very existence of the system of collective bargaining at the plant level, that is the informal one.

Finally, another matter for investigation with regard to the business cycle influence upon the wage structure, is whether or not this influence is exerted by the actual or the past values of inflation and unemployment, that is with or without a time-lag. For instance, in the context of institutionalised bargaining with contracts lasting for two to three years, as in the U.S. inflation may exert its influence with a relevant lag. In the context of informal and erratic bargaining procedures, the influence of inflation may be caused by its actual rates. In the same context the analogy can be drawn concerning the actual or lagged influence of unemployment.

To summarise, the hypotheses with regard to the business cycle influence upon the wage structure are that prices inflation causes the wage structure to become wider, while unemployment affects it negatively.

6.3.4. The Influence of Strike Activity.

Our hypothesis about the influence of strike activity upon variation in the wage structure is rather ambiguous. Prima facie, strike activity may be considered as being positively associated with changes in the wage structure. However, it is equally reasonable to assume that it may be, also, negatively associated with variations in the wage structure. It all depends on how the role of strike activity is perceived with respect to collective bargaining and wage determination.

If strike activity is considered as indicator of bargaining power, then, it is quite reasonable to expect a positive relationship between strike activity and the wage structure. But it is not definite that strike activity denotes bargaining power. Industrial branches with high bargaining power, which is established and recognised in a certain system of industrial relations, do not need to recourse to industrial action for achieving their aims. Inversely, industrial branches with low bargaining power can be more strike-prone than high-bargaining power branches, although less effective. In other words, strike activity is only a part of the bargaining process and not an indicator of the balance of power within it.

In the context of the system of industrial relations in the Greek manufacturing sector as developed since 1975, strike activity at the plant level indicates that an informal system of collective bargaining was initiated. Indeed, strike activity may have offered the only way via which employees were able to initiate

this informal system. To build an informal bargaining structure, the establishment of a new bargaining relationship is required. Strike activity, by demonstrating a certain level of bargaining power, offers the means through which such a new collective bargaining relationship can be set up. Furthermore, employees may have to recourse to strike activity periodically in order to preserve this new relationship, if the latter is not formal and institutionalised. Perhaps, industrial branches of low bargaining power, in their effort to initiate and preserve bargaining relationships of this kind, may have had to recourse to industrial action more often than high- bargaining power branches.

In the Greek manufacturing sector over the years 1975-88, as discussed in section 6.3.1 and described in Figure 6.3.1, the coexistence of all four combinations of industrial branches ranked according to their high/low wages and their high/low strike activity records, underlines the question which lies behind this ambiguity about the influence that strike activity exerts upon the wage structure. This is the question of the character and the role of the strike activity in Greece, and especially in the manufacturing sector. The same question is also raised by statistics which indicate that during the hey-days of strike activity, in the period 1976-81, only 6 per cent of strikers where reported as being fully or partially successful (Ioannou, 1989:102).

After all, does strike activity play a central role in wage

determination processes across the board, or reflects only, or mainly, discontent and adversarial industrial relations? In such a case the possibility of a negative association of the strike activity with variation in the wage structure should not be discarded. If variation in strike activity indicates variation in collective bargaining power a positive association should be expected. But as bargaining power, usually does not vary to the extent that strike activity does, the rationale on the strike activity as indicating occasionally either discontent and low bargaining power or high bargaining power may capture better the role of strike activity in the Greek manufacturing sector. By testing on the relationship between the variation in the wage structure and strike activity, this question is to be addressed to a certain extent.

To summarise, the hypothesis concerning the influence of strike activity remains open. There are good reasons for not to consider strike activity as a good proxy for collective bargaining power. This is due to the ambiguity on the character and the role of strike activity in the system of wage determination in the Greek manufacturing sector.

6.4. EQUATIONS SPECIFICATION AND REGRESSION RESULTS.

On the basis of the analysis developed in earlier Sections, variation in the wage structure is considered as a function of five sets of factors, which include i) incomes policy, ii) the business cycle, iii) product market characteristics, iv) human capital characteristics and v) developments in industrial relations.

From these sets of factors, we have not elaborated the influence of both product market and human capital characteristics. Unavailability of the relevant data do not allow to test for their influence on variation in the wage structure. Thus, with respect to product market conditions the assumption is made that over the period under examination any differences among the manufacturing industrial branches in the competitive structure of their product markets remained constant. In other words, we assume that there was no variation in the degree of their monopolistic power in the product market. Furthermore, with regard to the human capital composition of the male labour force in manufacturing, the assumption is made that there was no variation over time in the skills mix within each industry.

Consequently, to test for the incomes policies' influence upon the wage structure, an equation is specified which includes as dependent variable the annual coefficient of variation of average hourly wage receipts of male workers across the twenty two-digit

industrial branches of the Greek manufacturing sector. The coefficient of variation is used to represent the relative variability of the average hourly pay of male workers across manufacturing industry. Five independent variables are included; two to control for the business cycle influence, one for the influence of strike activity and two for the types of incomes policy under examination.

The norm-based and the indexation incomes policy variables are represented by two dummy variables taking the value of 1 for the years they were used. Both incomes policy variables take the value of 0 for the benchmark period 1967-74. According to our hypotheses, it is expected a positive association of the normbased incomes policy variable with the coefficient of variation in wages. Similarly, it is expected the positive association of the indexation policy variable with the coefficient of variation in the wage structure, but in this case the estimated parameter should be indicating the indexation's lower positive influence upon the wage structure.

The business cycle variables are one for inflation and one for the labour market conditions. As far as the business cycle influence is concerned, slightly modified specifications are tested with inflation and labour market conditions variables regressed with and without one-year lag. Of course, the lagged specifications imply no change in the way the explanatory variables are expected to be associated with the coefficient of variation in manufacturing wages.

The variable for inflation is the annual rate of inflation. According to our hypothesis, the inflation variable is expected to be positively associated with the coefficient of variation in manufacturing wages.

Conditions in the labour market are represented alternatively by four relevant variables. First, the average annual level of unemployment is expected to be negatively associated with variation in the wage structure. Second, the reciprocal of the annual level of unemployment, as its inverse is expected to be negatively associated with the coefficient of variation in wages, should be positively associated with the dependent variable. Third, the year to year change in the annual index for the registered unemployment is also used as a proxy for labour market conditions and, according to our hypothesis, it is expected to behave in the same way as the variable for the level of unemployment does, that is to be negatively associated with variation in relative wages. Four, the year to year change in the total employment of employees in the manufacturing sector is used as independent variable for labour market conditions. T+ is expected to be positively associated with the coefficient of variation in wages.

Not each one of these slightly different variables for conditions in the labour market are considered as being of the same explanatory value. They do represent alternative proxies of labour market conditions. Statistics on registered unemployment underestimate the total level of unemployment in the Greek labour market. This level of unemployment is better captured by the

annual labour force survey from which the variable for the average annual level of unemployment is drawn. However, statistics on registered unemployment are more reliable for labour market conditions faced by those unemployed with long experience in the labour market and more stable participation in the labour force. Normally, registration as unemployed is made for claiming benefits which are only paid to those with a certain minimum period in work. Thus, statistics on registered unemployment underestimate the number of unemployed by excluding new job seekers, casual labour, women dropping in and out of the labour force etc.

Moreover, both the level of unemployment and the level of registered unemployment may provide a biased view with respect to the labour market conditions prevailing in the manufacturing sector and, in particular, in industries employing more than 10 employees from which the sample for manufacturing wages is drawn. These industries employ less than three quarters of the total number of employees in employment, while employees in employment account, throughout the post-war period, for less than 50 per cent of the total labour force. This is why the variable representing the annual change in the average annual employment of employees in the manufacturing sector may be a better proxy for the labour market conditions prevailing in the Greek manufacturing sector.

Finally, the strike activity variable is the annual coefficient of variation in days lost in strike activity per 1000 employees.

It takes the value of 0 in the period 1967-74, during which no strike activity was recorded. According to our hypothesis, as variation in strike activity is not considered as a good proxy of bargaining power, there is no strong view on its positive or negative association with variation in the wage structure.

testing for the incomes policy influence upon In the manufacturing wage structure 22 annual observations, covering the period 1967-88, are used. Regression results of three sets of equations are given in Tables 6.4.1, 6.4.2, 6.4.3 and 6.4.4. Table 6.4.1 presents results for the equation in which the business cycle variables are incorporated in their actual annual values. Table 6.4.2 provides the results for the specification in which the business cycle variables are regressed with one-year lagged values. Table 6.4.3 presents the results of the regression in which the labour market conditions variable is incorporated with one-year lag while the inflation variable is incorporated in its actual annual values. Table 6.4.4 examines various specifications of the strike activity variable. In Tables 6.4.1-6.4.3 four equations are reported, as four variables for labour market conditions are used alternatively. In Table 6.4.4 four equations are reported as four varying specifications of the strike activity variable are tested.

In general, all the hypotheses with regard to the influence exerted upon variation in the wage structure by the incomes policy and the business cycle variables are supported by the regression results. It appears that incomes policy dominates the evolution of the manufacturing wage structure, while the business

cycle variables play a second but noticeable role.

All equations suggest that the norm-based incomes policy is associated with the widening of the wage structure. As far as indexation policy is concerned, the hypothesis that it caused the wage structure to become wider in comparison with the benchmark period, but relatively narrower than caused by the norm-based policy is also corroborated. The incomes policy coefficients for the norm-based policy are strongly significant across all sets of equations. The estimated coefficients of the indexation policy variable are less significant in some equations due to collinearity with the variable of inflation. But, overall, they suggest that indexation policy is associated with the widening of the wage structure as much as half as the norm-based policy did. This result fits with our hypotheses regarding the influence of incomes policy upon the manufacturing wage structure over the period under examination.

How much does the categorisation of incomes policy make difference in widening or narrowing the wage structure? This question can be answered by considering the standardised regression coefficients. The standardised regression coefficients for equation 4, which provides better results than any other equation in the set examined, are the following:

CVW = 0.32CPI + 0.22CHME - 0.40CVSTR + 1.13NIP + 0.42IIP

These results imply that the influence of the norm-based incomes policy in widening the manufacturing wage structure is three

times the influence of the indexation policy.

As hypothesised the independent variable for inflation, when regressed without time-lag, gives significant estimates with positive sign. But its estimates in the lagged specification are neither stable nor significant. A relatively better performance of the lagged specification would have been consistent with the predominance of formal and institutionalised annual wage bargaining. But the better performance of the actual rate of inflation is more consistent with the erratic wage rounds evolving within the informal system of collective bargaining.

The positive correlation indicates that the higher the rate of inflation the wider the wage structure becomes as high-bargaining power branches adjust their wages to past and expected inflation, while low-bargaining power branches fail to keep up with. In this context, the influence of inflation upon the wage structure is not minimal as the standardised estimated coefficient compares to that of the indexation policy variable.

Labour market conditions influence the wage structure in a quite stable way. Variation in the wage structure is a negative function of the level of unemployment. This negative association is corroborated by all three alternative variables representing the state of unemployment. Stability in signs as well as consistency are remarkable. In all sets, the variable representing the annual change in manufacturing employment, provides estimates which are consistent with the negative sign in the parameters for unemployment. The two variables for

registered unemployment and change in manufacturing employment perform better than the variable for the level of unemployment and its reciprocal.

Indeed, equations which include changes in manufacturing employment as a proxy for labour market conditions, provide the overall best fit. When it turns to how much influence the labour market variables exert upon the wage structure it appears that it is not very important in quantitative terms, as the standardised estimated coefficient for change in manufacturing employment is nearly half of that for the indexation policy variable.

The negative association between unemployment and variation in the wage structure implies that an increase in unemployment leads to a narrower wage structure. This indicates that industrial branches leading the widening of wage differentials are vulnerable to higher unemployment. This is consistent with the erratic character of the informal collective bargaining procedures which are not protected by the law and are not a central element of the official trade-unionism in Greek manufacturing.

Another interesting finding in all sets of regressions is the negative coefficient of the strike activity parameter which appears stable both in sign and in magnitude. The negative sign indicates that variation in the wage structure develops inversely to variation in strike activity. This inverse relationship indicates that strike activity it is not an indicator of bargaining power. The question which way works the causal link

between variation in the wage structure and variation in strike activity is addressed in Table 6.4.4 where two alternative variables on strike activity are tested with and without a one year lag. The estimated coefficient for the strike activity variable in the equation with the lagged specification is found insignificant. However, is found significant and negative in the specification without lag. This result indicates that strike activity varies inversely to the variation in the wage structure and suggests that variation in strike activity does not determine variation in the wage structure. Therefore the conclusion can be drawn that strike activity is not always an indicator of bargaining power, especially in the context of the informal system of collective bargaining.

To summarise, regression results suggest that, in comparison with the benchmark period 1967-74, formal incomes policies do widen the wage structure, with the norm-based policy exerting the stronger positive effect while indexation policy exerts a lower positive influence. Incomes policies dominate the making of the manufacturing wage structure over the period 1975-1988 with the business cycle variables exerting a rather secondary but noticeable influence. Inflation is positively associated with variation in the wage structure, while unemployment is negatively associated. Variation in strike activity is found negatively associated with variation in the wage structure. The regression results support the original hypotheses. They confirm that country-and period-specific characteristics with respect to the system of industrial relations, as analysed in earlier Chapters,

account for this unconventional association between variation in the wage structure and business cycle.

,

.

.

6.5. CONCLUSIONS.

In the Greek manufacturing sector, after a long tradition of permanent intervention in wage determination, the adoption of formal incomes policy since 1975 for controlling rising inflation, coincided with the rise of an informal system of collective bargaining at the plant level. In this context, the comparison of two types of formal incomes policy, with the period 1967-74 used as benchmark, leads to the conclusion that both types of policy, namely the norm-based and the indexation, are positively associated with variation in the wage structure. Indeed, while both policies were aiming at a narrow wage structure, although not with equal emphasis, the norm-based policy failed and, in fact, did widen the wage structure. Indexation policy was relatively more successful in compressing the manufacturing wage structure.

This difference in performance may be attributed to the workings of the informal system of collective bargaining in the manufacturing sector, via which high-wage high-bargaining power industrial branches were extracting their wage premium in the 1975-81 period. The indexation policy did cause a relatively narrower wage structure by protecting institutionally the relative wage of the low-wage low-bargaining power industrial branches. From this point of view, if the dispersion of the wage structure is used as criterion for the incomes policy evaluation, indexation can be considered as more successful. It appears that the indexation policy succeeded in narrowing relatively the wage

structure via two ways. First, by improving the relative position of the low-wage industrial branches. Second, by controlling the high-wage industrial branches. Thus, it is not a paradox that trade-unions while being strongly in favour of the indexation system, cast their criticism upon the scale used for partial indexation which causes the decline in inter-industry wage differentials.

Behind the issue of the incomes policy effectiveness and failure lies the issue of the institutional framework of wage determination in the manufacturing sector. Industrial relations developments such as the lack of a formal system for collective bargaining at the plant level and the rise of informal collective bargaining, cause the wage structure to become wider when the incomes policy is rather voluntary and allows room for informal collective bargaining.

Indeed, in this context, the business cycle causes contradictory influences upon the wage structure. Inflation undermines the relative wage of the low-wage less powerful industrial branches. Unemployment controls the high-wage high-bargaining power industrial branches in extracting a wage premium. In other words, because of incomes policies and the institutional developments in the manufacturing bargaining structure it appears that changes in the wage structure not only became less responsive to business cycle conditions but, indeed, their responsiveness is the inverse than that observed in industrialised market economies such as the U.S. and the U.K.

where collective bargaining evolves through relatively stable structures and processes.

With regard to industrial relations developments, given that strike activity was the main weapon for initiating new bargaining relationships and setting up the informal system of collective bargaining at the plant level, the inverse variation in strike activity suggests that the inter-industry efficiency of strike activity was not a function of the hours lost in industrial conflict. The evidence on the expansion of strike activity, along with its inverse variation compared to variation in the wage structure suggest that in many cases the low-wage low-bargaining power industrial branches have been locked in a vicious cycle of low relative wages and adversarial industrial relations.

TO EXPLAIN INTER-BRANCH VARIANCE IN WAGES, 1967-1988.									
		Equation							
Variable	SD	M	ES	1	2	3	4		
CVW	2.03	15.09							
СТ				13.94	12.73	12.80	12.48		
				(15.40)	(17.91)	(29.03)	(25.53)		
CPI	8.33	14.05	+	0.06*	0.06**	0.09***	* 0.08****		
				(1.68)	(1.93)	(2.62)	(2.61)		
U	2.43	4.46	-	-0.20					
RECU	0.17	0.31	+	(-1.04)	1.22				
CUDII	20 12	4 4 2			(0.61)	-0.02*			
CHRU	20.13	4.42	-			(-1.36)			
CHME	2.52	1.91	+			. ,	0.18 ^{**} (1.90)		
CVSTR	85.51	98.47	?	-0.01**	0.01**	-0.01**	-0.01**		
				(-1.85)	(-1.80)	(-1.74)	(-1.84)		
NIP			+	4.85***	** 4.88***	** 4.93***	** 4.84****		
				(4.20)	(4.07)	(4.42)	(4.53)		
IIP			+	2.29**	1.66*	1.40*	1.78**		
				(1.75)	(1.50)	(1.42)	(1.85)		
R ² -adjus	ted			.7 97	.788	.805	.823		
Durbin-W	latson f Esti	mation		1.75 0LS	1.71	1.87 OLS	1.73 OLS		
<pre>SD = Standard Deviation. M = Mean. ES = Expected Sign. CVW = Coefficient of variation in hourly wage of male workers across twenty two-digit manufacturing branches. CPI = Annual rate of inflation. U = Level of unemployment. U = Level of unemployment. RECU = Reciprocal of unemployment level. CHRU = Change in registered unemployment. CHME = Change in manufacturing employment. CHME = Coefficient of Variation in hours lost per 1000 employees in strike activity across twenty two-digit industrial branches. NIP = Norm-based incomes policy.</pre>									
<pre>NIP = Norm-based incomes policy. IIP = Indexation incomes policy. t-statistics in parentheses. Levels of Significance *=.1, ***=.025, *****=.005, **=.05, ****=.01</pre>									

REGRESSION EOUATIONS

<u>T0</u>	<u>EXPLA</u>	IN INTE	R-BR	ANCH VAR	IANCE IN WA	<u>GES, 1967-</u>	1988.
Variable	SD	M 	ES	5	6	7	8
CVW	2.03	15.09					
СТ				15.15 (13.40)	12.38 (15.78)	13.05 (30.18)	12.75 (25.29)
CPIt-1	8.55	13.66	+	-0.02 (-0.26)	0.003 (0.05)	0.05 (1.02)	0.01 (0.27)
^U t-1	2.32	4.34	-	-0.37^{*}			
RECUt-1	0.17	0.31	+	(1.07)	4.15 [*]		
CHRU _{t-1}	20.04	4.70	-		(1.54)	-0.3***	
CHMEt-1	2.53	2.00	+			(-2.10)	0.23 ^{**} (2.10)
CVSTR	85.51	98.47	?	-0.01 (-1.19)	-0.01 [*] (-1.37)	-0.01 ^{**} (-2.02)	-0.01 [*] (-1.50)
NIP			+	4.94 ^{**} (3.89)	*** 4.74 ^{***} (3.47)	** 6.04 ^{***} (5.60)	** 5.58 ^{*****} (5.13)
IIP			+	3.30 ^{**} (2.64)	** 2.66 ^{**} (2.42)	2.50 ^{***} (2.48)	2.76 ^{**} (2.66)
R ² -adjus Durbin-W Method o	ted atson f Esti	.mation		.770 1.88 OLS	.765 1.72 OLS	.789 2.41 OLS	.788 2.06 OLS
Notes: SD = Sta M = Mean ES = Exp CVW = C across t CPI = An U = Leve RECU = R CHRU = C CHRE = C CVSTR = in strik NIP = No IIP = In t-statis Levels	ndard ected oeffic wenty nual r l of u ecipro hange hange Coeffi e acti orm-bas dexati tics i of Sig	Deviati Sign. cient of two-dig tate of inemploy ocal of in regi in manu cient of vity ac sed inco on inco in paren gnifican	on. f va it m infl ment unen fact f Va ross mes thes ce *	ariation anufactu ation. ployment red unemp uring em riation twenty policy. policy. ses. =.1, ***	in hourly ring branch level. loyment. ployment. in hours lo two-digit i	<pre>wage of r es. ost per 100 ndustrial **=.005, *</pre>	nale workers 0 employees branches. *=.05, ****=.

REGRESSION EQUATIONS

×.

<u>T0</u>	EXPLA	<u>IN INTE</u>	R-B	RANCH VA	RIANCE IN	WAGES, 1	967-1988.			
	Equation									
Variable	SD	M 	ES	9	10	11	12			
CVW	2.03	15.09								
СТ				13.98 (13.32)	12.66 (16.41)	12.8 (33.8	2 12.50 6) (27.14	6 4)		
CPI	8.33	14.05	+	0.05 [*] (1.38)	0.05 [*] (1.33)	0.0 (2.3	7 ^{***} 0.09 3) (1.7	5 ^{**} 7)		
^U t-1	2.32	4.34	-	-0.19						
RECUt-1	0.17	0.31	+	(=0.95)	1.89					
CHRUt-1	20.04	4.70	-		(0.65)	-0.0	3**			
CHMEt-1	2.53	2.00	+			(-2.1	7) 0.19 (1.83	9 ** 8)		
CVSTR	85.51	98.47	?	-0.01 [*] (-1.73)	-0.01 [°] (-1.71)	* -0) (-1.6	.008 [*] -0.09 4) (-1.59	08 [*] 9)		
NIP			+	4.72 [*] (3.92)	**** 4.70 (3.65)	***** 5.2) (5.0	4 ^{*****} 4.9 6) (4.6	4 ^{*****} 4)		
IIP			+	2.06 [*] (1.68)	1.76 [*] (1.53)	* 1.8) (1.9	7 ^{**} 1.99 8) (2.0	9 ^{**} 0)		
R ² -adjus Durbin-W Method o	sted Natson of Esti	mation		.794 1.76 OLS	.788 1.69 OLS	3 .8 2.2 OL	32 .83 7 1.9 S OL	 22 6 S		
Notes: SD = Sta M = Mear ES = Exp CVW = C across t CPI = Ar U = Leve RECU = H CHRU = C CHME = C CVSTR = in stri NIP = No IIP = In t-statis Levels	andard bected Coeffic twenty nual r el of u Recipro Change Change Coeffic ce action ndexation stics i of Sig	Deviat: Sign. cient of two-dic two-dic two-dic two-dic in employ ocal of in reg: in manu- cient of twity ac sed inco in pare gnificat	ion. of v git inf une iste ufac of V cros omes nthe nce	ariation manufact lation. t. mploymen red unem turing e ariation s twenty policy. policy. ses. *=.1, **	in hour uring bran t level. ployment. mployment in hours two-digit	ly wage nches. lost per t industr	of male w 1000 emploial branch	orkers oyees es. ****=.		

.

REGRESSION EOUATIONS

<u>T0</u>	EXPLAIN	<u>INTEI</u>	<u>REG</u> R-BR	<u>RESSION</u> <u>E</u> ANCH VARI	QUATIONS ANCE IN WAG	<u>568, 1967-1</u>	988.	
Variable	SD	м	ES	4	Equati 13	ion 14	15	
CVW	2.03	15.09						
СТ				12.48 (25.53)	12.40 (20.23)	12.47 (19.09)	12.42 (19.49)	
CPI	8.33	14.05	+	0.08 ^{**} (2.61)	** 0.09 ^{***} (2.88)	** 0.09 ^{***} (2.60)	* 0.09 ^{****} (2.61)	
CHME	2.52	1.91	+	0.18 ^{**} (1.90)	0.18 ^{**} (1.76)	0.21 ^{***} (2.08)	0.22 ^{***} (2.31)	
CVSTR	85.51	98.47	?	-0.01**				
RSTR	85.12	97.72	?	(~1.04)	0.003 (1.12)			
RSTR _{t-1}	355 355	277	? ?			(0.34)	0.001*	
NIP			+	4.84 ^{**} (4.53)	**** 2.43 ^{***} (3.47)	*** 2.62 ^{***} (3.71)	(1.38) ** 2.35***** (3.45)	
115			+	1.78 ^{**} (1.85)	-0.33 (-0.38)	0.10 (0.12)	-0.11 (-0.15)	
R ² -adjusted .823 .821 .808 .829 Durbin-Watson 1.73 1.82 1.84 1.81 Method of Estimation OLS EIIM AR(1) EIIM AR(1) EIIM AR(1)								
Notes: SD = Standard Deviation. M = Mean. ES = Expected Sign. CVW = Coefficient of variation in hourly wage of male workers across twenty two-digit manufacturing branches. CPI = Annual rate of inflation. CHME = Change in manufacturing employment. CVSTR = Coefficient of Variation in hours lost per 1000 employees in strike activity across twenty two-digit industrial branches. RSTR = Rate of strike activity i.e. NIP = Norm-based incomes policy. IIP = Indexation incomes policy. t-statistics in parentheses. Levels of Significance *=.1, ***=.025, *****=.005, **=.05, ****=.01								

CHAPTER 7. CONCLUSIONS AND POLICY IMPLICATIONS

In a nutshell the conclusion of the thesis is that in the Greek manufacturing sector, despite the extensive and continuous regulation of wage determination procedures, changes in industrial relations which occurred after 1975 and were marked by decentralised and fragmented collective bargaining, were accompanied by flexibility in relative wages. The causes of this flexibility, which is largely non-competitive, are related to the rise of decentralised, fragmented and informal collective bargaining at the company and plant level.

This conclusion has been reached through an approach developed in three stages. First, we examined the main characteristics of the national system of industrial relations, that is the formal bargaining structure, the machinery of compulsory arbitration and the behaviour of salary and wage earners as expressed in strike activity. Second, we analysed the wage and employment policies undertaken over the period 1966-88 along with the basic developments as far as wages and employment in manufacturing are concerned. Third, based upon the inferences drawn regarding incomes policies and developments in industrial relations, we tested first for the existence of flexibility in relative wages and second for the influence exerted upon the wage structure by incomes policies, inflation, unemployment and strike activity.

Our approach was based on the assumption that industrial relations, through which labour markets operate, influence labour

market outcomes. More specifically, to analyse the national system of collective bargaining in Greece, we adopted as framework the twofold distinction first between centralised and decentralised bargaining, and second between coherent and fragmented bargaining structures. The aim of this approach has not been just to classify the Greek bargaining structure in a certain category, but to trace its continuity and its change over the period 1966-88, and when necessary beyond that.

Since 1955, when the basic legislation on collective bargaining was introduced, two have been the central elements of the national system of collective bargaining; first, the formalised and regulated by the law bargaining levels and units and second, the extensively used machinery of compulsory arbitration. The legislation, namely and mainly Law 3239, defined the levels and units at which bargaining can take place by permitting company and plant level collective bargaining only for exceptional cases and by excluding manufacturing industries. Therefore, the formal bargaining structure favoured by the existing legislation is rather centralised. Indeed, it has been supposed to be coherent as well, because the legislation by defining bargaining levels and units aimed to avoid overlapping among different levels of negotiation.

Arbitration, in turn, was made compulsory in the sense that one party can unilaterally refer a dispute to the arbitration court the award of which is binding for the parties involved in the dispute. The aim of compulsory arbitration when it was adopted

was to control industrial conflicts and minimise the occurrence of strikes as well as to promote tripartite collaboration.

After examining the aims and the provisions upon which the national system of collective bargaining has been based, we conclude that since 1975, as far as bargaining procedures and structures are concerned, there has been a gap between aims and outcomes, exemplified by the creation, especially in manufacturing, of a dual structure of collective bargaining. First, strong tendencies towards decentralised and fragmented bargaining have been observed which resulted in the creation of an informal system of collective bargaining at the company and plant level. Second, despite the central role played by arbitration, as a major proportion of dispute cases in the Greek labour market have been going to arbitration, its extensive and increasing use has been positively associated with the rise of strike activity. The evidence on the characteristics of strike activity suggests that the use of arbitration led to higher rates of strike activity rather than the other way round.

The increase after 1975 in the number and density of bargaining units acting inside and outside the formal bargaining structure has been impressive. This increase, indicating decentralisation and fragmentation of collective bargaining, can be explained by four factors referring to political, structural, economic and industrial relations developments. First, the political developments refer to the fall of the military regime and its succession by a parliamentary democracy, which led to a more liberal climate in industrial relations and trade unions. More

specifically, the freeze of trade-union activities during the period of the military regime, along with the appointment of progovernmental leaderships in the official structure of tradeunionism, resulted, after the fall of the military regime, in a political and organisational vacuum in Greek trade-unionism.

Second, the period 1967-74 was characterised by the most rapid expansion of manufacturing employment ever observed in Greece. This expansion was largely associated with the creation of new establishments in which new workers, lacking experience and tradition with regard to trade-union activities, had to organise from the very beginning their bargaining institutions and processes. Third, since 1975 the rate of inflation, which over the 1960s remained on average at levels lower than 2 per cent, soared at double-digit levels, and generated the need for wage and salary earners to keep up, at least, with inflation rates to preserve their real incomes.

Fourth, the formal bargaining structure, more specifically the formal levels and units for collective bargaining determined and certified by the law, were not compatible with the changing structure of the expanding manufacturing employment as company and plant level bargaining units were excluded. Furthermore, within the formal system of collective bargaining the control of compulsory arbitration by the Ministry of Labour, and its use as a machinery for implementing the governmental incomes policy, did not encourage salary and wage earners to seek wage settlements through the formal system.

Strike activity, the rate of which increased impressively after 1975 and since then remained at high levels, has been the vehicle by which salary and wage earners, representing new bargaining units, attempted to establish new bargaining relationships at the company and plant level. However, it is noteworthy that strike activity and informal collective bargaining were not equally developed across the board. Some industrial branches were more strike-prone while others where less active.

There are two reasons for considering that strike activity reflects efforts to initiate procedures and institutions alternative to the formal system of collective bargaining. First, it is the positive association between direct and indirect regulation of collective bargaining by the Ministry of Labour and strike activity. Second, it is the fact that strike activity was largely evolving at the company and plant level. Even if some of those strikes were protest-strikes, the collective action undertaken by salary and wage earners can be considered as indicator of new bargaining units at their early stages. In other words, it seems that while regulation and compulsory arbitration had a "freezing effect" upon the formal collective bargaining, at the same time they had a "de-freezing effect" upon collective bargaining evolving out of the formal bargaining structure.

After all, in Greek manufacturing the number and the density of bargaining units within the formal system as well as the number and the density of conflicts occurring at the company and plant level indicate the decentralisation and fragmentation of

the bargaining structure. Therefore, the developments over the period 1975-88 suggest that the Greek system of collective bargaining while typically, that is its formal part, is rather centralised, in reality, that is after taking into account its informal part, is decentralised and fragmented too. It falls somewhere between the two extremes, with fragmented collective bargaining being a dominant feature.

These developments in industrial relations over the period 1975-88, not surprisingly, were not very favourable for the successful implementation of formal incomes policies introduced since 1975. This is because, normally, incomes policies in order to effective require centralised collective bargaining. The two types of policy adopted over the period 1975-88 had the common characteristic that both favoured, more or less, the low paid and aimed at the compression of the wage structure. In conditions of rising inflation, incomes policies based largely on flat-rate and scale elements, (the system of automatic and partial indexation is typical of this case) normally result in a decline in wage differentials. This has been the case as far as wages of individuals are concerned. But it appears that in 1975-81, despite the egalitarian incomes policy, inter-industry wage differentials in manufacturing increased. Indeed, the variability in relative wages observed over the whole period 1966-88 indicates the existence of some flexibility.

Given the developments in the bargaining structure and the rationales and mechanics of incomes policies adopted since 1975, we examined the extent to which the variability in relative wages

reflects competitive or industry-specific non-competitive flexibility. The main issue examined with regard to competitive flexibility has been whether or not increases in relative wages were associated with the use of wages for retaining and attracting labour in expanding sectors. The main issue examined with regard to non-competitive flexibility has been whether or not the increases in relative wages reflected industry-specific productivity gains shared by the workforce or reflected industryspecific bargaining power and militancy.

The results suggest that the observed variability over the period 1966-88 mainly reflects non-competitive flexibility associated with long-run industry-specific productivity gains while in the short-run reflects industry-specific bargaining power and militancy. It appears that the decentralised and fragmented collective bargaining via the informal system created conditions conducive to non-competitive flexibility in relative wages. Indeed, the overall context of wage and employment regulation by governmental policies disfavoured competitive flexibility.

However, some sub-period characteristics are noteworthy. First, in 1967-74, when no formal incomes policy was in force and strike activity was banned, competitive forces were given a chance and were at work. Low unemployment, rapid expansion in manufacturing employment creating strong demand for labour. The wage fixing in line with the average productivity growth allowed expanding industries to use relative wages for retaining and attracting

labour. Second, in 1977-88, that is over the period of egalitarian incomes policies, the asymmetries in wage changes and in productivity changes at the industry level were found employment hindering because wages in industrial branches with below-average productivity growth were less responsive to that relatively poor productivity performance than wages in industrial branches with above-average changes in productivity. The causes of this asymmetry can be traced first, to the egalitarian incomes policies themselves and second, to the workings of the informal system of collective bargaining at the plant level.

The effects of the decentralised and fragmented collective bargaining upon relative wages were also expressed in the way that incomes policies, inflation and unemployment were associated, over the period 1967-88, with the evolution of the inter-industry wage structure. The predominant role of factors such as the establishment size, the skill and the sex composition of the labour force as determinants of the manufacturing wage structure and the stable hierarchy at the top and at the bottom of the manufacturing wage structure indicate that a two-sectors model of high-wage, concentrated and high-bargaining power industries on the one hand, and low-wage, less concentrated and low-bargaining power industries on the other, is substantiated.

In the context of decentralised, fragmented and informal collective bargaining it has been found that the norm-based incomes policy was associated with the widening of the wage structure twice as much as the indexation policy. Our explanation for this is that under the norm-based policy the high-wage high-

bargaining power branches, by considering the norm as the floor for additional increases in nominal wages, were able to extract a wage premium via the informal system of collective bargaining, while the low-wage industries had to struggle for getting wage settlements near the governmental norm.

The relatively narrower wage structure associated with the indexation policy is consistent with the fact that automatic indexation along with the scales for partial indexation protected and increased the relative wage of low-wage low-bargaining power branches, while at the same time, by being made statutory in 1983 and 1985-87 controlled, to a certain extent, wage rises in highwage high-bargaining power industrial branches.

The effects of the decentralised, fragmented and informal bargaining are also prominent in the way that economy-wide factors such as inflation and unemployment influenced the interindustry wage structure over the period 1967-88. Relative wages not only became less responsive to inflation and unemployment but their responsiveness was inverse than that observed in the U.S. and the U.K. manufacturing sector.

In the U.S. and the U.K., for instance, the existence of formal collective agreements lasting one, two or three years causes a certain stickiness in wage changes of the high-wage industries. But in Greece high-wage industries keep up with rising inflation via the informal system of collective bargaining, while low-wage industries fail to do so. Thus, as indicated by the positive correlation between inflation and variation in the

wage structure, rising inflation leads to wider inter-industry wage differentials.

A similar inverse association has been observed with regard to unemployment and variation in the wage structure. It appears that unemployment affects more the high-wage high-bargaining power industries, probably because rising unemployment undermines the workings of the informal system of collective bargaining. This negative association between unemployment and variation in the wage structure was not observed in the U.S. and the U.K. where formal collective bargaining in high-wage and high-bargaining power industries is not affected by changes in the level of unemployment.

With regard to the annual inter-industry variation in strike activity the results indicate the existence of a negative association with variation in the wage structure. In other words, the wider variation of the rate of strike activity from year to year did not cause an analogous variation in the wage structure. This could be explained on the basis that strike activity does not always indicate bargaining power. It may well operate as a way of expressing collective discontent and protest.

Incomes policy in Greece can be evaluated from two standpoints. First, according to its, more or less, stated rationales and objectives. Second, according to its overall contribution in macroeconomic and industrial relations developments. Normally these two standpoints are identical. But it appears that in the case of incomes policy in Greece these standpoints differed. This

has been due to the conflict between economic and political calculations of successive governments when designing and implementing incomes policy over the period 1975-88.

When, the annual change in nominal wages or the annual change in unit labour costs are adopted as criteria, as they are from a macroeconomic point of view rather important with respect to cost-push inflation, the distinction between success and failure of the norm-based and the indexation policy becomes less clear as both are related to good and bad performances and both, under the pressure of deteriorating current accounts, had to change their stance from nominal to real incomes policy. It appears that real incomes policy implemented via the indexation scheme has been far more effective in controlling real wages and unit labour costs than such policies implemented via the norm-based scheme.

As far as the inter-industry wage structure is concerned the experience suggests that it may become wider or narrower because of quite different influences such as the pressure exerted by the high-wage high-bargaining power industries or the indexationbased incomes policies securing relatively high relative wages for the low-wage low-bargaining industries. However, both cases signal the problems in the functioning of the Greek labour market. These problems are first related to governmental incomes policy itself, and second, refer to bargaining structures and procedures.

On incomes policy the main question is whether or not the system of indexation should be abandoned altogether. Such an option

would have to answer the questions how to control high-wage high-bargaining power industrial branches and how to offer protection for minimum wages to low-wage low-bargaining power industrial branches. Because it seems unlikely that changes in formal indexation provisions could lead to a reduction of the willingness of employees and trade-unions to preserve real wages and their share in income distribution. It is not the indexation system which makes them to behave in the bargaining process without having any money illusion.

Moreover, it is an open question as to whether the wide dispersion in manufacturing wages observed over the period 1976-81 set long-term standards which were suppressed by the indexation policy but were not disposed of by the narrower standards imposed in the period 1982-88. Indeed, a related question is whether or not the abandon of incomes policies would be followed by major disturbances to the wage structure.

On bargaining structures and procedures the main problem refers to the harmonisation of the formal bargaining structure with the real bargaining structure as well as to the creation of really independent machinery able to provide conciliation and arbitration services without being overridden by the governmental policies. The harmonisation of the dual bargaining structure, in turn, is related to the recognition of the right to bargain and contract collective agreements at the company and plant level. The institution of independent arbitration services depends largely on the initiatives of the Ministry of Labour which has
been addicted to direct and indirect regulation of wage determination. dependent it comes the question of obtaining more coherence in the bargaining structure.

The recognition of the right to bargain at the plant level may initiate changes in the bargaining structure towards more coherence, as many not really representative formal bargaining units would be made redundant. But as the existing structure has been built over a quite long period, decrees and rational normative approaches and decisions at the top of the trade-unions structure do not seem, in the short-run, able to generate any impressive reform.

Finally, in the context of decentralised and fragmented collective bargaining it is not easy to achieve corporatist-like arrangements based upon a quid pro quo for greater flexibility in relative wages and moderation in wage determination on the one hand, and continued policies for employment protection on the other. The prevailing conditions may well continue working in favour of non-competitive flexibility which is not a solution but part of the problem.

APPENDICES

APPENDIX 1: SYSTEMS OF ARBITRATION IN THE U.S. THE U.K. AND AUSTRALIA

These three systems of arbitration developed in a common cultural tradition but in different contexts of industrial relations. Blain, Goodman and Loewenberg (1987) compared mediation, conciliation and arbitration frameworks in these countries and concluded that the British and the American industrial relations systems can be essentially characterised as collective bargaining systems with conciliation mediation and arbitration occupying an ancillary position, while the Australian system involves the greatest degree of compulsion. They drew the line by using as main criterion the degree of compulsion in the existing legislation and machinery.

From our point of view these distinct models of arbitration which are associated with different systems of industrial relations provide the basis to discuss the conditions under which arbitration may be compatible with collective bargaining first with regard to the key issue who is who performs the role of arbitrator, considers the parties offers and external information to arrive to the final award, be it compulsory or not, and second with regard to the degree of governmental intervention, as the state is at the same time responsible for setting the legal framework, for protecting the national economic interest and often for serving as industrial peacemaker. For instance, incomes

policies have been more often at work in the U.K. and in Australia than in the U.S.

First, let us consider the system of arbitration provided by independent professionals which is used in the U.S. Similar system of arbitration is used in Canada. According to Cordova (1987:331), in Jamaica, Malaysia, Panama, the Philippines and Zambia systems of private arbitration like the north-american are practised too.

In the U.S. the collective bargaining structure is highly decentralised. Collective bargaining takes place mainly at the company and the plant level. Collective agreements are very detailed and the great majority of them content arbitration clauses. (Wheeler and Hoyt, 1987). Conventional and final-offer or binding arbitration are common forms of interest arbitration in use. Binding arbitration is mainly used in the public sector. Many state laws provide for such arbitration of unresolved disputes. Arbitrators are selected among lawyers and academics. Their decisions are usually considered as final, binding and unappealable. Overall, as Bloom and Cavanagh (1987:354) concluded, the American system of industrial relations exhibits a strong normative preference for resolving disputes without the aid of third parties.

Second, the system of voluntary arbitration provided in the U.K. by the ACAS an autonomous institution. In the U.K. arbitration is voluntary and therefore an unwilling party cannot be compelled to go to arbitration. Moreover, arbitration awards are not legally

binding but arrangements for arbitration are made on the basis that the award will be honoured by the parties. In this context of voluntary arbitration even the terms of reference are usually agreed by the parties. Arbitrators are not members of the staff of the ACAS but are appointed by ACAS from lists of industrial relations specialists.

There are three types of arbitral bodies. In the first type, the arbitration procedure is performed by a single arbitrator. In the second type it is performed by specially appointed boards of arbitration consisted of a chairman and one a side members. In the third type arbitrator is the Central Arbitration Committee which is a more permanent tripartite body the members of which are appointed by the Secretary of State for Employment from lists proposed by the interested parties.

However, as Brown (1987:113) argued, the predominant position of collective bargaining in conflict resolution in Britain and the fact that the government has largely and for a long period avoided involvement in dispute resolution, has meant that there has been little scope for third party institutions. Moreover, the main bulk of disputes referred to ACAS is concerned with conciliating over industrial grievances and only a small proportion of disputes procedures involve an ACAS arbitrator at the final stage. Overall, in the U.K. arbitration has not been a central element in conflict resolution and in industrial relations in general.

Third, the Australian system of compulsory arbitration provided

by the ACAC. an autonomous institution. Similar systems of compulsory arbitration are used in New Zealand and Israel. According to Blanpain (1987:7) the Australian system of compulsory arbitration has been adopted by a great number of developing countries like Singapore, Kenya, Trinidad and Tobago, Zambia, Sierra Leone and Malaysia. Perhaps this wide adoption of the Australian model of compulsory arbitration is related to the fact that in the 1960s compulsory arbitration was considered and proposed as the best choice, with regard to equity in distribution and political stability, for developing and less developed countries. (Turner, 1965 and Braun, 1975:212-213).

In Australia the system of arbitration has been under continuous and systematic scrutiny . (Committee of Review into Australian Industrial Relations Law and Systems, 1985). It includes both federal and state arbitral bodies. Either party to the bargaining process can refer a dispute to the arbitral body. The main arbitral body is the Australian Commission of Conciliation and Arbitration which has a long history going back to the beginning of the century. All members of the Commission are appointed by the Governor General on advice of the government in office. However, the ACAC is independent from the government. The government is only responsible for the legislation governing the system and cannot intervene directly in the wage and industrial relations regulation. The Commission itself is empowered to disputes extending to more than one state. intervene in Arbitration is compulsory in two ways. First, both parties to the bargaining process are required to submit to a compulsory

327

procedure for representing their arguments. Second, awards are binding on the parties. Awards specify minimum standards of pay and conditions. Federal awards tend to set the pattern for all other tribunals.

Under the legislation on arbitration all federal unions are required to register with the arbitration authorities. Only registered organisations are entitled to institute proceedings before the Commission and obtain or enforce awards. Conciliation procedures are the first step undertaken by the commission. If parties are able to reconcile their differences through conciliation may present a memorandum of agreement to the Commission for certification or request that an award be made. A certified agreement has the value of an award. If conciliation is unsuccessful totally or partially the commission is obliged to resolve the dispute by arbitration.

The Commission consists of a President, Deputy Presidents, a Senior Commissioner and Commissioners. They are permanent members of the Commission. The President is required to have a legal background but the rest should be persons with experience at "high level" in industrial relations or have a relevant university degree. All members have a tenure until age 65. Presidential members appointed before 1972 have tenure until 70. The Commission works in a panel system. Panels are responsible for a number of sectors or industries.

Thus, in Australia traditional tasks of collective bargaining have largely been shouldered by the statutory conciliation and

arbitration machinery. Therefore the scale and the significance of collective bargaining vary considerably when compared with other industrialised market economies. Collective bargaining appears to have been in a secondary position as a mode for wage and job regulation. This is why Windmuller (1987:92) in his comparative study of systems of industrial relations considers that collective bargaining is not one of the central features of the Australian system. The centrality of conciliation and arbitration makes it fundamentally different from other systems in industrialised market economies. A rather similar system of arbitration has been used also in New Zealand which has also a long history going back in the beginning of the century.

The Australian system of compulsory arbitration has been the subject of extensive research comparing its labour market outcomes with those of systems of collective bargaining. Overall, as the Committee of Review into Australian Industrial Relations Law and Systems (1985: 169) reported, the evidence as to how the the system of compulsory arbitration has affected Australia' s experience in respect of inflation and unemployment is less decisive and "certainly does not support confident assertions about ill-effects due to the Commission's presence".

Mulvey (1986) reviewed the literature on the effects that compulsory arbitration might have had upon the wage structure and concluded that the evidence is ambiguous. Many studies such as those by Brawn et all (1980) and Norris (1980) who compared the wage structure in Australia with that in the U.K. and the U.S. to

test whether compulsory arbitration brings about a more egalitarian wage structure than would obtain under free collective bargaining show less dispersion in the interoccupational and inter-personal distribution of earnings in Australia as compared with the U.K. and the U.S. Only the lowpaid workers were found better off in Australia and this can be attributed to arbitration. But the differences are of modest proportion and they cannot be interpreted as indicating that compulsory arbitration induces severe distortions of the wage structure.

Withers et al (1983) examined how well the Australian system of wage determination facilitates disaggregated labour market adjustment by comparing it with the major alternatives in the U.S. the U.K. and Sweden and found not important differences between the Australian system of compulsory arbitration and its alternatives. Therefore, we can draw the conclusion that, while arbitration may affect the wage structure in an egalitarian way through comparability, not important differences between labour outcomes under collective bargaining and under arbitration are observed at the level of national systems of industrial relations.

APPENDIX 2: EMPLOYMENT OF SALARY AND WAGE EARNERS IN GREEK MANUFACTURING.

In Section 4.4 we discussed the characteristics and the evolution of the employment of wage earners in manufacturing over the period 1965-1988. This appendix provides the Tables which present the characteristics, the distribution and the evolution of the employment of both salary and wage-earners in the Greek manufacturing sector. Tables with A prefix correspond to those included in Section 4.4.

Table A4.4.8.

CHANGES IN EMPLOYMENT	OF <u>SALAR</u> <u>BY</u> Selecte	<u>Y AND WAGE</u> BRANCH ed Periods	E EARNERS	IN MANUFACTURING
Period	1967/74	1975/81	1982/88	1965/88
Branch		change pe	er cent	
20	39.91	-10.05	49.94	99.20
21	54.52	-10.26	39.22	155.73
22	-29.00	16.63	-10.59	-44.42
23	29.84	1.11	8.37	41.99
24	126.85	9.15	34.54	298.91
25	62.30	-10.48	6.92	40.55
26	55.94	-12.48	1.26	27.22
27	9.83	-1.45	45.80	68.99
28	19.18	-5.09	-0.06	12.10
29	1.35	-5.17	34.54	11.43
30	49.64	27.69	-17.01	107.80
31	49.18	11.54	19.24	102.81
32	160.84	41.71	8.75	334.83
33	28.01	-8.25	36.96	50.28
34	92.06	-14.65	33.86	214.02
35	55.47	10.22	-2.10	55.16
36	40.11	-5.18	-16.01	12.05
37	92.00	-7.75	11.97	86.99
38	116.26	68.97	-26.23	157.35
39	82.76	-29.08	55.22	115.29
A11	48.24	5.95	12.37	78.64
Number of Branches with Expanding				
Employment	19	8	14	19

Sources: 1967-1975, NSSG Annual Industrial Survey, Table 6. 1976-1988, NSSG Labour Force Statistics, Table 3.

332

Table A4.4.9.

Year	1965	1970	1975	1980	1985	1988
Branch						
			per cent	:		
20	13.09	14.31	13.01	11.03	13.54	14.60
21	2.01	2.68	2.67	2.35	3.03	2.88
22	6.72	3.94	2.40	2.66	2.17	2.09
23	20.26	17.48	17.76	17.77	16.13	16.10
24	5.37	6.32	9.40	9.58	11.95	11.98
25	2.50	2.73	2.57	2.26	2.09	1.96
26	2.04	2.16	2.05	1.71	1.50	1.45
27	2.77	2.64	2.18	2.09	2.46	2.62
28	3.35	3.17	2.63	2.53	2.22	2.10
29	1.81	1.42	1.09	0.99	1.33	1.13
30	3.00	3.27	3.64	4.15	3.75	3.49
31	5.66	5.73	5.62	5.78	6.45	6.43
32	0.61	0.84	1.09	1.46	1.29	1.48
33	7.88	7.41	6.52	5.85	6.82	6.63
34	1.50	2.65	2.82	2.41	3.05	2.63
35	6.66	6.47	6.54	6.54	5.83	5.79
36	3.36	3.41	3.15	2.89	2.30	2.11
37	4.42	5.56	5.27	4.66	4.80	4.63
38	6.23	6.79	8.52	12.43	8.42	8.97
39	0.78	1.03	1.07	0.86	0.87	0.93
A11	100	100	100	100	100	100
(20, 23,						
24, 38)	44.95	44.90	48.69	50.81	50.04	51.65
Total	218938	248892	335853	360284	391840	391119

DISTRIBUTION OF SALARY AND WAGE EARNERS ACROSS MANUFACTURING INDUSTRI

Sources: 1965-1975, NSSG Annual Industrial Survey, Table 6. 1976-1988, NSSG Labour Force Statistics, Table 3.

.

Table A4.4.10.

<u>AVERAGE</u> <u>EMPLOYMENT</u> <u>OF</u> <u>SALARY</u> <u>AND</u> <u>WAGE</u> <u>EARNERS</u> <u>IN</u> <u>MANUFACTURING</u> <u>INDU</u> Selected Periods

	Average 1967-74		Average 1975-81		Average 1982-88		Average 1966-88	
		Share		Share		Share		Share
Branch		8		8		*		8
20	35632	13.62	40097	11.24	49743	13.04	41069	12.62
21	7016	2.68	8438	2.37	10765	2.82	8514	2.62
22	9722	3.72	9471	2.66	8305	2.18	9400	2.89
23	46653	17.83	61961	17.38	62196	16.30	55987	17.20
24	17651	6.75	34068	9.55	45939	12.04	31020	9.53
25	7203	2.75	8013	2.25	7960	2.09	7614	2.34
26	5399	2.06	6873	1.93	6165	1.62	6045	1.86
27	6692	2.56	7579	2.13	9488	2.49	7807	2.40
28	8083	3.09	9120	2.56	8696	2.28	8573	2.63
29	3495	1.34	3686	1.03	4522	1.19	3877	1.19
30	9010	3.44	14238	3.99	15105	3.96	12375	3.80
31	15108	5.77	20331	5.70	25074	6.57	19634	6.03
32	2211	0.84	5064	1.42	5371	1.41	4005	1.23
33	18926	7.23	21536	6.04	24954	6.54	21508	6.61
34	6767	2.59	8541	2.40	10458	2.74	8332	2.56
35	16816	6.43	23223	6.51	23026	6.03	20566	6.32
36	8808	3.37	11548	3.24	8547	2.24	9512	2.92
37	14852	5.68	17453	4.89	17622	4.62	16253	4.99
38	18980	7.25	41918	11.76	34434	9.02	30443	9.35
39	2594	0.99	3425	0.96	3172	0.83	2989	0.92
All	261617	100	356584	100	381542	100	325522	100
(20 23								
24 38)	45.45		41.93		50.40		48.70

Sources: 1967-1975, NSSG Annual Industrial Survey, Table 6. 1976-1988, NSSG Labour Force Statistics, Table 3.

Table A4.4.11.

			1965-198	8		
Vear	1965		1967	1968	1060	1970
Branch	1905	1900	1907	1908	1909	1970
20	2.05	6.92	0.13	-1.04	3.79	12.98
21	10.13	19.56	12.39	-8.00	-0.17	22.93
22	7.62	-4.94	-13.63	-8.31	-12.47	1.28
23	0.98	2.31	-3.85	-6.44	-2.39	9.17
24	5.06	3.90	5.34	0.65	0.62	20.74
25	13.75	3.73	2.38	-2.05	4.94	13.82
26	3.16	2.17	0.07	0.63	-3.28	20.90
27	16.07	8.26	-0.02	1.01	-5.65	5.20
28	0.73	6.38	-3.83	-2.96	3.32	4.92
29	-1.22	-5.22	0.77	-5.67	-1.51	0.68
30	4.72	8.88	3.33	4.20	15.81	-8.76
31	19.28	3.91	-1.60	2.13	4.98	4.95
32	21.64	3.45	2.47	11.54	-5.02	39.57
33	14.05	3.50	-2.18	-2.70	-0.96	9.57
34	23.82	37.13	2.42	7.18	18.10	12.91
35	3.07	1.09	-5.65	1.82	1.96	11.43
30	5.89	4.04	1.61	-3.13	7.69	4.80
37	3.77	-1.95	9.29	1.01	8.8/	21.29
20	0.88	1.8/	-5.91	-1.96	11.39	18.38
נוא	10.14	1.42	1.15	6.94	1.67	2/.8/
		J.//		-1.90	2.10	11.00
Year	1971	1972	1973	1974	1975	1976
Branch						
20	4.82	0.65	3.96	9.92	1.80	-8.95
21	10.56	6.51	5.17	10.49	-1.91	-8.99
22	-3.13	-9.62	-1.77	1.56	-6.10	18.83
23	6.45	6.35	8.15	6.38	5.31	2.53
24	8.92	10.98	12.45	36.47	8.28	-4.12
25	6.11	10.52	10.12	7.42	-8.28	-8.09
26	1.32	4.55	3.79	20.53	-3.62	14.12
27	2.34	0.91	0.90	5.13	1.57	5.78
28	5.80	0.54	4.23	2.19	-1.20	6.46
29	-0.54	-12.96	1.04	23.87	-4.87	1.17
30	8.68	9.83	6.42	7.01	10.55	7.97
31	2.27	10.60	10.39	6.17	-0.07	12.23
32	-3.54	15.24	33.39	18.96	-0.65	44.68
33	4.81	1.00	5.12	8.94	-2.09	0.54
34 25		/.01	14.02	1.99 11 FO	1.03	-10.53
32	4.00	0.20	0.93 11 15	0 4 0 8 C . T T	1.4/ -2 76	
30 27	3.32 11 71	10 50	10 43	7.40 7 77	-2.70	14.JU 0 12
37 72	14./4 20 <i>1</i> 7	70.30	10.43 Q 7/	2.13 17 33	-1.10	0.13
20 20	5 96	0.33 7 65	2.34 11 20	11.22	1.22 6 70	40.75
50			17467		U./7	U . 44
LIA	6,56	5.21	7.64	10.50	1 19	5.09

ANNUAL CHANGES IN MANUFACTURING EMPLOYMENT OF SALARY AND WAGE EARNERS BY BRANCH

(continued)

Table A4.4.11 (continued)

<u>IN</u>	MANUFACTUR	ING	<u>EMP</u>	LOYMENT	OF	BALARY	AND WAGE	<u>EARNERS</u>	<u>BY</u>	<u>BRANCH</u>	
					196	5-1988					
	Vear	10		1978	2	1979	1980	1081		1082	
	Branch			1970	•	1777	1900	1901	•	1902	
	20	-0.	37	1.14	ł	-4.25	3.54	-1.10)	-3.14	
	21	0.	67	4.17	7	1.19	-2.20	-4.99)	0.51	
	22	4.	62	-1.81		0.25	-2.54	-2.20)	-2.63	
	23	2.	90	-0.38	3	0.35	1.78	-5.81		-3.66	
	24	9.	84	7.69)	7.67	-10.52	-0.10)	1.04	
	25	Ο.	91	-3.11	L	0.85	4.09	-5.10)	-7.12	
	26	-4.	67	-4.93	3	-6.27	-7.72	-2.17		-6.57	
	27	0.	22	0.73	3	-1.18	-2.25	-4.45	5	-2.60	
	28	-0.	94	-0.16	5	1.58	-3.48	-8.06	,	-1.90	
	29	2.	44	2.89	•	-3.85	-4.87	-2.80)	-5.15	
	30	3.	51	6.74	ł	5.98	-3.33	4.48	;	5.37	
	31	-3.	86	-2.46	5	1.08	3.63	1.19)	0.05	
	32	2.	32	-3.13	3	2.95	-2.88	-1.16	,	2.72	
	33	0.	80	-0.63	3	-2.50	-1.90	-4.75	,	-5.76	
	34	-1.	67	-0.57	7	2.38	2.19	-6.74	:	-4.80	
	35	6.	76	1.96	5	0.65	-1.55	2.73		-4.43	
	36	5.	61	0.08	3	-5.89	-13.66	- 3.58	;	-2.32	
	37	-7.	. 67	-0.52	2	-3.73	-0.76	-2.79)	-1.03	
	38	9.	.44	-1.27	7	0.73	2.20	7.93		-1.60	
	39	0.	.09	-6.99	•	-1.89	-11.89	-17.20)	-7.78	
_	A11	2.	58	0.57	7	0.22	-1.28	-1.24	:	-2.18	
	Year	10	983	1984	 L	1985	1986	1987	,	1988	
	Branch		.05	100		1905	1900	1507		1900	
	20	17.	92	6.76	5	10.71	-0.05	1.92		5.59	
	21	26.	60	3.96	5	11.66	-0.89	-2.55		-1.91	
	22	-10.	49	-7.31	L	11.90	8.52	-20.17	,	11.19	
	2 3	3.	80	0.52	2	4.24	3.60	-1.28	}	-2.58	
	24	44.	51	0.17	7	-7.12	-1.31	-0.27	,	1.67	
	25	19.	86	-6.18	3	1.55	-0.32	-5.02		-1.11	
	26	34.	85	-6.85	5	-16.85	-4.57	2.81		-1.18	
	27	32.	.33	1.30)	2.50	4.17	7.44		-5.19	
	28	14.	. 53	-2.85	5	-4.88	-1.11	-1.12		-3.41	
	29	24.	94	10.85	5	14.86	2.25	-11.69)	-6.33	
	30	0.	36	-1.69	•	-9.56	-1.81	-4.07	,	-1.25	
	31	30.	69	-5.62	2	-2.85	-0.09	0.16	5	-0.56	
	32	-3.	06	-0.49	•	-1.72	8.69	2.74		2.73	
	33	32.	.79	1.54	l I	4.71	-0.23	-3.09)	0.34	
	34	32.	.79	5.38	3	11.11	-1.95	-10.20)	-2.23	
	35	-0.	. 08	-0.19	•	-0.94	2.46	-1.76	5	-1.55	
	36	-16.	. 09	-2.27	7	12.05	-5.34	-6.49)	3.26	
	37	6.	. 17	-2.17	7	11.95	-0.88	-4.88	;	2.14	
	38	-34.	.06	-4.01	L	9.61	-5.47	4.81	-	7.31	
	39	24.	.97	0.05	5	15.35	1.37	0.54		5.59	
_	All	9.	.27	-0.07	7	3.10	0.10	-1.30)	1.03	

ANNUAL CHANGES

336

.

7

APPENDIX 3: RELATIVE WAGE FLEXIBILITY IN GREEK MANUFACTURING, CORRELATION AND REGRESSION RESULTS AFTER INCLUDING IN THE EMPLOYMENT VARIABLE BOTH SALARY AND WAGE EARNERS INSTEAD OF WAGE EARNERS ALONE.

A methodological issue concerning the variable for employment has to be addressed before proceeding in testing for the existence of relative wage flexibility in Greek manufacturing. Which variable for employment should we use? Employment of salary and wage earners or employment of wage earners alone? Theory, of course, does not specify which changes in employment should be looked at primarily. For instance, when we discuss the expanding employment, and we consider it as indicator of excess demand for labour in the short-run, under inelastic supply of labour, should we refer to both salary and wage earners or only to wage earners?

This is a matter which, at first sight, should be resolved on the basis of data homogeneity. From this point of view, as the variable for wages is the annual average hourly pay of male and female wage earners, the variable for employment of wage earners was preferred. But in the correlation analysis as well as in the cross-section analysis, to allow for any important difference caused by, or causing, substitution effects, we used both variables for employment, that is employment of salary and wage earners and employment of wage earners alone.

Changes in nominal wages refer to average hourly wage of workers

of both sexes, while changes in employment are represented by two alternative variables, one for salary and wage earners and a second for wage earners alone. Correlation coefficients for salary and wage earners and for wage earners alone, were, sometimes, found slightly different but not important differences are observed with regard to either their sign or their level of significance. In fact, as shown by correlation analysis results not reported here, correlation between changes in these two alternative variables for manufacturing employment was found very high, sometimes approaching the value of unity.

Moreover, an interesting issue is the relatively higher, compared to changes in employment of wage earners alone correlation coefficient found for changes in employment of both salary and wage earners. It seems that higher wage increases were offered when the employment of salary and wage earners was expanding rather than when the employment of wage earners alone was expanding. Does this indicate that increases in employment of salary and wage earners give the signal of higher demand for wage earners? This can be explained by an internal labour market situation in which filling and hiring from within the firm is observed.

In the test for industry-specific influence upon changes in industry nominal wages we used these two alternative variables for manufacturing employment as well. One including both salary and wage earners and one including only wage earners. Not important differences were observed. This appendix contains the correlation and regression results of the specifications where

both salary and wage earners were included in the employment variable. The following Tables with A prefix correspond to those included in Sections 5.4 and 5.5.

Table A5.4.1.

	BIVARIATE CORRELATION COEFFICIENTS RELATING CHANGES IN NOMINAL WAGES AND CHANGES IN EMPLOYMENT. (Time periods: 1966-88, 1967-74, 1975-81, 1982-88).										
Cha	anges	<u>Full</u> <u>Period:</u>	<u>1966-1988.</u>	<u>Period: 1967</u>	-1974.						
1	year	65/66-87/88	08*	66/67-73/74	.21***						
2	years	65/67-86/88 n=440	08*	n=100 65/67-72/74 n=160	.16**						
3	years	65/68-85/88 n=420	16***	65/68-71/74 n=140	.09						
5	years	65/70-83/88 n=380	31***								
10	years	65/75-78/88 n=280	38***								
23	years	66/88 n=20	20								
Cha	anges	<u>Period 1975-</u>	<u>1981.</u>	Period 198	2-1988.						
1	year	74/75-80/8 n=140	1.007	81/82-87 n=14	/88 .02	2					
2	years	73/75-79/8 n=140	1.09	80/82-86 n=14	/88 .19 0)**					
3	years	72/75-78/8 n=140	103	79/82-85 n=14	/88 .13 0	3					
5	years	70/75-76/8 n=140	126***	78/82-83 n=14	/88(0)7					
10	years	65/75-71/8 n=140	119**	72/82-78 n=14	/88 .19) ~ ~					

Notes: Levels of Significance: * = 0.10, ** = 0.05, *** = 0.01.

· •,

Table A5.4.2.

BIVAR	IATE CORRI	ELATION COE	FFICIENTS	RELATING	CHANGES I	N NOMINAL WAGES
	AND CHANC	<u>GES IN EMPI</u>	OYMENT OF	SALARY A	ND WAGE EAD	RNERS.
	(Short	-run and Lo	ong-run ch	anges ove	r the perio	od 1966-88)
		• • • • • • • • • • • • • • • •				
	Period	of Change:	•	-	• •	••
W = = ==	1 year	2 years	3 years	5 years	10 years	23 years
rear						
/66	.10					
/67	.27	02				
/68	24	.03	30			
, /69	.28	04	08			
/70	06	003	23	19		
/71	.31	.22	.21	15		
/72	.27	.37	.15	11		
/73	.20	.19	.30	.29		
/74	03	28	22	.05		
/75	26	01	31	06	28	
/76	27	18	23	10	21	
/77	.16	.29	.11	18	.07	
/78	16	10	.10	13	.17	
/79	57 ^^^	55^^^	36	24	.07	
/80	.28	.14	06	.33	.19	
/81	.17	.50	.24	.19	.05	
/82	.04	.11	.40	.13	.04	
/83	05	.18	.07	.02	.10	
/84	.29	.16	.24	.08	.07	
/85	.17	.49	•06	.11	.11	
/86	.15	.28	.56	.22	10	
/ 8 /	.06	.13	.1/	.003	06	20
/88	13	.09	10		02	20

Notes: N = 20 (Two-digit industrial branches of manufacturing). Levels of Significance: * = 0.10, ** = 0.05, *** = 0.01.

Table A5.4.3.

BIVARIATE CORRELATION COEFFICIENTS RELATING ANNUAL CHANGES IN NOMINAL WAGES AND EMPLOYMENT OF SALARY AND WAGE EARNERS. (comparison between coefficients for all branches and coefficients for expanding branches)

	Per	iod of Change:1 year	
Year	CHSWEi	CHSWEi>0	
/66	.10	.09	
/67	.27	.14	
/68	24	41	
/69	.28	.36	
/70	06	04	
/71	.31	.32	
/72	.27	.17	
/73	.20	.21	
/74	03	03	
, 75	26	20	
/76	27	24	
, 77	.16	.04	
/78	16	35	
, /79	57***	45**	
/80	.28	01	
/81	.17	.14	
/82	.04	.25	
/83	05	04	
/84	.29	.33	
/85	.17	.15	
/86	.15	.24	
/87	.06	14	
/88	13	18	

Notes: N = 20 (Two-digit industrial branches of manufacturing). Levels of Significance: * = 0.10, ** = 0.05, *** = 0.01.

•

Table A5.5.1.

ESTIMATES OF THE INDUSTRY SPECIFIC DETERMINANTS OF NOMINAL WAGE CHANGES ACROSS MANUFACTURING INDUSTRIES IN GREECE.									
Depend	lent Variabl	le: CHWi							
Period	l CT (CHSWEi	CHPRi	WLi	CHFEI R^2/R^2	F-value			
1966	4,695.76	.51	2.37**	-87.21**	.46	4.58**			
/88 n=20	(12.14)	(.60)	(2.29)	(-2.76)	.36				
	4,691.77		2.27**	-82.84**	.45	6.95**			
	(12.36)		(2.27)	(-2.75)	.39				
1967	124.25	02	.13	82	.08	.45			
/74 n=20	(4.29)	(18)	(.94)	(35)	10				
	125.49		.14	-1.01	.08	.69			
	(4.59)		(1.12)	(50)	03				
1975	215.80	.33	.16	1.62**	.31	2.39**			
/81 n=20	(7.46)	(1.31)	(1.05)	(2.06)	.18				
	212.85		.06	1.81**	.24	2.62**			
	(7.23)		(.42)	(2.28)	.15				
1982	192.34	07	.001	17	40 .22	1.04			
/88 n=20	(10.19)	(05)	(.005)	(-1.64)	(55) .01				
	191.07		.03	16	41 .20	1.36			
	(10.46)		(.23)	(-1.67)	(59) .01				
1977	887.61	18	29	-2.74^{**}	$-4.62^{*}.61$	5.88**			
/88 n=20	(15.91)	(42)	(87)	(-2.87)	(-2.01) .51				
	885.61		29	-2.72**	* -4.63*.61	8.20***			
	(16.36)		(89)	(-2.93)	(-2.07).53				
Notes	CHWi = Char CT = Con CHSWFi = Char CHSWFI = Ch	ange in nstant T	Nominal 1 erm. Employmen	Hourly Wa	ge Paid in Br	anch i. Farners			
	in	Branch	i.	no or bur	und huge				
	CHPRi = ChiWLi = In	ange in itial Ia	Productive vel of Wa	vity in B age in Br	ranch i. anch i.				
	CHFEi = Ch	ange in	the Share	e of Fema	le Employment	in Branch i.			
	Levels of a t-statisti	Signific cs in br	ance: * ackets.	= 0.10,	** = 0.05, **	* = 0.01.			

Table A5.5.2.

<u>ESTIMA</u>	TES OF CHAN	THE INDU GES ACROS	ISTRY SPEC B MANUFACT	URING IN	TERMINAN NDUSTRIE	<u>NTB OF</u> B IN GF	NOMINAL REECE.	L WAGE
Depend	lent Va	riable: Cl	 					
Period	l CT	CHSWEI	CHPRi	SIi	WLi	CHFEI	R ²	F-value
1966 -88 n=460	17.83 (28.50	12 ^{**})(-2.42)	11** (-2.80)	* .20** (5.36)	** .000 (.093)3 5)	.09	10.90***
1967 -74 n=160	-4.29 (-2.54	05) (73)	12 ^{**} (-3.02)	* 10.62 (2.40)	*** .87* (9.80)	**	.43	29.80***
1975 -81 n=140	23.47 (22.99	.04) (.64)	.03 (.63)	005 (19)	.02 (1.44)		.02	.56
1982 -88 n=140	35.44 (20.57	.02) (.23)	.10 [*] (1.69)	.18 ^{**} (3.75)	**05* (-10.78)	***16 (-2.56	5 **. 53 5)	29.56***
1977 -88 n=240	27.76 (35.38	04) (.50)	03 (78)	.05 [*] (1.72)	03 [*] (-10.44)	***19 (-3.33	9 ^{***} .34 3)	24.43***
Notes:	CHWi CT CHSWEi CHPRi SIi WLi CHFEi Levels	<pre>= Change = Constan = Change = Change = Index o = Initial = Change of Signi istice in</pre>	in Nominal t Term. in Employm ch i. in Product f Strike A Level of in the Sha ficance: brackets	Hourly Ment of s civity in Activity Wage in Are of Fo * = 0.10	Wage Pa Salary a n Branch in Brar Branch emale En 0, ** =	aid in H and Wage n i. nch i. i. nploymer 0.05, #	Branch E Earne Int in	i. rs Branch i. .01.

344

1

. •

REFERENCES

- Alivizatos, N. 1983: <u>The Political Institutions in Crisis: 1922-</u> <u>1974</u>, Themelio, Athens, (in Greek).
- Artis M. J. 1981: Incomes Polices. Some Rationales, in Fallick J. L. & Elliott R. F. (editors) 1981: <u>Incomes Policies,</u> <u>Inflation and Relative Pay</u>, Allen & Unwin, London.
- Ashenfelter, O. 1985: Evidence on U.S. Experience with Dispute Resolution Systems, Princeton University, Industrial Relations Section, Working Paper No 185, March.
- Ashenfelter, O. 1987: Arbitrator Behaviour, <u>American Economic</u> <u>Review</u>, Vol. 77, No 2, pp. 342-346.
- Ashenfelter, O. and Johnson, G. E. 1969: Bargaining Theory, Trade-Unions and Industrial Strike Activity, <u>American</u> <u>Economic Review</u>, Vol. 59, No 1, pp. 35-49.
- Baglioni, G. 1989: Industrial Relations in Europe in the 1980s, Labour and Society, Vol.14, No 3, July, pp. 233-249.
- Bamber, G. J. and Lansbury, R. D. (editors) 1987: <u>International and</u> <u>Comparative Industrial Relations. A Study of Developed</u> <u>Market Economies</u>, Allen and Unwin.
- Bell, L. and Freeman, R. B. 1985: "Does a Flexible Industry Wage Structure Increase Employment? The U.S. Experience", National Bureau of Economic Research Working Paper, No

1985, April.

- Blackaby F. 1971: Incomes Policies and Inflation, <u>National</u> <u>Institute Economic Review</u>, No 58, November, pp. 34-53.
- Blain, N. Goodman, J. and Loewenberg, J. 1987: Mediation, Conciliation and Arbitration. An International Comparison of Australia, Great Britain and the United States, <u>International Labour Review</u>, Vol.126, No 2, March-April, pp. 179-198.
- Blandy, R. and Niland, J. (editors) 1986: <u>Alternatives</u> to <u>Arbitration</u>, Allen and Unwin.
- Blanpain, R. (editor), 1987: <u>Comparative</u> <u>Labour</u> <u>Law</u> <u>and</u> <u>Industrial</u> <u>Relations</u>, Kluwer Law and Taxation Publishers.
- Bloom, D. E. 1981: Is Arbitration Really Compatible with Bargaining? <u>Industrial Relations</u>, Vol. 20, No 3, Fall, pp. 233-244.
- Bloom D. E. and Cavanagh, C. L. 1986: An Analysis of the Selection of Arbitrators, <u>American Economic Review</u>, Vol.76, No 3, pp. 408-422.
- Bloom D. E and Cavanagh, C. L. 1987: Negotiator Behaviour under Arbitration, <u>American Economic Review</u>, Vol. 77, No 2, pp. 353-358.
- Bowen, W. G. 1960: <u>The Wage-Price Issue: A Theoretical Analysis</u>, Princeton University Press.

- Braun, A. R. 1974: Compulsory Arbitration as a Form of Incomes Policy: The Australian Case, <u>International Monetary Fund</u> <u>Staff Papers</u>, Vol.21, No 1, March, pp. 170-216.
- Brown A. R. 1975: The Role of Incomes Policy in Industrial Countries since World War II, <u>IMF Staff Papers</u>, Vol. 22, No 1, March, pp. 1-36.
- Braun, A. R. 1986: <u>Wage Determination and Incomes Policy in Open</u> <u>Economies</u>, International Monetary Fund, Washinghton D. C.
- Brown, W. 1987: Industrial Conflict Resolution in Great Britain, in Hanami, T. and Blanpain, R. (editors) 1987: <u>Industrial</u> <u>Conflict Resolution in Market Economies</u>, Kluwer Law and Taxation Publishers.
- Brown, W. Hayles, J. Hughes, B. and Rowe, L. 1980: Occupational Pay Structures Under Different Wage Fixing Arrangements: A Comparison of Intra-Occupational Pay Dispersion in Australia, Great Britain and the United States, <u>British Journal of Industrial Relations</u>, Vol. XVIII, No 2, July pp. 217-230.
- Bruno, M. and Sachs, J. D. 1985: <u>Economics of World-wide</u> <u>Stagflation</u>, Basil Blackwell.
- Calmfors, L. and Driffil, J. 1988: Centralisation of Wage Bargaining: Bargaining Structure, Corporatism and Macroeconomic Performance, <u>Economic Policy</u>, April, pp. 14-61.

- Chamberlain, N. W. 1961: Determinants of Collective Bargaining Structures, in Weber, A. R. (editor) 1961: <u>The structure of</u> <u>Collective Bargaining. Problems and Perspectives</u>, The Free Press of Ulence Inc., Chicago.
- Chan-Lee, J. H., Coe, D. and Prywes, M. 1987: Microeconomic Changes and Macroeconomic Wage Disinflation in the 1980s, <u>OECD Economic Studies,</u> No 8, Spring, pp. 121-157.
- Clarke, O. 1987a: <u>Industrial Relations in a Changing Economic</u> <u>Environment: The Post-war Experience of Advanced Market</u> <u>Economies</u>, in Blanpain, R. (editor), 1987: <u>op.cit.</u>, pp. 39-55.
- Clarke, O. 1987b: <u>Industrial</u> <u>Conflict:</u> <u>Perspectives</u> <u>and</u> <u>Trends</u>, in Blanpain, R. (editor), 1987: <u>op.cit.</u>, pp. 383-399.
- Clegg, H. 1976: <u>Trade Unionism under Collective Bargaining.</u> <u>A Theory Based on Comparisons of Six Countries</u>, Blackwell.
- Committee of Review on Incomes Policy, 1976: <u>Development Plan</u> <u>1976-80: Report on Incomes Policy</u>, Centre for Planning and Economic Research (KEPE), Athens, (in Greek).
- Committee of Review into Australian Industrial Relations Law and Systems 1985: <u>Report, Volume 2</u>, Australian Government Publishing Service, Canberra.
- Cordova, E. 1987: <u>Collective Bargaining</u>, in Blanpain, R. (editor) 1987: <u>op. cit.</u>, pp. 307-335.

Creigh, S. W. 1986: Australia's Strike Record: The International

Perspective, in Blandy, R. and Niland, J. (editors) 1986: op.cit., pp. 29-51.

- Crouch C. 1977: <u>Class Conflict and Industrial Relations Crisis</u>, Heinemman, London.
- Deaton, D. R. and Beaumont, P. B. 1980: The Determinants of Bargaining Structure: Some Large Scale Survey Evidence for Britain, <u>British Journal of Industrial Relations</u>, Vol. XVII, No 2, July, pp. 202-216.
- Dolton, P. J. and Tremble, J. G. 1985: On "Final Offer" and "Not Quite Compulsory" Arbitration, <u>Scottish</u> <u>Journal</u> <u>of</u> <u>Political</u> <u>Economy</u>, Vol. 32, No 2, June, pp. 181-190.
- Dubois, P. 1978: <u>New Forms of Industrial Conflict</u>, in Crouch, C. and Pizzorno, A. (editors) 1978: <u>op.cit.</u>, pp. 1-34.
- Economou G. 1981: <u>Incomes Policies in Greece and International</u> <u>Experience</u>, Centre for Planning and Economic Research (KEPE), Athens (in Greek).
- Fakiolas, R. 1978: <u>Trade-Unionism in Greece</u>, Papazisis, Athens (in Greek).
- Farber, H. S. 1980: An Analysis of Final-Offer-Arbitration, Journal of Conflict Resolution, Vol. 24, No 4, pp. 683-705.
- Farber, H. S. and Bazerman, M. H. 1987: Why is There Disagreement in Bargaining? <u>American Economic Review</u>, Vol. 77, No 2, pp. 347-352.

- Feuille, P. 1979: Selected Benefits and Costs of Compulsory Arbitration, <u>Industrial and Labour Relations Review</u>, Vol. 33, No 1, pp. 64-76.
- Flanagan R. J., Soskice D. W. & Ulman L. (1983): <u>Unionism, Economic</u> <u>Stabilisation and Incomes Policies</u>, Brookings Institution, Washinghton D.C.
- Flanders A. 1970: Management and Unions, Faber, London.
- Freeman, R. B. 1987: Are British Wages Unresponsive to Market Forces? University of Kent, Seminar Hand-out, February 19.
- Galin, A. 1980: Aspects of Compulsory Intervention in Australia and Israel, <u>The Journal of Industrial Relations</u>, March, pp. 19-35.
- Gaudier, M. 1987: <u>Labour Market Flexibility: A Magic Wand or the</u> <u>Foundation of a New Industrial Society? A BIbliographical</u> <u>Essay</u>, International Institute of Social Studies, Geneva.
- Gregory, M. Lobban, P. and Thomson, A. 1986: Bargaining Structure, Pay Settlements and Perceived Pressures in Manufacturing 1979-84: Further Analysis from the C.B.I. Databank, <u>British Journal of Industrial Relations</u>, Vol. 24, No 2, July, pp. 215-232.

Hicks, J. R. 1932: The Theory of Wages, Macmillan and Co, London.

Holt, J. 1986: <u>Compulsory Arbitration in New Zealand</u>, Auckland University Press, Auckland.

- Hunter, L. C. 1983: Arbitration in Robinson, D. and Mayhew, K. (editors), <u>Pay Policies for the Future</u>, Oxford University Press.
- I.L.O 1949: <u>Labour Problems in Greece. Report of the Mission of</u> <u>International Labour Office to Greece</u>, International Labour Office, Geneva.
- I.L.O 1980: <u>Conciliation and Arbitration Procedures in Labour</u> <u>Disputes, A Comparative Study</u>, International Labour Office, Geneva.
- I.L.O 1984: <u>Collective Bargaining: A Response to the Recession</u> <u>in Industrialised Market Economy Countries</u>, International Labour Office , Geneva.
- Ioannou, C. 1986: The Manufacturing Workers in Trade-Union Movement, 1974-1984, in <u>Social Classes, Social Change and Economic Development in Mediterranean</u>, Conference Proceedings, Foundation for Mediterranean Studies, Athens, (in Greek).
- Ioannou, C. 1988: Wage Policy, Collective Bargaining and Strike Activity in Greece, <u>Greek Review of Social Sciences</u>, No 68, pp. 73-110 (in Greek).
- Ioannou, C. 1989: <u>Employment</u> and <u>Trade-Unionism</u> in <u>Greece</u>, Foundation for Mediterranean Studies, Athens, (in Greek).
- Jechinis C. 1967: <u>Trade-Unionism</u> in <u>Greece</u>, Roosevelt University, Chicago.

- Kahn-Freund, O. 1974: On Uses and Misuses of Comparative Law, <u>The</u> <u>Modern Law Review</u>, Vol. 37, No 1, January, pp. 1-27.
- Karantinos, D. 1989: Open Unemployment in Urban Labour Markets in Greece, 1974-1985, <u>Greek Review of Social Research</u>, No 70, pp. 50-115.
- Kasimatis, B. 1978: Factory Unions: A Path-Breaking Institution, <u>Politis</u>, No 17, March, pp. 4-9.
- Katsanevas T. 1984: <u>Trade Unions in Greece</u>, National Centre for Social Research, (EKKE), Athens.
- Keating, M. 1983: Relative Wages and the Changing Industrial Distribution of Employment in Australia, <u>The Economic</u> <u>Record</u>, Vol. 59, No 167, December, pp. 384-397.
- Kennan, J. 1986: The Economics of Strikes, in Ashenfelter, O. and Layard, R. (editors) 1986: <u>Handbook of Labour</u> <u>Economics, Volume II</u>, Elsevier Science Publisher.
- Kioulafas, K. 1987: <u>Wages in Greece: Developments in Public</u> <u>Sector and Manufacturing, 1975-1985</u>, Institute for Economic and Industrial Research, (IOBE), Athens, (in Greek).
- Koukoules, G. 1984: <u>Greek Trade-Unions, Financial Autonomy and</u> <u>Dependence, 1938-1984</u>, Odysseas, Athens, (in Greek).
- Koukoules G. 1988: The Trade-unionism that Never Was, <u>Oikonomikos Tahydromos,</u> 3 March, pp. 139-160 (in Greek).

- Lansbury, R. and Davis, E. 1987: Australian Industrial Relations. in Bamber, G. J. and Lansbury, R. D. (editors) 1987: <u>op.cit.</u>, pp. 97-117.
- Lash, S. 1985: The End of Neocorporatism? The Breakdown of Centralised Bargaining in Sweden, <u>British Journal of</u> <u>Industrial Relations</u>, Vol. 23, No 2, pp. 215-240.
- Layard R. 1982: <u>More Jobs Less Inflation.</u> The Case for a <u>Counter-Inflation Tax</u>, Grant McIntyre, London.
- Leslie, D. 1985: Real Wage Growth, Technical Change and Competition in the Labour Market, <u>The Review of Economics</u> <u>and Statistics</u>, November, Vol. LXVII, No 4, pp. 640-647.
- Manicas, N. 1984: The Factory Trade-Union Movement, 1974-1981: An Overview, in Manicas, N. and Charalambidis, M. 1984: <u>Towards a New Trade-Union Movement</u>, Aletri, Athens, (in Greek).
- Marsden, D. 1981: Vive la Difference: Pay Differentials in Britain, West Germany, France and Italy, <u>Employment</u> <u>Gazette</u>, Vol. 89, No 7, pp. 309-318, 324.
- Marsden, D. 1983: Wage Structure, in Bain, G. S. (editor) 1983: <u>Industrial Relations in Britain</u>, Blackwell.
- Meade, J. E. 1982: <u>Stagflation Volume 1, Wage-Fixing</u>, George Allen and Unwin.
- Metcalf D. 1977: Unions, Incomes Policy and Relative Wages in Britain, <u>British Journal of Industrial Relations,</u> Vol.

15, No 2, July, pp. 157-175.

- Metcalf, D. 1987: Labour Market Flexibility and Jobs: A Survey of Europe, in Layard, R. and Calmfors, L. (editors) 1987: <u>The</u> <u>Fight Against Unemployment</u>, The MIT Press, London.
- Mulvey, C. 1986: Alternatives to Arbitration: Overview of the Debate, in Blandy, R. and Niland, J. (editors) 1986: <u>op.cit</u>. pp. 11-28.
- Newell, A. and Symons, J. S. V. 1987: Corporatism, Laissez-Faire and the Rise in Unemployment, <u>European Economic Review</u>, Vol. 31, No 3, pp. 567-614.
- Norris, K. 1980: Compulsory Arbitration and the Wage Structure in Australia, <u>The Journal of Industrial Relations</u>, September, pp. 249-263.
- OECD (1971): Economic Surveys, Greece, February, OECD, Paris.
- OECD (1972): Economic Surveys, Greece, November, OECD, Paris.
- OECD 1985: Employment Outlook, September, Paris.
- OECD 1986: <u>Flexibility in the Labour Market: The Current Debate</u>, Paris
- OECD, 1987: <u>Structural Adjustment and Economic Performance</u>, OECD, Paris
- Paldam, M. 1983: Industrial Conflicts and Economic Conditions, A Comparative Empirical Investigation, <u>European</u> <u>Economic</u>

Papers, Vol.20, No 1-3, pp. 231-256.

- Paldam, M. 1989: A Wage Structure Theory of Inflation, Industrial Conflicts and Trade-Unions, <u>Scandinavian Journal of</u> <u>Economics</u>, Vol.91, No 1, pp. 63-89.
- Paldam, M. and Pedersen, P. J. 1984: The Large Pattern of Industrial Conflict, A Comparative Study of 18 Countries, 1919-1979, <u>International Journal of Social Economics</u>, Vol.11, No 5, pp. 3-28.
- Pavlopoulos, P. G. 1986: <u>Income Shares: Trends, Causes and</u> <u>Effects,</u> Institute for Economic and Industrial Research, (IOBE), Athens, (in Greek).
- Pissarides, C. and McMaster, I. 1984: Economy-wide and Sectorspecific Influences on Relative Wages, London School of Economics, Centre of Labour Economics, Working Paper No 571.
- Pissarides, C. and Moghadam, R. 1989: Relative Wage Flexibility in Four Countries, London School of Economics, Centre of Labour Economics, Discussion Paper No 331.
- Pollis, A. 1987: <u>The Greek Judiciary</u>, Unpublished Study, Foundation for Mediterranean Studies, Athens.
- Plowman, D. 1989: <u>Holding the Line: Compulsory Arbitration and National Employer Co-ordination in Australia</u>, Cambridge University Press.

Reddaway, W. B. 1959: Wage Flexibility and the Distribution of

.

Labour, Lloyds Bank Review, October, London.

- Reder, M. W. 1962: Wage Structure: Theory and Measurement, in NBER, 1962: <u>Aspects of Labour Economics</u>, Princeton University Press, Princeton.
- Sabethai, I. D. 1986: <u>Report on the Greek Labour Market</u>, mimeographed, Economic Research Department, Bank of Greece, Athens.
- Salter, W. E. G. 1960: <u>Productivity</u> and <u>Technical</u> <u>Change</u>, Cambridge University Press, Cambridge.
- Sarfati, H. and Cobrin, K. 1988: <u>Labour Market Flexibility: A</u> <u>Comparative Anthology</u>, Gower, London.
- Sargan J. 1971: A Study of Wages and Prices in the U.K., 1949-68 in Johnson H. G. & Nobay A. R. (editors) 1971: <u>The</u> <u>Current Inflation, Macmillan, London.</u>
- Saunders, C. and Marsden, D. 1981: <u>Pay Inequalities in the</u> <u>European Community</u>, Butterworths.
- Schmitter P. & Lembruch G. (editors) 1979: <u>Trends</u> <u>Towards</u> <u>Corporatist Intermediation</u>, Sage, New York.
- Seferis, G. 1976: <u>Greek Trade-Unions, 1860-1975</u>, New Trade-Union Movement, Athens, (in Greek).
- Sisson, K. 1987: <u>The Management of Collective Bargaining: An</u> <u>International Comparison</u>, Basil Blackwell.

- Soskice, D. 1978: Strike Waves and wage Explosions, 1968-1970. an Economic Interpretation, in Crouch, C. and Pizzorno, A. (editors) 1978: <u>op.cit.</u>, pp. 221-246.
- Starr, G. 1981: <u>Minimum Wage Fixing</u>, International Labour Office, Geneva.
- Stevens, C. 1966: Is Compulsory Arbitration Compatible with Bargaining, <u>Industrial Relations</u>, Vol.5, No 2, February, pp. 38-52.
- Suzuki, H. 1980: Wage Indexation in Industrialised Market Economies, <u>International Labour Review</u>, Vol. 119, No 4, pp.455-466.
- Tarantelli, E. 1984: The Regulation of Inflation in Western Countries and the Degree of Neo-Corporatism, in Cavaco Silva, A. (editor) 1984: <u>Economic and Social Partnership</u> <u>and Incomes Policy</u>, Faculdade de Ciencias Humanas da Universida Catolica Portuguesa, Lisbon.
- Tarling, R. and Wilkinson, F. 1982: Changes in the Inter-industry Structure of Earnings in the Post-war Period, <u>Cambridge</u> <u>Journal of Economics</u>, Vol. 6, No 3, pp. 231-248.
- Turner, H. A. 1965: <u>Wage Trends, Wage Policies and Collective</u> <u>Bargaining. The Problems of Underdeveloped Countries</u>, Occasional Paper No 6, Cambridge University Press.
- Turner-Samuels, K. C. 1951: <u>Industrial Negotiation</u> and <u>Arbitration</u>, The Solicitors' Law Stationery Society

Limited, London.

- Wachter M. 1970: Cyclical Variation in Interindustry Wage Structure, <u>American Economic Review</u>, Vol. 60, No 1, March, pp. 75-84.
- Wachter M. 1974: Phase II, Cost-Push Inflation and Relative Wages, <u>American Economic Review</u>, Vol. 64, No 3, June, pp. 482-491.
- Webb, J. 1986: Third Parties at Work: Conflict Resolution or Social Control? <u>Journal of Occupational Psychology</u>, No 59, pp. 247-258.
- Weber, A. R. 1967: Stability and Change in the Structure of Collective Bargaining, in Lloyd Ulman (editor) 1967: <u>Challenges to Collective Bargaining</u>, Prentice Hall, New Jersey.
- Windmuller, J. P. et all 1987: <u>Collective Bargaining in</u> <u>Industrialised Market Economies: A Reappraisal</u>, International Labour Office, Geneva.
- Withers, G. Pitman, D. and Whittingham, B. 1983: Wages and Labour Market Adjustment: A Cross-Country Causality Approach, Bureau of Labour Market Research, Working Paper No 32.

358

•